INVESTIGATIONS ON MICROGRID INTERCONNECTION TO DISTRIBUTION SYSTEM

Submitted in fulfillment of the requirements of the degree of

DOCTOR OF PHILOSOPHY

by

NANDKISHOR GUPTA

(2012REE9030)

Under the supervision of

PROF. R. A. GUPTA

to the



DEPARTMENT OF ELECTRICAL ENGINEERING MALAVIYA NATIONAL INSTITUTE OF TECHNOLOGY JAIPUR JAIPUR (INDIA) AUGUST, 2016

© MALAVIYA NATIONAL INSTITUTE OF TECHNOLOGY JAIPUR

ALL RIGHTS RESERVED



Department of Electrical Engineering MALAVIYA NATIONAL INSTITUTE OF TECHNOLOGY JAIPUR

CERTIFICATE

This is to certify that the thesis entitled "Investigations on Microgrid Interconnection to Distribution System" submitted by Nandkishor Gupta (ID. No. 2012REE9030) to Malaviya National Institute of Technology Jaipur for the award of the degree of Doctor of Philosophy in Electrical Engineering is a bonafide record of original research work carried out by him under our supervision.

It is further certified that:

 The results contained in this thesis have not been submitted in part or in full, to any other university or institute for the award of any degree or diploma.

0F

TEC

2. Nandkishor Gupta has fulfilled the requirement for the submission of this thesis.

(Prof. R. A. Gupta) Supervisor Department of Electrical Engineering MNIT Jaipur Jaipur - 302 017, India

Date: 13.08.2016



Department of Electrical Engineering MALAVIYA NATIONAL INSTITUTE OF TECHNOLOGY JAIPUR

CANDIDATE'S DECLARATION

I hereby declare that the thesis entitled "Investigations on Microgrid Interconnection to Distribution System" is my own work conducted under the supervision of Prof. R. A. Gupta, Department of Electrical Engineering, Malaviya National Institute of Technology, Jaipur (Rajasthan), India. I firmly declare that the presented work does not contain any part of any work that has been submitted for the award of any degree either in this University or in any other University/Deemed University without proper citation.

University without proper citation. Date: 13/08/2016 MNIT, Jaipur (Nandkishor Gupta) 2012REE9030

ABSTRACT

Increasing penetration of renewable energy resources to the distribution system in evolving deregulated power system has great public attention, recently. Among the renewable energy sources, the penetration of wind and solar generators has been increased due to their evolved technology and wide spread availability. To integrate these renewable sources as Distributed Energy Resources (DERs) into the utility grid, Micro-Grids (MGs) play an important role. MG is a small-scale power supply network that is designed to provide power for a small community and can increase the reliability and economics of energy supply. MGs are self-sustainable and generally operated in two modes: (1) grid connected and (2) grid isolated. In grid-connected mode, MG can export its excess power to the grid and can import its deficit power from grid. MG can manage its demand independently in grid-isolated mode.

An Energy Management System (EMS) is used in MGs to optimize their operation, schedule local generation, and control all the interactions with the upstream grid. Optimizing the operation of a MG is essential to reduce fuel cost, energy unscheduled, power losses, and gas emissions. Due to random nature of demand as well as renewable sources and to operate the generators within its technical limitations, the efficient and economic management of the generators is highly recommended. The grid price volatility in deregulated environment of power system has increased the complexity of MG's EMS. Therefore, development of efficient approaches for MG management under different uncertainties is necessary. The work presented in this thesis concentrates on development of efficient approaches for MG management with an objective to minimize the total cost of operation subject to different constraints considering renewable power uncertainties.

In order to maximize the benefits of the resources available in a MG, an optimal scheduling of the power generation is considered in this work. Generation scheduling problem is an optimization problem that consists of two sub-problems: Unit Commitment (UC), and Economic Dispatch (ED). The unit commitment problem provides the on/off status of the dispatchable generation units over a daily or weekly time horizon. On the other hand, the economic dispatch problem finds the optimal output power for the units committed by the unit commitment problem over shorter time horizons: *i.e.*, hourly or in real time. The optimal solution thus obtained satisfies the operational constraint.

MG operator can solve the energy scheduling problem with an objective to minimize the total operation cost considering the day-ahead power balance constraint, real time power balance constraint, dispatchable power constraint, imbalance constraint and grid power constraint. In grid connected mode, the total cost of operation of MG is "The sum of operating cost of dispatchable generating unit's power, reserves and cost of power imported/exported from/to grid". In grid isolated mode also, the objective of MO is to minimize the total cost of operation, similar to the grid connected mode. However, the cost of exchange of power between main grid & MG is replaced by the cost of demand shedding. Since, the cost of demand shedding is very high, it is generally avoided till the complete utilization of the reserve capacity. In grid connected mode, the MG is connected to the utility grid thereby facilitating the power transaction between the MG and the utility grid. In the deregulated environment, MG can earn revenue by selling its reserve capacity to the spot market. Additionally, MG can also earn revenue by selling its reserve capacity to the spot market if prices are higher than its marginal cost. On the other hand the cost of operation can be minimized by meeting its deficit power from the utility grid when grid power is cheaper than MG's reserve cost.

One of the benefits of establishing MGs is increasing the penetration of renewable resources in the distribution system but a major problem is the intermittent nature. The amount of power generated by renewable resources solely depends on the weather parameters such as wind speed and solar radiation. Therefore, these generators can neither be dispatched nor committed like conventional thermal generators. Furthermore, renewable generators can cause load mismatch and voltage instability in the system. These problems are more significant in the case of MG due to higher penetration level of renewable generators must be taken into consideration when scheduling the power generation in MGs in order to achieve reliable solutions. Hence, reformulating the scheduling problem and developing new models is necessary to produce efficient and robust commitment schedules. In the proposed work, wind and solar power are considered as uncertain parameters.

Random nature of wind and solar power can be modeled by stochastic process. ARIMA model is a popular approach for modeling any stochastic process and is widely used for forecasting electricity prices & demand. As the wind and solar power vary with atmospheric conditions, the time series variation of wind power is non-stationary, since, ARIMA model is applicable for only stationary data series, the wind and solar power are converted into stationary series by differentiation. In proposed work, ARIMA model is used for forecasting upper and lower limit of wind power and solar power on 95% confidence interval. These forecasted limits are used in proposed robust optimization based MG scheduling approach considering wind and solar power as uncertain parameter.

A robust optimization approach has been extensively used for solving different decision making problems with different uncertainties. For modeling of these problems, Robust Optimization (RO) provides a solution that is guaranteed to be "good" for all or most possible realization of the uncertain parameter. In any decision making problem there are three types of robustness in optimization problem (i) constraint robustness (ii) objective robustness (iii) combinational robustness. If uncertainty is present in only constraint parameter then problem is defined as constrained robustness and if uncertainty is present in only objective function parameters, the problem is defined objective robustness. Due to effective modeling of uncertain parameters, the problem formulated by the robust optimization is tractable and has low computational burden. In this work, since wind and solar are considered as uncertain parameters in the constraints, it is formulated as a constrained robustness problem.

A comparative analysis on MG operation is carried out. For this purpose, the same problem is simulated using deterministic and stochastic programming approach. In deterministic approach, wind power output is forecasted using ARIMA model and the forecasted value of wind power is considered to compute the scheduling of MG.

The impact of degree of robustness on operation cost, in both cases has been evaluated by varying the degree of robustness. Per unit cost of operation, at any degree of robustness is obtained by dividing its actual value by the reference value, which represents the cost of operation at zero degree of robustness. From the analysis, it was observed that the optimal value of degree of robustness is 24, where per unit cost of MG operation is minimum. At this degree of robustness, optimal condition is obtained because the uncertainty of wind power is considered for 24 hours. This degree of robustness takes into account all possible deviations of the uncertain parameter for whole day. Additionally, at zero degree of robustness the cost of operation obtained using proposed robust optimization based approach is equal to the cost of operation obtained using deterministic approach.

Further in this work the solar power uncertainty is considered in MG generation scheduling problem. The Modeling of solar power uncertainty has been done through Time series based ARIMA model. Robust optimization based formulation of MG generation scheduling problem in both grid connected and isolated mode considering solar power uncertainty has been proposed. The results obtained through robust optimization approach has been compared with traditional stochastic and deterministic approach and the impact of degree of robustness on the optimal generation scheduling through robust optimization based approach has been evaluated through realistic case studies. A study of multiple uncertainties considering wind power and solar power uncertainty is incorporated in MG generation scheduling problem and the results obtained through robust optimization approach have been compared with traditional stochastic and deterministic approaches.

A robust optimization based approach has been proposed for optimal generation scheduling of MG in both grid-connected and grid-isolated modes. Wind power and solar power uncertainties have been modeled through interval forecasting using ARIMA model. A comparative analysis on daily cost of operation of MG using proposed robust optimization based approach with deterministic and stochastic approach has been performed. A significant reduction in cost of operation in the proposed approach shows the strength of proposed robust optimization based approach in MG generation scheduling in the deregulated environment. The impact of degree of robustness on the cost of operation in both cases has also evaluated and compared with existing methods. The proposed approach is more advantageous in incorporating different type of uncertainties. The consideration of correlation between different uncertainties is also an important extension of this work. This work may also be enhanced by incorporating multiple-micro grids, different type of consumers and considering different technical environmental constraints. It would not have been possible to write this thesis without the help and support of the kind people around me, to only some of whom it is possible to give particular mention here.

Foremost, I would like to express my deepest gratitude to my research supervisor, **Prof. R. A. Gupta,** Professor Department of Electrical Engineering, Malaviya National Institute of Technology (MNIT), Jaipur, for his valuable guidance, scholarly inputs and consistent encouragement. This work is possible only because of the unconditional moral support provided by him. I had a great freedom to plan and execute my ideas in research without any pressure. This made me to identify my own strength and drawbacks, and particularly boosted my self-confidence. I am inspired from you about the true ideas, which are most important for budding researchers like me. I always ended up with confidence and full of energy after the discussions with you. You gave me your full support whenever I had problems. Thank you again for your care and kindness. I look forward to even greater level of support and guidance for my future endeavours.

I feel profound privilege to thank **Prof. A. B. Gupta**, present Director MNIT Jaipur and **Prof. I. K. Bhat**, Ex-Director MNIT Jaipur for providing me an opportunity to work towards my Ph.D. degree and for excellent infrastructure facilities in the institute.

My heartily thanks goes to **Dr. Vikas Gupta**, present HOD and **Prof. Manoj Fozdar**, Ex-HOD, Electrical Engineering Department for extending all the necessary support and providing a healthy research atmosphere in the department. I thank all the faculty members of the department for their continuous encouragement and moral support, which motivated me to strive for better. I wish to express fruitful thanks to **Dr. Rajesh Kumar**, Convener, DPGC, and **Prof. K. R. Niazi**, Ex-Convener, DPGC for their help and kind cooperation. I express my appreciations to **Dr. H. P. Tiwari**, **Dr. Rajive Tiwari** and **Dr. Nitin Gupta**, member of Departmental Research Evaluation Committee (DREC), for the inspiring discussions and fruitful suggestions that enhanced the quality of my research work. I would also like to thank the technical and support staff of the department for their support and help, whenever I needed.

I have no words to express my thanks to my research group members Dr. Manoj Gupta, Dr. Ajay Kumar Bansal, Dr. Virendra Swaroop Sangtani and Mr. Bharat Bhushan Jain for their cooperation and lifelong memorable association. It was an immense pleasure to work alongside with them.

Acknowledgements

My heartfelt thanks to **Dr. Kailash Chand Sharma**, it was wonderful to share my office room with you. The discussion I used to have with you on various topics provided a relaxing time from research. Apart from this, your advice during my struggling period with research was of great significance.

Special thanks to my present and ex colleagues with whom I have enjoyed my past three years, starting from Dr. Pushpendra Singh, Dr. Manjur A. Ali, Dr. B. C. Kharbas, Dr. S. A. Siddiqui, Dr. Vinay Jadoun, Mr. Narayanan K., Dr. Gulshan Sharma, Mr. Dhanraj Chittara, Ms. Neeraj Kanwar, Mr. Pradeep Anjana, Mr. Manoj Kumawat and Mr. Ajeet.

I would also like to thank my lifelong friends Mr. Ashish Singhal, Mr. M. P. Singh, Mr. Sunil Agarwal, Mr. Satyaveer Singh, and Mr. Lalit Gupta for their friendship and unconditional moral support.

I owe my special thanks to my supervisor's family members particularly **Mrs. Madhu Gupta and Mr. Siddharth Gupta**, for always being on my side morally and for their continuous encouragements during this work. They deserve my special thanks for sparing the family time of my supervisor during this work.

I would like to express profound gratitude to my family members for all they have undergone to bring me up to this stage. I have no words to express towards my Grand Father Lt. Sh. Shankar Lal Gupta and Grand Mother Smt. Rukmani Devi for their encouragement. I sincerely express my special thanks and gratitude to my father Shri Bhagwan Sahay Gupta, my mother Smt. Meena Devi for their affection, encouragement and support during all these years of my education. You are my greatest strength and I am blessed to be your son, there are no words to express my gratitude towards you.

I would like to give my special thanks to my father in law Shri Subhash Chand Goyal and my mother in law Smt. Beena Devi Goyal for showing great confidence on me. Again I want to express my special thanks to my beloved inlaws Ms. Divya and Mr. Balkrishan Goyal for giving me unconditional support at each instant of time and always keeping blind faith on me.

I would like to record high appreciation for my wife **Suman** for her constant encouragement and tremendous patience during the completion of this work.

Finally, it is a pleasant task to express my thanks to all those who contributed directly or indirectly in many ways to the success of this study and made an unforgettable experience for me.

NANDKISHOR GUPTA

CONTENTS

Certificat	e		i
Candidate's Declaration			ii
Acknowledgement			iii
Abstract			v
Contents			ix
List of Fig	gures		xiii
List of Ta	bles		XV
List of Sv	mbols		xvii
List of Ab	breviatio	ns	xix
Chapter 1	INTRO	DUCTION	1-8
1.1	General		1
1.2	Motivati	on for the Present Work	2
1.3	Contribu	itions of the Present Work	5
1.4	Organiza	ation of the Thesis	6
Chapter 2	2 LITERA	ATURE REVIEW	9-26
2.1	Introduc	tion	9
2.2	Microgr	id Operation	10
2.3	Critical	Review	13
	2.3.1	Generation Scheduling in Microgrid Operation	13
	2.3.2	Microgrid Operation Considering Renewable Power Uncertainties	14
	2.3.3	Robust Optimization Based Uncertainty Characterization	19
		2.3.3.1 Static Robust Optimization	19
		2.3.3.2 Multi-Stage Decision-Making	21
2.4	Research	n Challenges and Objectives	25
2.5	Summar	У	26
Chapter 3	UNCER	TAINTY CHARACTERIZATION	27-40
3.1	Introduc	tion	27
3.2	ARIMA	Model	28
	3.2.1	Non-Seasonal ARIMA Models	30
	3.2.2	Seasonal ARIMA Models	30

	3.2.3	Statistical Features	32
	3.2.4	ACF and PACF	33
	3.2.5	White Noise	34
3.3	Uncerta	ainty Characterization in Stochastic Optimization Framework	35
3.4	Uncerta	ainty Characterization in Robust Optimization Framework	37
3.5	Summa	ıry	40
Chapter	4 ROB ENV	UST MICROGRID OPERATION IN DEREGULATED	41-66
4.1	Introdu	ction	41
4.2	Probler	n Description	42
	4.2.1	Grid Connected Mode	42
	4.2.2	Grid Isolated Mode	44
4.3	Uncerta	ainty Modeling	46
4.4	Robust	Optimization Approach	47
4.5	Propos	ed Problem Description	47
	4.5.1	Grid Connected Mode	47
	4.5.2	Grid Isolated Mode	50
4.6	Propos	ed Algorithm	53
4.7	Case St	tudy	55
	4.7.1	Data	55
	4.7.2	Simulation Results	56
	4.7.3	Discussion	63
4.8	Summa	ıry	65
Chapter	5 ROBU POWI	JST MICROGRID OPERATION CONSIDERING SOLAR ER UNCERTAINTY	67-98
5.1	Introdu	ction	67
5.2	Probler	n Description	68
5.3	Uncerta	ainty Modeling	68
5.4	Stochas	Stochastic and Robust Optimization Approach	
5.5	Propos	ed Problem Formulation in Stochastic Framework	70
	5.5.1	Grid Connected Mode	70
	5.5.2	Grid Isolated Mode	72
5.6	Propos	ed Problem Formulation in Robust Optimization Framework	73
	5.6.1	Grid Connected Mode	74
	5.6.2	Grid Isolated Mode	76
5.7	Propos	ed Methodology	79
5.8	Results	and Discussions	81

	5.8.1	Solar Power Uncertainty Modeling	83
	5.8.2	Stochastic Optimization Based MG Operation in Grid Connected Mode	86
	5.8.3	Stochastic Optimization Based MG Operation in Grid Isolated Mode	88
	5.8.4	Robust Optimization Based MG Operation in Grid Connected Mode	91
	5.8.5	Robust Optimization Based MG Operation in Grid Isolated Mode	93
	5.8.6	Comparative Analysis	95
	5.8.7	Impact of Degree of Robustness	96
5.9	Summary		97
Chapter	6 ROBUS AND SC	Г MICROGRID OPERATION CONSIDERING WIND DLAR POWER UNCERTAINTY	99-120
6.1	Introducti	on	99
6.2	Problem I	Description	100
6.3	Uncertain	ty Modeling	101
6.4	Robust Op	t Optimization Approach	
6.5	Proposed	Problem Description	102
	6.5.1	Grid Connected Mode	103
	6.5.2	Grid Isolated Mode	105
6.6.	Proposed	Simulation Procedure	108
6.7	Result and	1 Discussions	111
	6.7.1	Data	111
	6.7.2	Case I: Microgrid Operation in Grid Connected Mode	113
	6.7.3	Case II: Microgrid Operation in Grid Isolated Mode	115
	6.7.4	Discussion	117
6.8	Summary		119
Chapter	7 CONCLU	ISIONS AND SUGGESTIONS FOR FURTHER WORK	121-124
7.1	General		121
7.2	Summary	of the Significant Findings	121
7.3	Suggestio	ns for Further Work	123
Referenc	es		125-138
Publicati	ions from th	e Work	139

The abbreviations used in the text have been defined at appropriate places, however, for easy reference, the list of abbreviations is given below.

Abbreviation	Explanation
MG	Micro Grid
DG	Distributed Generation
DER	Distributed Energy Resource
EMS	Energy Management System
ESS	Energy Storage System
ACF	Auto Correlation Function
PV	Photo Voltaic
ARIMA	Auto Regressive Integrated Moving Average
ARMA	Auto Regressive Moving Average
AMI	Advanced Meter Infrastructure
DSM	Demand Side Management
ANN	Artificial Neural Network
LP	Linear Programming
MILP	Mix Integer Linear Programming
QP	Quadratic Programming
MADS	Mesh Adaptive Direct Search
MR-CGA	Matrix Real-Coded Genetic Algorithm
DP	Dynamic Programming
ADP	Advance Dynamic Programming
SR	Spinning Reserve
SMES	Smart Energy Management System
RH	Rolling Horizon
RO	Robust Optimization
CVaR	Conditional Value at Risk
PACF	Partial Autocorrelation Factor

List of Abbreviations

LIST OF FIGURES

Figure No.	Figure Description	Pg. No.
Fig. 2.1	Typical architecture of MG	11
Fig. 2.2	The energy management system (EMS)	13
Fig. 3.1	Possible uncertainties in MG operation	28
Fig. 3.2	Computational framework for stochastic MG operation	36
Fig. 3.3	Computational framework for robust MG operation	38
Fig. 4.1	Flow chart of proposed algorithm	54
Fig. 4.2	ACF plot of sample wind power data	57
Fig. 4.3	PACF plot of sample wind power data	57
Fig. 4.4	Hourly wind power	58
Fig. 4.5	Scheduled power of dispatchable units in Case I	59
Fig. 4.6	Scheduled upward reserve in Case I	59
Fig. 4.7	Scheduled downward reserve in Case I	60
Fig. 4.8	Power imported/exported to/from grid in Case I	60
Fig. 4.9	Dispatchable unit scheduled power in Case II	61
Fig. 4.10	Scheduled upward reserve in Case II	62
Fig. 4.11	Scheduled downward reserve in Case II	62
Fig. 4.12	Demand shedding in Case II	63
Fig. 4.13	Impact of degree of robustness on operational cost of MG	64
Fig. 5.1	Flow chart of proposed algorithm	81
Fig. 5.2	Daily market price and demand profile	82
Fig. 5.3	Forecasted wind power output profile	83
Fig. 5.4	ACF plot of sample solar power data	84
Fig. 5.5	PACF plot of sample solar power data	84
Fig. 5.6	Generated 1000 solar power scenarios (green) along with mean scenario (red) and forecasted solar power (black)	85
Fig. 5.7	Reduced 25 solar power scenarios (green) along with mean scenario (red) and forecasted solar power (black)	85
Fig. 5.8	Hourly forecasted solar power with upper and lower limits	86
Fig. 5.9	Scheduled power of dispatchable units in grid connected mode	86

List of Figures

Figure No.	Figure Description	Pg. No.
Fig. 5.10	Scheduled downward reserve in grid connected mode	87
Fig. 5.11	Scheduled upward reserve in grid connected mode	87
Fig. 5.12	Power imported/exported to/from grid in grid connected mode	88
Fig. 5.13	Dispatchable unit scheduled power in grid isolated mode	89
Fig. 5.14	Scheduled upward reserve in grid isolated mode.	89
Fig. 5.15	Scheduled downward reserve in grid isolated mode	90
Fig. 5.16	Demand shedding in grid isolated mode	90
Fig. 5.17	Scheduled power of dispatchable units in grid connected mode	91
Fig. 5.18	Scheduled upward reserve in grid connected mode	91
Fig. 5.19	Scheduled downward reserve in grid connected mode	92
Fig. 5.20	Power imported/exported to/from grid in grid connected mode	92
Fig. 5.21	Dispatchable unit scheduled power in grid isolated mode	93
Fig. 5.22	Scheduled upward reserve in grid isolated mode	94
Fig. 5.23	Scheduled downward reserve in grid isolated mode	94
Fig. 5.24	Demand shedding in grid isolated mode	95
Fig. 5.25	Impact of degree of robustness on operational cost of MG	96
Fig. 6.1	Generation scheduling problem of grid connected and isolated MG	101
Fig. 6.2	Flow chart of proposed algorithm	111
Fig. 6.3	Scheduled power of dispatchable units	113
Fig. 6.4	Scheduled upward reserve	114
Fig. 6.5	Scheduled downward reserve	114
Fig. 6.6	Power imported/exported to/from grid	115
Fig. 6.7	Dispatchable unit scheduled power	116
Fig. 6.8	Scheduled upward reserve	116
Fig. 6.9	Scheduled downward reserve	117
Fig. 6.10	Demand shedding	117
Fig. 6.11	Impact of degree of robustness on operational cost of MG.	118

The symbols used in the text have been defined at appropriate places, however for easy reference, the list of principle symbols is given below.

Symbol	Explanation
A. Sets or Ind	lices
j	Index of conventional dispatchable units running from 1 to N_j
t	Index of time period running from 1 to N_t
W	Index of wind power units running from 1 to N_w
k	Index of robustness running from 1 to N_k
B. Constants	or Parameters
N_{j}	Number of dispatchable units
N_t	Number of time periods
λ_{grid}, t	Grid power price at time $t (\$ / kWh)$
λ_d^{shed}	Load shedding price $(\$/kWh)$
$P_{d,t}$	Forecasted system's demand at time $t(kW)$
$P^f_{w,t}$	Forecasted wind power generation at time $t(kW)$
PV_t	Forecasted solar power generation at time $t(kW)$
P_w^{\max}	Total Installed capacity of wind units (kW)
P_j^{\min}	Minimum output power of unit $j(kW)$
P_j^{\max}	Maximum output power of unit $j(kW)$
P_{grid}^{\max}	Capacity of the line linking the upstream grid and the $MG(kW)$
$R_j^{u,\max}$	Maximum upward reserve capacity of conventional generating units
$R_j^{d,\max}$	Maximum downward reserve capacity of conventional generating units
C. Variables	
$P_{j,t}$	Power generated by dispatchable unit j at time $t(kW)$
$P_{grid,t}$	Power imported/exported from/to the grid at time $t(kW)$
$P_{d,t}^{Shed}$	Demand shedding at time $t(kW)$
$r_{j,t}^u$	Scheduled upward reserve of unit j at time $t(kW)$
$r_{j,t}^d$	Scheduled downward reserve of unit j at time $t(kW)$
$\Delta P_{w,t}$	Expected wind power deviation at time $t(kW)$

Symbol	Explanation
$\Delta PV_{v,t}$	Expected PV power deviation at time $t(kW)$
$\Delta P_{d,t}$	Expected demand deviation at time $t(kW)$
$P^a_{w,t}$	Actual wind power generated at time $t(kW)$
$W_{t,k}$	Expected value of wind power at k^{th} iteration at time $t(kW)$
$PV^{a}_{v,t}$	Actual PV power generated at time $t(kW)$
$P_{d,t}^a$	Actual demand of MG at time $t(kW)$
G^k	Incremental factor at k^{th} iteration

LIST OF TABLES

Table No.	Table Description	Pg. No.
Table 3.1	Shapes of ACF and PACF to identify AR or MA	34
Table 4.1	Dispatchable Unit Parameters	55
Table 4.2	Daily Market Prices, Load and PV Profiles	56
Table 4.3	Comaparitive Analysis of MG Daily Operation Cost	63
Table 5.1	Dispatchable Unit Parameters	82
Table 5.2	Comaparitive Analysis of MG daily operation cost	95
Table 6.1	Daily Market Price and Demand Profile	112
Table 6.2	Comparative Analysis of MG Daily Operation Cost	118

List of Tables

Chapter 1

Introduction

Τ

HIS chapter explains the background, context, and motivation for this work. It also explains main contributions of this research and then organization of the thesis chapters.

INTRODUCTION

1.1 General

Due to increase in electricity demand, scarcity of fossil fuel sources and environmental concerns with greenhouse gas emissions, integration of environment friendly renewable power sources in electric power system is increased throughout the world. Among the renewable power sources, integration of wind and solar power sources has been increased rapidly due to their mature technology and widespread availability. Integration of such renewable power sources in the electric power system at both transmission and distribution level causes several technical and financial challenges for designers and operators of electric power system due to their random nature. Therefore power generation at the consumer center is required to be distributed locally, which is also referred as Distributed Generation (DG).

DG's fuel source type can be selected on the basis of its local availability, operating cost, technological advancement of conversion system and environmental impact. In standby mode gas power generation sets or large diesel generators can be used to power up large load centers such as hospitals, telecommunication centers and large industries etc. Solar/PV cells, wind turbines and fuel cells etc. are the new comers in energy sector that are competing with many available generating sets in terms of size, efficiency and per kW cost. Today's DGs cost rate (\$/kWhr) is becoming more and more competitive as efficiency/technology behind the modern energy conversion units is continuously being improved and diversified. By combining a variety of dispersed DGs, a distributed energy resource (DER) domain is developed. Various mixtures of different energy sources are then controlled under a central energy management system (EMS) in order to improve efficiency and reliability of the operation. As incorporated the modern concept of DER and EMS theory, the Micro-Grid (MG) concept is put forwarded [1].

As newly created concept, there are various definitions existing for the MG. In recent literature, MG defines in different ways. A MG is a group of interconnected loads and distributed energy resources within clearly defined electrical boundaries that acts as a single controllable entity with respect to the grid. A MG can connect and disconnect from the grid to enable it to operate in both grid- connected and island-mode.

As another important entity which leads lots of MG research, Department of Defense (DOD), UK also provided their specific definition of MG [4] as follows:

"A MG is an integrated energy system consisting of interconnected loads and energy resources which, as an integrated system, can island from the local utility grid and function as a stand- alone system."

It can be observed from those three definitions that although different entities have various perspectives on MG, the common concept can be extracted that MG is a bunch of electrical loads within certain geographic boundary powered by local distributed energy resources (DER) which include DGs and energy storage system (ESS). In such a system, wind turbines, PV panels and other renewable generations can be used as DGs and the stability and reliability should be maintained by the system itself with the help of modern operation and control techniques. Furthermore, such a system can be either connected to the utility grid or operated in an islanded mode. In this dissertation, both operation modes are investigated and the corresponding decision- making model is proposed respectively.

1.2 Motivation for the Presented Work

One of the benefits of establishing MGs is increasing the penetration of renewable resources in the distribution system but a major problem is the intermittent nature of these sources. The amount of power generated by renewable resources solely depends on the weather parameters such as wind speed and solar radiation. Therefore, these generators can neither be dispatched nor committed like classical thermal generators. Furthermore, renewable generators can cause load mismatch and voltage instability in the system [3]. These problems are more significant in case of MG due to higher penetration level of renewable generators must be taken into consideration when scheduling the power generation in MGs in order to achieve reliable solutions. Hence, reformulating the scheduling problem and developing new models is necessary to produce efficient and robust commitment schedules.

MGs can operate under two different modes of operation *i.e.* grid connected mode and grid isolated mode. The major difference being the sources of power generation used to supply the demand and the reserve requirement. This has a direct impact on the generation scheduling problem making it more challenging. Therefore, the formulation must be

updated to account for the objectives and constraints in each mode of operation. This will result in a better distribution of the available generation capacities in the MG and will allocate the required amount of spinning reserve to maintain the system's stability and to mitigate the effects of uncertainties. Developing more accurate and reliable scheduling models that account for the effects of uncertainties and modes of operation is the main motivation behind this research. The work presented in this thesis examines the impacts of these added difficulties on the day-ahead unit commitment problem which is the first stage of the generation scheduling problem in a MG in distribution system. Two optimization models are presented; one for each mode of operation. The proposed models will tackle uncertainties by integrating uncertainty handling techniques in the problem formulation. Case studies are presented to study MG operation, and to evaluate the uncertainties and spinning reserve's effect.

In grid-connected mode, MG can export its excess power to the grid and can import its deficit power from grid. On the other hand, MG can manage its demand independently in grid-isolated mode. Due to uncontrollable nature of demand and generators' technical limitations, efficient and economic management of these generators is necessary. With increasing penetration of renewable sources like wind and solar in the distribution power system MG management system may become complex. The complexity of MG's energy management is further increased by the deregulation of power systems due to grid price volatility in MG management [4]-[5]. The work in this thesis focuses on overcoming these complex issues in MG operation. Further High cost of imbalance penalties and generation uncertainty in wind power generation in pool based day-ahead electricity markets provides a great challenge in the operation of MG. This research concentrates on the development of a new robust optimization based approach for optimal MG operation in deregulated environment considering renewable power uncertainties.

Several approaches have been proposed for MG's energy management considering renewable resources. Some of them are focused on control and scheduling of DERs, including renewable generation in MG [6]-[8]. A simple optimization based approach has been used for MG operation considering different market policies [9]. However, renewable power and market price uncertainties are modeled deterministically. A Lagrangian Relaxation along with genetic algorithm approach can optimally schedule generators of isolated MG [10]. However, the uncertainty in renewable generation is still modeled deterministically. Stochastic programming based approaches considering wind

and solar power uncertainties suggested for the optimal operation of MG [11], [12]. The impact of network losses and network constraints has also been investigated in [12]. Renewable power uncertainties have been modeled through different scenarios. An artificial intelligence based methodology can solve MG generation scheduling problem with an objective minimization of operation and emission cost considering renewable power and demand uncertainties. These uncertainties are modeled using artificial neural network [13].

All the above studies focus on minimizing the generation cost in MG using deterministic and stochastic based approaches. However scheduling in both modes i.e. grid connected and grid isolated is not considered and complexity of optimization problem in both deterministic and stochastic approach is not justified. These approaches have mainly two difficulties: (i) knowledge of the exact distribution of the data, and (ii) the size of the resulting optimization model. Difficulties in solving the problem drastically increase as a function of the number of scenarios, which is subjected to substantial computational challenges.

As mentioned in previous section, there are numbers of benefits of incorporating MG in distribution system. However, it is necessary for MG to maintain a stable and reliable operation to achieve those benefits. Due to its unique characteristics, there are multiple challenges for the MG operation.

Considering the penetration level of renewable generation, the volatility of renewable energy resources can impact the MG operation significantly. Those types of volatilities pose uncertainties in MG operation, especially in the MG resource scheduling. Currently, most of the large power system operators try to use renewable generation forecasting to alleviate the uncertainties in their scheduling. However, it is almost impossible to achieve a no-mismatch forecasting results with current techniques due to stochastic nature of renewable energy resource. For example suppose in an hour ending at 19:00, the day-ahead forecasting predicted about 1300 MW from the wind generation while the actual output is as 400 MW during the operation hour. The shortage of 900MW may lead to severe reliability problems. Due to the geographic scale of MG, the prediction for the renewable generation in a MG are particularly difficult thus the forecasting errors are more significant than those for bulk power system. This kind of uncertainties brings considerable challenges for the resource scheduling of MG operation.

Another challenge faced by MG operation is the mismatch between renewable generation and load pattern. This phenomenon is observed more in wind power and solar power production. In MG, similar type of mismatch occurs due to the same reason since the penetration level of renewable generation in MG might be even higher than bulk power system. This type of mismatch can lead to operational challenges when MG trying to integrate large amount of renewable energy like wind and solar energy. Consequently, the utilization factor of renewable energy resources would be jeopardized. Also, this type of mismatch also shows uncertain behaviors. It can be difficult to fully capture the mismatch in the day- ahead forecasting results [14].

Moreover, MG operation is challenged to optimally participate in the deregulated market under the current market framework. For a grid-connected MG, it is assumed as a 'modern citizen' from the bulk power system point of view [5]. Consequently, there are numbers of opportunities for MG to participate in the power market. For example, in current ERCOT market the offer price cap keeps increasing since the nodal market opened in 2010 [15], The optimal operation scheme should be developed to help the stakeholders of MG harvest the maximum economic benefit by strategically participating in the power market.

1.3 Contributions of the Presented Work

The research work in this thesis starts by addressing MG interconnection issues under renewable power generation uncertainties. And after addressing the issues a better solution for MG generation scheduling problem under renewable power uncertainties is achieved. The main contributions of thesis are follows:

- (i) On the basis of literature review pertaining to MG operation under renewable power uncertainties, an overview of MG's generation scheduling problem along with associated issues is presented. The detail study helps to understand the considered problem in MG operation under penetration of DERs in the distribution system.
- (ii) On part of research work contributes by formulating MG generation scheduling problem for both modes i.e. grid connected and isolated mode in robust framework considering renewable power uncertainties. The proposed formulation help to

understand applications of robust optimization based approaches in the MG operations.

- (iii) Wind and solar power are two major sources of uncertainties faced by the MG operator during the scheduling of generation in MGs. These uncertainties are modeled through interval forecasting using a Time series based Autoregressive Integrated Moving Average (ARIMA) model in the presented thesis.
- (iv) Robust optimization based approach is developed for optimal MG scheduling considering renewable power uncertainties.
- (v) Proposed algorithms and approaches are illustrated through realistic case studies.
- (vi) Wind and solar power uncertainties affect the optimal MG generation scheduling in deregulated power system. Therefore, comparative study is presented to evaluate the impact of wind, solar and both uncertainty individually and simultaneously on the optimal operation of MG in both grid connected and isolated mode.
- (vii) A comparative study is presented to compare the robust optimization approach with traditional stochastic and deterministic approach.
- (viii) An analysis is presented to evaluate the impact of degree of robustness on the optimal generation scheduling obtained through robust optimization based approach.

1.4 Organization of the Thesis

Chapter organization is an important part of thesis that provides a complete overview of the thesis. Current chapter introduces major issues involved in microgrid interconnection to distribution system. It analyses the involved problems in this area that motivated the work, and further contributions of the presented thesis. The rest of chapters of this thesis are organized as follows:

Chapter 2 provides a comprehensive literature review on issues pertaining to generation scheduling in distribution system, including those of uncertainty modeling like wind power uncertainty and solar power uncertainty and robust optimization approaches. It offers details on causes, models and solutions approaches of the associated issues.

Chapter 3 provides an overview of uncertainty modeling approaches and robust optimization framework. This chapter deals strength and limitations of different approaches for modeling uncertainties and formulating decision making problems in robust framework.

Chapter 4 presents algorithms for MG generation scheduling considering wind power uncertainty in robust optimization framework. Proposed algorithms are illustrated through different case studies based on MG operation. Comparative analysis helps to validate the proposed algorithms for MG generation scheduling considering wind power uncertainty.

Chapter 5 proposes a robust optimization based generation scheduling solution for MG operation considering solar power uncertainty. Proposed algorithms are illustrated through different case studies based on MG operation. Comparative analysis helps to validate the proposed algorithms for MG generation scheduling considering solar power uncertainty. The impact of degree of robustness is also carried out on the proposed algorithm.

Chapter 6 enhances the proposed robust optimization based MG generation scheduling algorithm by modeling multiple uncertainties in single problem. Here both solar power and wind power are modeled in uncertain environment simultaneously. And proposed algorithm is tested on MG generation scheduling problem considering multiple uncertainties.

Chapter 7 concludes the main findings of the work presented in this thesis and suggests directions for future research in this area.

Chapter – 1

Chapter 2

Literature Review

HIS chapter begins with fundamental thought of MG operation and underlining the major issues involved in MG operation in distribution system. Further, uncertainty characterization in renewable power generation in stochastic and RO framework and identified research challenges are discussed.
CHAPTER – 2

LITERATURE REVIEW

2.1 Introduction

Increasing penetration of renewable energy resources in evolving deregulated power system has great public attention, recently. Among the renewable energy sources, the penetration of wind and solar generators has increased due to their much evolved technology and wide spread availability. To integrate these renewable sources as Distributed Energy Resources (DERs) into the utility grid, Micro-Grids (MGs) play an important role [1]. MG is a small-scale power supply network that is designed to provide power for a small community and can increase the reliability and economics of energy supply [2]. A typical architecture of MG is shown in Fig. 1, which consists of dispatchable, non dispatchable generating unit and consumers. MG can be operated in two modes: grid connected and grid isolated. In grid-connected mode, MG can export its excess power to the grid and can import its deficit power from grid. MG can manage its demand independently in grid-isolated mode [3].

Due to uncontrollable nature of demand and to operate the generators within its technical limitations, the efficient and economic management of the generators is highly recommended. With increasing penetration of renewable sources like wind and solar in the distribution power system, MG management system may become complex. The grid price volatility in deregulated environment of power system has further increased the complexity of MG's energy management system. Therefore, development of efficient approaches for MG management under different uncertainties is necessary.

This chapter presents a theoretical background and review of literature pertaining to challenges for MG operation in distribution network. Initially, a fundamental background on MG operation in terms of generation scheduling problem has been provided. Next, issues pertaining to renewable power uncertainty (wind power and solar power uncertainty) have been analyzed. This is further supported by a discussion on uncertainty characterization, which is present in wind power and solar power through interval forecasting. Modeling of such uncertainties can help MG operator in optimal generation scheduling for total cost of MG operation minimization. Approaches developed to model MG generation scheduling problem considering renewable power uncertainties in competitive distribution system have been discussed to identify the research gaps.

Since uncertainty present in renewable power generation, it is necessary to model uncertainty in an effective manner to solve MG generation scheduling problem. Thus, the last part of the chapter discusses Robust optimization based approach for uncertainty characterization and associated potential issues to model uncertainty through robust optimization based approach.

2.2 Microgrid Operation

As MG can be defined as a small-scale power supply network that is designed to provide power for a small community. A typical MG consists of dispatchable units, non dispatchable units and consumers. MG can be operated in two modes: grid connected mode and grid isolated mode. In grid-connected mode, MG can export its excess power to the grid and can import its deficit power from grid. MG can manage its demand independently in grid-isolated mode [1-3].

Benefits of Implementing Microgrid (MG)

It can be summed up from the designing principle and all those on-going and demonstration projects that MG brings lots exciting benefits and opportunities. Those merits of implementing MGs are summarized as the following.

The dependency on imported fuel sources can be reduced by implementing MG and prime fuel market competition can be regulated easily. The increased interest in DERs, which includes multiple types of energy sources, has the potential to influence the market and competition level for prime energy sources. Developing countries like India and China have initiated a rapidly moving production infrastructure of energy that records a higher growth in global fuel markets.

Current MG technologies can able to compete with the traditional generation facilities with the help of local government encouragement and public support following with the dynamically changed fuel prices and environmental concerns over global warming. Available local prime fuel sources can be used to fuel up profitable MGs instead of imported ones. The MG market thrust is obviously driven by positive impact of environment by using local resources along with the economic benefits.

Further, implementing MG can improve the utilization of renewable generation resources. Although renewable energy conversion systems are having their intermittent energy nature and minimal environmental impact, most of developing countries like India and China have an abundance of natural renewable resources, that can be definitely integrated to meet a portion of the demand. Due to its innovative architecture and intelligent interface, MG can be a good entity to integrate high penetration level of renewable energy. Also, the grid-connected MG, which in cooperated with ESS and responsive load, can make the power flow on the point of common coupling (PCC) smoother thus reduce the generation fluctuation caused by intermittence nature of renewable generation from the utility point of view.



Fig. 2.1 Typical architecture of MG

Moreover, implementing MG can be able to enhance the demand side management (DSM) considering the deployment of modern communication system and advanced meter infrastructure (AMI). MG is capable of optimally managing its demand side without requiring further investment. The AMI system installed in the MG can be a useful tool for load pattern recognition and demand forecasting [16]. Also, the relatively small scale makes MG operators easier to categorize different load groups based on their load

characteristics and price responsiveness. In such a manner, the managing of demand side for MG can be further improved.

Furthermore, implementing MG can be able to defer the transmission lines construction or extension. MG can be the best option to support peak period demand. Therefore, the need of peak generation unit installation can be eliminated and security benefits are enhanced [16]. MG can also be useful in nearby commercial and industrial areas to meet the heat and electricity demand. It is helpful to relieve the centralized power plant from long distances power dispatching.

Last but not the least, implementing MG can help rural electrification, especially for under-developed areas. Currently, electrification resolution is needed in many developing and third world countries like as India and China. There are Over 1.2 billion peoples without electricity access and more than 2.8 billion people are dependent on solid fuels for their daily life activities like heating and cooking. Further largest percentage of rural darkness is present in African and southeastern countries [17]. In such type of rural darkness countries a very less resources to build large centralized generation plants or transmission infrastructure are present. In these countries MG can be considered as an important solution with availability of local renewable resources like ad wind power and solar power. As incorporated with modern EMS, dispersed MGs can be the best solution with less financial burden to meet out the local demand in that particular area of rural communities.

Due to uncontrollable nature of demand and generator's technical limitations, efficient and economic management of the generators is highly desirable. With increasing penetration of renewable sources like wind and solar in the distribution power system MG management system may become complex. The grid price volatility in deregulated environment of power system has further increased the complexity of MG's energy management system.

When a MG has more than two DERs, the energy management system (EMS) is needed to impose the power allocation among DER, the cost of energy production and emission. The EMS in a MG is shown in the Figure 2.2. As can be seen from this Figure, the forecast values of load demand, the distributed energy resources and the market electricity price in each hour on the next day are denoted as inputs. Furthermore, the operation objectives are considered to optimize the energy management, are given as follows:

- Economic option
- Technical option
- Environmental option
- Combined objective option.



Fig. 2.2 The energy management system (EMS)

2.3 Critical Review

This section represents the critical review on generation scheduling in MG operation, uncertainty characterization in both stochastic frame work and robust framework .

2.3.1 Generation Scheduling in Microgrid Operation

In recent studies several approaches have been proposed for MG's energy management considering renewable resources. Most of them are based on DERs scheduling and control, including renewable generation in MG. A simple optimization based approach has been used for MG's operation considering different market policies [9]. A Lagrangian Relaxation along with genetic algorithm approach has been incorporated for optimally scheduling of generators in isolated MG system [10]. Agent-Based heuristic approach has been used for MG management [18]. In above studies, uncertainty involved in renewable power sources is modeled deterministically. Due to lack

of accurate forecasting models, such type of uncertainty modeling is not reasonable for MG management.

Some algorithms for the optimization of MG energy management are proposed [9-27]. An ANN model has been used to solve optimal MG generation scheduling problem to minimize operation and emission cost considering renewable power and demand uncertainties [11, 12, 13]. For islanded MG operation optimal Energy Management system (EMS) has been presented in [19], by using a rule-based management. The MG operation depends upon developed rules, therefore, constraints are always satisfied but the optimization does not contain global results. Further to improve rule-based technologies in MG operation, fuzzy logic based estimated rules are proposed in [20]. Although, linear programming (LP) and mix integer linear programming (MILP) based technologies used to solve MG operation in [21] gives good results but the main limitation is known as the need of a specific mathematical solver. Quadratic programming (QP) based approaches have been addressed for optimal EMS in grid connected power system having Solar-PV/battery and electrical vehicle system in [22]. The results obtained through QP approach are better; however this method has a limitation of convex objective function. In [23], MG's optimal EMS is solved by using Game Theory and multi-objective optimization. The operating cost and the emission level are given as two objectives functions. Optimization of MG's total operating cost function has been achieved by Mesh Adaptive Direct Search (MADS) algorithm in [24], while to search the optimal generation schedule in MG operation MRC-GA (matrix real-coded genetic algorithm) is proposed in [25]. Further, the PSO (Particle Swarm Optimization) techniques [26] and dynamic programming (DP) approaches and advance dynamic programming (ADP) approaches [27] has been developed to optimize EMS in MG operation.

All of the above studies focus on MG operation and consists of algorithms that are used to solve the MG generation scheduling optimization problem. However these methodologies are having some limitations for solving optimization problem in MG operation, but this has provided better platform to MG researchers to overcome these critical issues of MG operation such as uncertainties involvement in renewable generation and demand side management etc.

2.3.2 Microgrid Operation Considering Renewable Power Uncertainties

Over the last couple of decades, worldwide power systems have experienced high growth in renewable power generation. Among the available renewable power generation technologies, generation from renewable sources is increasing rapidly due to its widespread availability and mature technology. Thus, renewable power is an encouraging alternative to fossil fuel power generation for mitigating climate changes. Availability of wind generation depends on atmospheric and geographical conditions. Due to inherent randomness of these conditions, renewable power like solar and wind power is highly uncertain [28]. Due to the intermittent nature of these renewable energy sources, their huge penetration may affect the reliable and secure MG operation. So accurate prediction of renewable power such as solar and wind power generation is required for secure and reliable operation of MG. However, prediction of Solar and wind power is never perfect, thus modeling of solar and wind power uncertainty is important for MG operators or system operators [28].

The main challenge for MG operator is to manage renewable generation uncertainties and demand uncertainties. Therefore it is necessary for MG operators to consider the effect of these uncertainties while determining the spinning reserve (SR) for protection against sudden change in generation and load [29]. In some literature MG reserve requirement has been determined without considering the probabilistic behavior of renewable generators before the scheduling in MG operation [30]. This type of approach is known as deterministic approaches in MG operation. To minimize total operation cost of MG operation quantum genetic algorithm [31] has been introduced without considering the uncertain nature of renewable generation. Further with consideration of probabilistic behavior of renewable generation and demand, stochastic methods are proposed for getting optimal solution in MG operation [32-33]. In stochastic methods renewable generation uncertainties are modeled through a number of scenarios [11]. It is evidenced in literature that stochastic methods have lower operating cost in comparison to deterministic methods.

Wind power and solar power are characterized by location constraints, low marginal cost, generation variability, limited dispatchability and difficulties in available resource forecasting [34-37]. Due to these, wind power and solar power integration in existing MG strongly affects its economic operation. Economic challenges relevant to MG operation include total operating cost MG generation scheduling problem and demand response [38-44].

Despite these problems, solar power and wind power generation are rapidly growing due to consistently decreasing cost of solar power and wind power generation and continuously improving accuracy of available forecasting tools. MG's generation scheduling problem can be treated as a decision-making problem due to uncertainty of wind power and solar power availability and load demands [45-48]. Modeling of involved uncertainties is necessary to solve these decision-making problems. Uncertainties are generally modeled using deterministic forecasting models, which focus on the use of point prediction. Point prediction provides single value of expected wind power or market price for a given lead-time, without any information about the possible deviation from forecasted value. Use of deterministic forecast for modeling decision-making problems may not provide the desired accuracy in solutions [49-52].

Decision-making problems are generally formulated using stochastic optimization. In this formulation, randomness of input parameters can be described by stochastic process. Stochastic process can be characterized via scenarios. Scenarios are possible outcomes of random input with corresponding occurrence probabilities [53,54].

As a large numbers of scenarios are required for accurate modeling of any stochastic process. Therefore, the computational efforts are required to reduce the original scenario set to solve scenario-based optimization models. Reduced scenarios are obtained in such a manner that the reduced set has a smaller number of scenarios, with minimally changed statistical properties [55-58].

Based on recent studies, scenario generation methods can be classified as: movement matching [59-60], conditional sampling [61] and path-based methods [62-65]. Movement matching and conditional sampling generate scenarios, which are based on direct sampling from distribution of stochastic process. Path-based methods generate scenarios using different statistical forecasting models such as ARMA and ARIMA. For a given lead-time, scenarios can be generated by repeated simulations of these forecasting models, with different confidence intervals.

These methods can be used for load, price, and wind speed scenario generation [66-68]. Monte Carlo simulation with path-based scenario-generation method has been used for generation of electric load scenarios in power management problems [66]. The wind speed scenario-generation method can use statistical time-series ARMA model for scenario generation of different sites [67]. This approach shows the relation of generated scenarios by autocorrelation plots between every two sites. Thermal rating of transmission lines have been forecasted by the time-series ARIMA model, considering the effect of weather uncertainty [68]. Bidding strategy problem using a stochastic programming approach is simulated by generating price scenarios of the Nordic market using ARMA model [69].

For wind power scenario generation, all methods assume that wind power forecasting error density function follows Gaussian or normal distribution. Statistical analysis shows that due to variable kurtosis, wind power forecasting error probability density function cannot be characterized by normal distribution [70]. Therefore, this assumption is not valid for practical applications. Generated scenario set is assumed to be stable, if several scenario sets are randomly obtained using a method to provide the same solution for a considered decision-making problem. Quality of scenarios is defined in terms of small bias that can crop up between true value and solution of scenario based optimization. A consideration of stability and quality of generated scenario set can improve modeling of wind power uncertainty [71].

To reduce computational burden for solving decision-making problems through scenario based optimization models, generated scenarios need to be reduced *via* scenario reduction techniques. Scenarios can be reduced by using fast forward selection and simultaneous backward reduction methods [72-75]. These methods can be used for reducing electric load, price and wind power scenarios. Forward and backward scenario-reduction algorithms have been proposed for price and load scenarios [74]. Recursive backward scenario-reduction technique has also been used for wind scenario reduction [76]. These methods are iterative and their mathematical formulation is complicated. Iterative elimination of scenario requires large computational time, when the preserved scenarios are mini scale as compared to generated scenarios.

Further an efficient and simplified algorithm for scenario generation and reduction has been proposed for application to multi-stage stochastic programming [77]. The scenariogeneration algorithm is based on a time-series ARMA model, while the reduction algorithm utilizes the concept of probability distance. Both algorithms are implemented for next-day scenario generation of a wind farm located at Barnstable, Massachusetts, USA. The results clearly showcase the strength of the proposed algorithms in wind uncertainty modeling. Owing to quantum scenario reduction, stochastic properties of the generated scenarios are maintained with minimal variation in the mean and standard deviation.

For solving MG generation scheduling problem a smart energy management system (SMES) has been presented in [78], which considered PV output with different weather

conditions and different hourly prices of main grid. Here reserve allocation is not done for renewable power generation uncertainty and load participation is also not considered. Further rolling horizon (RH) based strategy for renewable energy based energy management system (EMS) has been proposed in [79], which minimizes the total operational cost of MG. In [80] MG operation for generation scheduling has been done by considering economic and emission objectives, where to minimize the cost function, a mesh adaptive direct search algorithm has been proposed, but here wind power and solar power forecasting errors and demand side participation in energy market have not been considered. For economical MG operation, application of HRDS (High Reliability Distribution System) has been considered. HRDS has been applied to loop networks in distribution system which has higher operational reliability and fewer outages in MG.

In [81] spinning reserve requirement is ensured by estimation model, where wind power and solar power generation uncertainty, unreliability of units and uncertainty caused by load demand are considered. Various uncertainties are aggregated to reduce computational burden in this approach. Further MG generation scheduling problem is optimized in consideration of multi-period islanding constraints. Where, the objective of MG generation scheduling problem is to minimize the total operation cost of MG operation. For decoupling of grid connected and grid isolated problem bender decomposition method has been employed. Here MG components such as generating units, loads, and energy storage system etc. are modeled through mixed integer linear programing (MILP) approach. In consideration of renewable energy generation uncertainties and demand uncertainties in MG operation a fuzzy logic based EMS (Energy Management System) has been proposed, which solves the cost and emission minimization problem. Further MG operation in stochastic framework has been proposed in grid connected mode operation of MG operation, which considers the uncertain nature of wind turbine generation, load forecast error, PV generation and market prices. Uncertainties of these uncertain parameters have been modeled by generating a number of scenarios through roulette wheel mechanism. However, the reserve scheduling has not been considered in this methodology.

Another stochastic programming based approaches for MG operation considering renewable power uncertainties and network constraints has been developed in [82-83]. In this approach, uncertainties are modeled through different scenarios. A large number of scenarios are required for accurate modeling of any uncertainty. However, with large

18

number of scenarios, the size of optimization problem drastically increases leading towards computational complexity. Additionally, stochastic programming approaches are based on knowledge of distribution of historical data of uncertain parameters. If the knowledge of the distribution of historical data is unavailable, the approach could lead wrong results.

2.3.3 Robust Optimization Based Uncertainty Characterization

Robust Optimization (RO) is an approach which is used to protect decision maker against data uncertainty and parameter ambiguity. At a high level, decision maker must have to ensure that solution is a robust solution in such a way, whose feasibility can always be guaranteed for any type of realization of uncertain parameters. To obtain robust solution the main approach relies on worst case analysis in which a solution is obtained when realization of uncertainty is most unfavorable.

In RO to compute the worst case is always a challenging task to researchers when uncertainty is involved. In some of literature the worst case realization uses a finite number of scenarios of uncertain parameter such as convex/continuous uncertainty set or historical data. While some literature states that worst case realization must be chosen in such a way that it should be neither too small nor too large *i.e.* it will trade off between system performance and protection against uncertainty.

2.3.3.1 Static Robust Optimization

In static RO framework, decision maker take decision on the basis of uncertainty. And after realization of uncertainty no recourse action would be possible. Therefore two types of uncertainty like uncertainty on feasibility of solution and uncertainty on objective value must be distinguished. Regarding infeasibility and sub optimality, decision maker have different opinions which justifies the need of separate realization of these two settings.

(a) Uncertainty on Feasibility

When solution feasibility is affected by uncertainty than RO seeks to obtain a feasible solution at any realization of uncertain parameter by unknown coefficients. However, in objective function the complete protection from adverse realization is rare. In some engineering applications of robustness this approach may be justified properly but in operational research it is less reliable because in operational research low customer demand adverse event may not be able to produce high profile repercussions. Robust methodology in MG operation may become an important solution if it will produce feasible solution for any kind of realization of an uncertain parameter for unknown coefficient within an uncertainty set. Uncertainty set is nothing but it is the smaller realistic set which belongs to uncertain parameter. The specific realization of an uncertain parameter within uncertainty set must be taken carefully by decision maker because it plays an important role for ensuring computational tractability of robust problem.

In recent literature RO based approaches are used by considering worst case optimization in which worst case is being determined on the basis of convex uncertainty set of an uncertain parameter. Convex uncertainty set is nothing but it represents the limit which bounds the maximum allowable deviation from nominal value of an uncertain parameter. In worst case optimization approaches the main objective is to derive the tractable reformulation which is able to provide optimal solution along with probabilistic guarantee against constraint violation. In order to understand RO a unified treatment under data uncertainty has been studied in [84], where convex inequalities of system represent the uncertainty set. Further heuristic based approaches have been investigated in [85] to model uncertainty, which couples RO with simplified stochastic approaches. Results obtained through this approach exhibit high flexibility towards optimization and simplicity of uses.

Further an optimization model has been presented in [86], which represent robust counterpart as a convex quadratic constraint, which is having ellipsoidal implementation error equivalent to quadratic conic constraint. The difference between local and global robustness has been analyzed in [87]. Global robustness is represented as 'Black Swans' which is the severe uncertainty faced by decision maker. Further safe tractable algorithm based on RO has been presented in [88], where approximations of chance constraints under data uncertainty are satisfied with high probability functions.

(b) Uncertainty on Objective Value

RO may be used to obtain a solution in such case when solution optimality is affected by uncertainty. In such cases results obtained from RO based approaches are good for any realization of an uncertain parameter within the uncertainty set. In RO based approaches generally worst case realization is to be optimized. Uncertainties on objective function can be better approached by worst case realization techniques by creating a new inequality set in such a manner that brings the objective function in feasibility sets. In [89] cost uncertainty, which present in objective function, has been considered with linear robust optimization. The problem is solved with single robustness by losing the objective value. Further multi-objective RO has been investigated [90], where the assessment of robustness has been done through evolutionary algorithms within multi-objective optimization. Further multi expert multi criterion robust weighed approaches are presented in [91], which uses robust pareto decision to minimize worst case weighted sum in problem.

More over a new robust criterion is proposed in [92], which seems great due to its simplicity of use and practical relevance. In this method, which is also known as *bw*-robustness, the decision maker is allowed to identify guaranteed solution of an objective value. Here maximization problem has been considered and constraint condition has been created by taking two variables 'b' and 'w' in such a way that at least 'w' is present in all scenarios of uncertainty set and it maximizes the probability of reaching at 'b', which is targeted value provided 'b>w'. Further this criterion is extended in [93] by using interval in uncertainty modeling instead of finite set of scenarios. Laxicographic robustness has been presented in [94], which is defined over finite set of uncertainty scenario. Here the primary role of worst case realization has been mitigated. Further, RO framework considering combinational optimization has been proposed in [95], which uses the dempster-shafer theory with portfolio tracking and sensor placement.

2.3.3.2 Multi-Stage Decision-Making

The earliest work on RO approach is based on static decision making framework, where decision makers are authorized to take decisions at once of all values that are take by all decision variables. And if multiple decision stages are incorporated in robust problem then problem is solved by incorporating the decision taken of only current stage. Since, it is difficult to incorporate multi stage decisions in RO framework, so most theoretical approaches are based on two-stage decision RO framework. In [96], two stage robustness based tractable algorithm has been developed with minimization of worst case regret in both scenario and interval based uncertainty set. Further Kelley's algorithm based cutting plane method has been proposed in [97] with adjustable convex robust optimization problem. Additional preliminary results based on two stage robustness with robust linear program and equivalent second order conic program have been presented in [98]. With considering right hand side uncertainty two stage linear robust problem has been analyzed in [99].

Further, for a class of multi-stage chance constrained LPP, a tractable approximation has been proposed in [100]. In this method, original optimization problem has been

converted into second order conic optimization problem. RO based framework has been described in [101] to extend affinely adjustable robust counterpart framework. In [102], one dimensional RO based has been described, which established the optimality of policies within uncertainty set. This methodology further explored in [103] by incorporating polynomial policies. To observe uncertainty parameter polynomial policies have been parameterized and after parameterization of policies optimal solution can be efficiently calculated by semi definite optimization methods.

Further decision rule policies have been considered in [104]. These policies are based on stochastic framework for both primal and dual version of problem. Here loss of optimality is analyzed with RO approximation approach. Further min max approach has been presented in [105], where dynamic policies are characterized by min max algorithm and discrete dynamic system is perturbed by noise. In continuation with multistage approaches on RO framework, adjustable multistage RO dynamic problem has been presented in [106]. Convex multi stage RO problem has been considered in [107], which approximate the decision of decision maker with finite linear combinations of prescribed uncertain functions. Here it is described how constraint randomization is utilized to optimize problem at low computation time and cost.

Further considering limited information of distribution process constrained stochastic linear system has been investigated in [108], in which only first two moments of distribution are known. Here two types of distributional robust constraints are used, one is the constrain that is come through known moments of given probability for all disturbance distribution and another one is CVaR (Conditional Value at Risk) constraint. CVaR constraint is used to bind expected constraint violation with given moment information for all disturbances consistent. A new deviation measure method has been used in [109] to construct uncertainty set for RO. The asymmetry of the distribution can be captured by these deviations measures. These are also known as improved approximations for chance constrained problem. In [110] uncertainty set formulation in Robust LP framework has been described with coherent risk measures. Further, for risk measures in finance, uncertainty set has been constructed in [111] with consideration of opposite perspective. For generating coherent risk measures a specific approach has been proposed in [112] with risk adverse optimization and developed sufficient and necessary condition for optimality of RO approach in convex set. Further considering smoothness assumptions necessary sufficient conditions are derived in non-convex uncertainty set with ambiguous and

inconsistent utility assessment of a decision making framework has been investigated and pareto efficiency to RO approach has been incorporated in [113].

Further RO formulations to non-linear problems have been provided in [114], which is valid in robust approach to first order and nominal parameter. Further robust non-convex optimization approach has been presented in [115] with arbitrary objective function. It will move iteratively in descent directions with termination at robust local minima. Uncertain quadratic programing based conic quadratic constraint has been analyzed in [116] by formulating approximate robust counterpart. Robust geometric programing with tractable approximations has been studied in [117], where pricewise linear convex approximation is used for each non-linear constraint. In [118] geometric programing has been investigated with injected robustness at algorithm level and seeks to get feasible solution.

Further Robust convex optimization has been proposed in [119], where optimization problem is solved with passimizing oracles. Robustness for uncertain convex quadratic programing problem has been discussed in [120] with consideration of ellipsoidal uncertainties. Here realization technique with consideration of robust deviation random sampling has been utilized. Further unconstrained non-convex problem has been considered in [121], which uses arise in partial differential equation. Here any specific structures are not assumed to solve the problem. Further with considering constraints non-linear problem is solved in [122].

In [123] a special type of optimization approach has been formulated, which has linearity in decision variables and convexity in uncertainty. Here worst case has been maximized and exact & tractable robust counterpart has been derived. Further a special class for non-convex set of uncertain parameter in RO has been analyzed in [124]. RO based approach which uses intervals of uncertainty, for both non convex and convex quadratic programing problem has been discussed in [125], while min max and robust model for polynomial optimization has been used in [126].

Further demand uncertainty in capacity vehicle routing problem has been considered in [127] with uses of interval of uncertain parameter. Robust salesman problem has been presented in [128]. Combinational robustness has been utilized in [129] for different bidding behaviors of combinatorial auctions. Min-max regret and two stage analysis has been done in [130], which consider a solvable polynomial under uncertain demand interval sets. Further in transportation services combinatorial auctions have been considered in [131] with taking shipment as uncertain variable.

Further robustness has been analyzed in [132] by considering scheduling stability measures. Scheduling has been done through tabu search algorithm. Discrete time/cost trade off problems for robustness measurement in robust scheduling problem has been considered in [133]. With non-linear objective functions an efficient and effective algorithm has been developed that gives most nearer optimal solution [134]. Network flow under travel time uncertainty and demand uncertainty has been considered in [135] as robust capacity expansion problem. Here tractable reformulation has been provided with considering broad set of assumptions. Further demand uncertainty has been considered in network flow and design problem [136]. Here two stage network topology is considered to solve problem. Considering transportation cost and demand uncertainty in network design a tractable approximation has been discussed in [137] for each commodity at single origin and destination.

Further facility location problem has been discussed in [138], where optimal location and optimal time both are determined. Optimal time is determined to establish capacitated facility with consideration of demand and cost variation. Problem is solved by heuristic approach like bender decomposition method or local search algorithm. Robust network design has been proved in [139] with uncertain parameter. Deviation in robustness has been analyzed in [140]. Further RO has been applied in location facility problem for network design over multi period in [141], where different uncertainty model are used in different network topologies. A dual based local search algorithm has been investigated in [142] for all deterministic, stochastic and robust study. Further robust version of transportation problem has been discussed in [143], where local transportation problem with uncertain demand has been reformulated using two stage formulation and non-linear convex cutting plane method.

Further RO in energy systems has been discussed in [144] with investigation of two stage unit commitment problem where demand uncertainties have been considered and tractable solutions are obtained. Strategic planning model has been proposed in [145] for oil supply chain problem. Further integer programing model has been proposed in [146] for robust power grid optimization problem. Two stage adaptive robust optimization model has been proposed in [147], which uses security constrained unit commitment problem. Here nodal net injection uncertainty has been considered and bender decomposition method is used. Further daily operation of pumping station has been optimized in [148] considering demand uncertainty with the help of adaptive RO.

RO applications in RE (Renewable Energy) field have been investigated with considering wind and solar power uncertainties. Robust unit commitment has been proposed in [149], which uses thermal generators scheduling and minimizes the total operational cost under wind power uncertainty. However results obtained through this method are some where tractable but represent confusions on selecting sites and larger system is unable to give feasible solution.

2.4 Research Challenges and Objectives

On the basis of critical review of literature pertaining to approaches or algorithms that have been used for modeling uncertainties through scenarios, following challenges are identified. Existing algorithms for scenario generation as well reduction has been used for modeling electric load and financial market uncertainties and their application need to extend for modeling wind power and electricity market price uncertainties. Additionally, existing algorithms are complicated and not well explanatory, thus need to be simplified for modeling wind power and solar power uncertainties. Quality and stability of generated scenarios needs to be considered for improving uncertainty modeling.

Renewable power integration in MG operations in deregulated environment poses several challenges for MG operators. Development of efficient generation scheduling algorithm for MG operator under uncertainties is a major challenge considered for further investigations in this thesis. Excellent bibliographical surveys and research reviews on such challenges published in recent years indicate that MG generation scheduling under uncertainty environment is a potentially interesting decision-making problem for MG operators in distribution system. Uncertainties of renewable power have generally been modeled through a number of scenarios, which are generated and reduced using different approaches. On the basis of critical review of literatures pertaining to considered problem, following challenges are identified.

- I. Development of MG generation scheduling problem in both grid connected and isolated mode.
- II. Development of wind and solar power uncertainty in the MG generation scheduling problem using a Time series based Autoregressive Integrated Moving Average (ARIMA) model.
- III. Development of robust optimization based approach for optimal MG scheduling considering renewable power uncertainties.

IV. Impact of renewable power uncertainties on optimal MG generation scheduling in deregulated power system.

On the basis of research challenges following objectives are created in this thesis

- (i) To formulate MG generation scheduling problem in both grid connected and isolated mode.
- (ii) To model wind and solar power uncertainty in the MG generation scheduling problem using a Time series based Autoregressive Integrated Moving Average (ARIMA) model.
- (iii) To develop robust optimization based approach for optimal MG scheduling considering renewable power uncertainties.
- (iv) To evaluate the impact of renewable power uncertainties on optimal MG generation scheduling in deregulated power system.
- (v) To compare the robust optimization approach with traditional stochastic and deterministic approach.
- (vi) To evaluate the impact of degree of robustness on the optimal generation scheduling obtained through robust optimization based approach.
- (vii) To illustrate the proposed work through realistic case studies.

2.5 Summary

This chapter presents an overview of MG operation in distribution system. Further working modes of MG operation are discussed. The benefits of implementing MG have been discussed. The critical review on MG operation, uncertainty characterization in MG operation, uncertainty characterization in stochastic framework and robust optimization based uncertainty characterization have been carried out. On the basis of critical review research challenges are identified and on the basis of research objectives are made.

Chapter 3

Uncertainty Characterization

Т

HIS chapter presents the basics of uncertainty characterization. Different types of uncertainties presented in MG operation have been illustrated. Further basics of ARIMA model for modeling of an uncertain parameter have been discussed in non-stationary time series signals. Further on the basis of ARIMA model, ACF and PACF formulation has been discussed.

UNCERTAINTY CHARACTERIZATION

3.1 Introduction

Occurring phenomena in the quantitative analysis related to engineering applications is first developed with mathematical models which are converted into simulation computer codes. A mathematical model can represent a real system, which is dependent on a number of parameters and hypotheses. The mathematical model can be represented in either deterministic or stochastic way. If occurring phenomena's underlying knowledge is incomplete than under analysis system cannot be characterized. This represents the uncertainty in the analysis of system. The uncertainty in quantitative analysis is due to the (i) occurring phenomenon's intrinsic variability and (ii) lack of information or knowledge of occurring phenomena. Such uncertainties within the model can propagate, which results in intrinsic variability in output. Therefore, the quality and characterization of resulting output are different that defines the scope of uncertainty analysis.

The aim of uncertainty analysis is to determine the behavior of uncertainty in quantitative result analysis [33]. Analysis of uncertainty can be illustrated with introduction of a new model f(x), which is dependent on input quantities x as y = f(x). With uncertainty in x, the uncertainty analysis of model y can be computed with propagation of model f.

Generally, both the uncertainties i.e. uncertainty in data structure x and uncertainty related to model structure f, are treated separately due to the existence of plausible hypothesis. In recent literature the first source of uncertainty i.e. the uncertainty in data structure x is widely investigated and improved by researchers, while, the uncertainty in model structure f is less investigated. So to handle-out the uncertainty in model structure f, effective and agreed methods are to be required [33]. Therefore, uncertainty analysis is an unavoidable component to model occurring phenomena and to study the behavior of system under their limits of operation. The uncertainty analysis of any system can be done by incorporating the understanding of uncertain system components, behavior of uncertain data, appropriate characterization of uncertain data etc.

The characterization of data uncertainty set using *ARIMA* model is described in next section of this chapter. And uncertainty characterization through stochastic and robust

framework has been described in section 3.3 and 3.4 of this chapter. Fig. 3.1 shows the different possible uncertainties that may affect the smooth MG Operation.



Fig. 3.1 Possible uncertainties in MG operation

3.2 ARIMA Model

In a particular realization of time series analysis an ARIMA (Autoregressive Integrated Moving Average) model is defined as a generalization of an ARMA (Autoregressive Moving Average) model. To predict future points in time series analysis, these models (i.e. ARIMA and ARMA) can be fitted. The prediction of future points are also known as forecasting, which is used for better understanding of data. These models (ARIMA and ARMA) are applied in stationary time series models, while in some cases, where data time series model is non-stationary, the non-stationarity is removed by an initial differencing step. This step is corresponding to the "integrated" part of the non-stationary model. Therefore, before describing the ARIMA model, the concept of stationarity and the technique of differencing time need to be discussed.

(i) Stationarity

Any time series model can be said stationary if properties of that system do not depend upon time. The time series model with seasonality and trends cannot be considered as stationary time series models because the value of time series model with seasonality or trends will be affected at any time. Similarly white noise series can be considered as stationary as its value will not depend upon time. The value of white noise series model is similar at any time period. Some more confusing examples like cyclic behaviour time series model is considered stationary if there is no trend or seasonality. Since the length of cycle is not fixed, so before observing the series, it cannot be sure about peaks and troughs of cycle. In a stationary time series model no predictable pattern for a long term is observed. The time plot of time series model is considered of roughly horizontal with constant variance of time series models.

(ii) Differencing

Differencing is nothing, but it is the way to make stationary time series from the nonstationary time series to compute the consecutive observation differences. Differencing is used to stabilize the mean of a time series model. It is done by removing the time series model level changes to eliminate the present trend and seasonality.

Non-stationarity of the time series model can be identified with ACF plot as well as time plot of data. ACF (autocorrelation factor) in stationary time series is required to drop to zero relatively, while in non-stationary time series model ACF of data decreases slowly.

(iii) Second-Order Differencing

If the differenced data of time series model is again non-stationary than it is required to go forward for second order differencing to obtain a stationary time series:

$$y_{t}^{"} = y_{t}^{'} - y_{t-1}^{'}$$
(3.1)

$$= (y_t - y_{t-1}) - (y_{t-1} - y_{t-2})$$
(3.2)

$$= y_t - 2y_{t-1} + y_{t-2} \tag{3.3}$$

Where,

- y'_t Represents first differentiation of time series
- $y_t^{"}$ Represents second differentiation of time series

(iv) Seasonal Differencing

Seasonal differencing is nothing but it is used to represent the difference of current time observations to same time observations of previous year as shown in equation (3.4)

$$y_t = y_t - y_{t-s}$$
 (3.4)

Where,

s = number of seasons

These are also called "lag-s differences" as we subtract the observation after a lag of s periods.

3.2.1 Non-Seasonal ARIMA Models

If differencing is combined with AR (Auto regression) and a MA (moving average) model, an ARIMA model with non-seasonality is obtained. ARIMA is an acronym for Autoregressive Integrated Moving Average model ("integration" in this context is nothing but it is the reverse of differentiating). The full model representation of a time series model can be given as

$$y'_{t} = c + \phi_{1} y'_{t-1} + \dots + \phi_{p} y'_{t-p} + \phi_{1} e_{t-1} + \dots + \theta_{q} e_{t-q} + e_{t}$$
(3.5)

Where,

c is constant term, included in forecasting

 ϕ_i is AR parameters

- θ_i is MA parameters
- e_i is Error term

 y'_t represents the differenced series, which might have been differenced more than once. The "predictors" on the right hand side include both lagged values of y_t and lagged errors (e_i) . This is referred as an ARIMA (p,d,q) model where, p represents the order of the autoregressive part, d represents the degree of first differencing involved, and q represents the order of the moving average part.

Once combining of components is done in this way to form more complicated models, it is much easier to work with the backshift notation. Then above equation can be written as

$$(1 - \phi_1 B - \dots - \phi_p B^p)(1 - B)^d y_t = c + (1 + \theta_1 B + \dots + \theta_p B^q)e_t$$
(3.6)

Where,

B is backshift operator

d is differentiation operator

3.2.2 Seasonal ARIMA Models

Non-stationary series can be converted into stationary ones with regular differences, *i.e.*, the difference from one period with respect to the next. Similarly to remove seasonality seasonal differences can be used. Non-stationary time series model with seasonality can be converted into a stationary time series model by using equation (3.7) as

$$w_t = \nabla_S^D \nabla^d z_t \tag{3.7}$$

Where,

- D is the number of seasonal differences (if there is seasonality)
- D is 1; if there is no seasonality
- d is the number of regular differences $(d \le 3)$
- ∇ is differentiation operator
- z_t is non stationary time series model
- w_t is stationary time series

When seasonality in non-stationary time series model is present than the ARMA model can be generalized by incorporating both the differencing i.e. measurement intervals as well seasonal dependence of time series model separated by s seasonal periods.

Further it is important to discuss, how to model these two types of dependence:

- In first solution, seasonal dependence is incorporated in regular time series model by using B operator. In B operator AR or MA are incorporated to represent the dependence between observations separated by s periods.
- This formulation represents complexity in system solutions as it leads to very large polynomials in AR and MA. For example, if a monthly data with seasonality s = 12, is related to the same month in three previous years, then an AR or MA of 36 order is required to represent the seasonal dependence.

Therefore, it is needed to model a simpler approach for separately seasonal and regular dependence. The multiplicative seasonal ARIMA model is given as equation (3.8)

$$\Phi_P(B^S)\phi_P(B)\nabla_S^D\nabla^d z_t = \theta_q(B)\Theta_Q(B^S)e_t$$
(3.8)

Where,

 $\Phi_P(B^S) = (1 - \Phi_1 B^S - \dots - \Phi_P B^{S^P})$ is the seasonal AR operator of order P

 $\phi_P = (1 - \phi_1 B - \dots - \phi_P B^P)$ is the regular AR operator of order P

 $\nabla_{S}^{D} = (1 - B^{S})^{D}$ represents the seasonal differences and $\nabla^{d} = (1 - B)^{d}$ the regular differences

 $\Theta_{Q}(B^{s}) = (1 - \Theta_{1}B^{s} - \dots - \Theta_{Q}B^{sQ}) \text{ is the seasonal moving average operator of order Q}$ $\theta_{q}(B) = (1 - \theta_{1}B - \dots - \theta_{q}B^{q}) \text{ is the regular moving average operator of order q}$ *e*, is a white noise This class of models, introduced by Box and Jenkins (1976), offers a good representation of many seasonal series that this model can be written as(p,d,q) in simplified form as the ARIMA $model(P,D,Q)_s \times (p,d,q)$. Where, p,d,q are used for non seasonal arima model while P, D, Q are used for seasonal ARIMA model.

In above equation 3.8, to represent the seasonal part of model uppercase notations are used, while to represent the non-seasonal parts of the model lowercase notation are used. The modeling of seasonal part and non-seasonal parts are very similar except involving of backshifts of the seasonal period.

3.2.3 Statistical Features

Characteristics of time series model can be represented by adding the following additional features:

(i) Mean

Generally the mean is represented by \overline{x} . The mean is given by the arithmetic average of observation sets. It is mathematically calculated by the total sum of the observation sets divided by the number of observation sets as shown in equation (3.9):

$$\overline{x} = \frac{1}{n} \sum_{i=1}^{n} x_i$$
 (3.9)

Where x is the value of observation sets and n is the number of observation sets.

(ii) Mean Absolute Deviation

Variability is measured by the Mean absolute deviation (*Mad*). The *Mad* for a sample size n is defined as equation (3.10):

$$Mad(x_1....x_n) = \frac{1}{n-1} \sum_{i=1}^n (x_i - \bar{x})^2$$
(3.10)

Where, \overline{x} is the distribution mean

(iii) Variance

Dispersion is widely used by another term variance, which estimates the squared mean deviation of x from its mean value \bar{x} as sown in equation (3.11)

$$Var(x_1....x_n) = \frac{1}{n-1} \sum_{i=1}^n (x_i - \overline{x})^2$$
(3.11)

(iv) Skewness

The degree of asymmetry around mean is measured by skewness. The skewness gives the idea of variation data set's direction. There are two types of skewness, one is positive skewness and another one is negative skewness. Positive skewness tells that the distribution tail of the time series model is more stretched above the mean side, while negative skewness tells that the distribution tail of the of the time series model extends out below the mean side. In normal distribution of time series model, the skewness is treated as zero. Skewness is defined as equation (3.12)

$$Skew(x_1....x_n) = \frac{1}{n} \sum_{i=1}^n \left(\frac{x_i - \bar{x}}{\sigma} \right)^2$$
 (3.12)

Where $\sigma = \sigma(x_1, \dots, x_n) = \sqrt{Var(x_1, \dots, x_n)}$ is the standard deviation of the distribution.

3.2.4 ACF and PACF

The autocorrelation factor (ACF) at lag k is computed by equation (3.13)

$$ACF(k) = \frac{E\left[\left(y_{t} - E\left[y_{t}\right]\right)\left(y_{t-k} - E\left[y_{t-k}\right]\right)\right]}{\sqrt{Var\left[y_{t}\right]Var\left[y_{t-k}\right]}}$$
(3.13)

The relationship between Y_t and Y_{t-k} is to be measured in time series model when effects of other time lags are removed. Because correlation does not measure the effect of other time lags. To measure the effect of other time lags partial autocorrelation is used. ACF is a plot of correlation as a function of lag. The autocorrelation is simply the ordinary Pearson product-moment correlation of a time series with itself at a specified lag. The autocorrelation at lag 0 is the correlation of the series with its unlagged self, or 1. The autocorrelation at lag 1 is the correlation of the series with itself lagged one step, the autocorrelation at lag 2 is the correlation of the series with itself lagged 2 steps, and so forth. The partial autocorrelation factor (PACF) of a time series at lag k is denoted a_k and is found as follows:

$$y_t = \phi_0 + \phi_1 y_{t-1} + \phi_2 y_{t-2} + \dots + \phi_k y_{t-k} + e_t$$
(3.14)

Then $a_k = \phi_k$ represents the fitted value of ϕ_k from the regression (Least Squares).

At different lags, the set of partial autocorrelations is called the PACF (partial autocorrelation function) and is plotted like the ACF *i.e.* PACF is a plot of partial autocorrelations with functions of lag. The partial autocorrelation at a given lag is the autocorrelation that is not accounted for by autocorrelations at shorter lags.

Chapter – 3

The principle way to determine appropriate AR /MA model is to look at the ACF and PACF of the time series model. Table 3.1 shows the principle way to find out the AR /MA with shapes of ACF and PACF.

MODEL	ACF	PACF
AR(1)	Exponential decay: on +ve side if	Spike at lag 1, then 0; +ve spike if
	$\varphi_1 > 0$ and alternating in sign, starting	$\varphi_1 > 0$ and -ve spike if $\varphi_1 < 0$.
	on –ve side, if $\varphi_1 < 0$.	
AR(p)	Exponential decay or damped sine	Spikes at lags 1 to p , then zero.
	wave. The exact pattern depends on	
	the signs and sizes of $\varphi_1, \ldots, \varphi_p$.	
MA(1)	Spike at lag 1, then 0; +ve spike if	Exponential decay: on +ve side if
	$\psi_1 < 0$ and -ve spike if $\psi_1 > 0$	$\psi_1 < 0$ and alternating in sign, starting
		on +ve side, if $\varphi_1 < 0$
MA(q)	Spikes at lags 1 to q, then zero.	Exponential decay or damped sine
		wave. The exact pattern depends on
		the signs and sizes of ψ_1, \ldots, ψ_q .

Table 3.1: Shapes of ACF and PACF to identify AR or MA

3.2.5 White Noise

A stationary time series \mathcal{E}_t is said to be white noise if

 $Corr(\varepsilon_t - \varepsilon_s) = 0 \qquad \forall t \neq s \qquad (3.15)$

Where,

 ε_i = White noise term

 $t = \text{Time period } 1, 2, \dots, n$

s =Seasonality

Thus white noise can be a sequence of uncorrelated random variables with both constant mean and constant. White noise series plots are very jumpy, erratic and unpredictable in behavior. Since white noise is uncorrelated, hence future values of white noise cannot be forecasted on the basis of previous values. The results of successive spins of a roulette wheel provide an example of a white noise series.

In economic time series, the white noise series is often thought of as representing innovations, or shocks. That is, ε_t represents those aspects of the time series of interest which could not have been predicted in advance.

3.3 Uncertainty Characterization in Stochastic Optimization Framework

Uncertainty is subjected to most of decision-making problems due to the natural phenomena's inherent randomness nature or to the input information's inaccurate knowledge. So, to overcome the effect of uncertainty decision makers are eager to develop such methods and tools which are less sensitive to influences or imprecise data. Besides developing models, decision makers are simultaneously working for increasing profit, reducing cost and improving reliability. Most of the decision-making processes based on optimization problems, where a set of the decisions (control variables) is to be obtained to optimize a certain objective like maximization of profit or minimization of total cost. The range of possible solution values of decision can take the value within economical, physical, technical and environmental limitations. Fig. 3.2 shows the Computational framework for stochastic MG operation in deregulated environment.

The following linear optimization problem equation (3.16) without loss of generality is considered for modeling

$$\begin{array}{ll} Max. \ C \ x \\ S.t. \ A x \le b \\ l \le x \le u \end{array}$$
(3.16)

Where x is the decision variable vector, and matrix A and vectors b and c are input data. Solution algorithms for linear optimization problems can be solved by the famous simplex method, if the parameters A, b and c are perfectly known. The decision variable vector x is best suitably optimized by simplex method. Now assume that some of these input parameter data are dependent on a certain random vector λ with a realization of λ_{ω} . Mathematically, this can be written as $A_{\omega} = A(\lambda_{\omega})$, $b_{\omega} = b(\lambda_{\omega})$, and $c_{\omega} = c(\lambda_{\omega})$. If the decision variable vector x is required to be determined before the random parameter vector λ with realization of λ_{ω} in the decision-making process, then to determine the optimal solution for decision variable vector x of problem equation (3.16) will be more intricate in the following aspects:



Fig. 3.2 Computational framework for stochastic MG operation

- (i) Feasibility: First of all, it is the main question for decision maker to make completely satisfied constraint while the matrix A and vector b are the unknown parameters. And similarly another question is about of guarantee of feasibility of decision variable vector x in optimal solution under uncertainty. In decision making problems, the way of feasibility can be measured by using uncertainty modeling approach.
- (ii) Optimality: If in above problem equation (3.16) uncertainty parameter is cost vector c, then objective function in equation (3.16) may not be a real-valued function. Now equation (3.16) becomes a random variable's family as $f(x, \omega) = C'_{\omega} x$. To get feasible decision vector $x, f(x, \omega)$ can be obtained as a function of random variable $f(x, \cdot)$ at an argument of ω . Hence, decision variable vector x is found through optimization under uncertainty, that belongs to the *best* random variable $f(x, \omega)$. Therefore, it is required to discriminate the goodness of each family member of the uncertainty set U. Ranking of random variables in function $f(x, \omega)$ is a common practice in stochastic programming framework. Ranking of random variables is done according to their expectations by picking the biggest in a maximization problem.
- *(iii)Solution algorithm*: in stochastic programing framework, the problem equation (3.16) of optimization problem is required to be recast for getting optimal solution of linear

programming problems. The most used method for solving LP problem is the simplex method to get the optimal value of decision vector x. Let the random parameter vector λ is continuously affecting the optimization problem equation (3.16), then in stochastic framework, a discrete approximation is required to make such a problem solvable. For this, random vector λ is modeled as a set of Ω , which are the plausible outcomes of different scenario set ω , where each scenario $\omega \in \Omega$ has a probability π_{ω} of occurrence as such that $\sum_{\alpha \in \Omega} \pi_{\omega} = 1$.

In decision making problems with uncertain data parameter, stochastic programming provides the required tools and concepts with optimize solutions. Uncertainty parameters using distribution functions have been traditionally characterized in optimization problem. These distribution functions are describing most plausible realization of the uncertain parameter by generating a number of scenarios. Further these scenarios using probabilistic distribution functions are used in stochastic programming based optimization to make optimal decision regarding decision variable vector x. The approach exhibits two drawbacks:

- (i) There are difficulties to characterize some uncertain parameters using distribution functions; for example, offering behavior of rival producers.
- (ii) Generally, to describe the most plausible outcomes of the uncertain parameters, there are a large number of scenarios are needed. This may lead to large-scale optimization problems which become intractable or very difficult to solve.

3.4 Uncertainty Characterization in Robust Optimization Framework

Robust optimization is nothing but it addresses those uncertain parameters in optimization problems with uncertain parameters which are not described by using probability distributions but uncertainty sets [84]. An uncertainty set has set structure which is used to characterize the uncertain parameter's possible outcomes. In a robust optimization the aim is to determine a solution to an optimization problem in such a way that the obtained solution is feasible for any realization of the uncertain parameters, solution is optimal.

To formulate Robust Optimization, uncertainty set is to be properly defined, so that uncertain phenomenon presented by uncertain parameters can be described properly. Uncertainty set can have either highly risky or conservative solution. Further complexity in uncertainty sets may lead to damage the tractability which characterizes the robust optimization with elliptical or polyhedral sets.

Fig. 3.3 shows the Computational framework for robust MG operation in deregulated environment.



Fig. 3.3 Computational framework for robust MG operation

Robust optimization is used to provide an alternative and compact manner to describe the uncertain parameters within the uncertainty set at the cost of reduced flexibility. Optimal results obtained from such approaches are generally more computationally efficient than their stochastic programming counterparts, while preserving sufficient modeling flexibility.

The following nominal linear optimization problem has been considered as similar to problem considered in stochastic optimization equation (3.16) for better understanding:

$$\begin{array}{ccc} Max. & C & x \\ S.t. & Ax \le b \\ & l \le x \le u \end{array} \tag{3.16}$$

In any decision making problem there are three types of robustness (i) constraint robustness (ii) objective robustness (iii) combinational robustness. If uncertainty is present in only constraint parameter, then the problem is defined as constraint robustness problem. If uncertainty is present in only objective function parameter, then the problem is classified as objective robustness problem. If uncertainty is present in both constraints parameter and objective function parameter then it is defined as combinational robustness. Due to effective modeling of uncertain parameters the problem formulated by the robust optimization is tractable and has low computational burden. In the above formulation, it is assumed that data uncertainty only affects the elements in matrix A. it is assumed without loss of generality that the objective function C is not subject to uncertainty. Therefore the problem is solved as constraint robustness problem.

A robust formulation of optimization problem equation (3.16) is able to withstand the parameter uncertainty under data uncertainty set U. further RO formulation is linear without affecting the value objective function and ready to extend discrete optimization problems. The equivalent RO based formulation of equation (3.16) is given as follows:

Max. C'x

Subject to

$$\begin{split} \sum_{j} a_{ij} x_{j} + z_{i} \Gamma i + \sum_{j \in J_{i}} p_{ij} \leq b_{i} & \forall i \\ z_{i} + p_{ij} \geq a_{ij}^{\Lambda} y_{j} & \forall i \neq 0, j \in J_{i} \\ p_{ij} \geq 0 & \forall i, j \in J_{i} \\ y_{j} \geq 0 & \forall j & (3.17) \\ z_{i} \geq 0 & \forall i \\ -y_{j} \leq x_{j} \leq y_{j} & \forall j \\ l_{j} \leq x_{j} \leq u_{j} & \forall j \\ x_{i} \in Z & \forall i = 1, 2, 3 \dots k \end{split}$$

In above reformulation of problem Γi has been introduced for robustness purposes, the values of Γi lies in the interval of $[0, |J_i|]$, where $J_i = \left\{j; a_{ij}^{\Lambda} > 0\right\}$. The value of Γi is assumed to be an integer. The role of the parameter Γi in the constraints is to adjust the degree of robustness of the proposed method against the level of conservatism of the solution. Consider the *i*th constraint of the nominal problem $a_i x \le b_i$. Let J_i be the set of coefficients a_{ij} , $j \in J_i$ that are subject to parameter uncertainty, i.e., a_{ij} , $j \in J_i$ independently takes values according to a symmetric distribution with a mean equal to the

nominal value a_{ij} in the interval $\left[a_{ij} - a_{ij}^{\Lambda}, a_{ij} + a_{ij}^{\Lambda}\right]$. It is unlikely that all of the $a_{ij}, j \in J_i$ will change. The parameter Γi controls the level of robustness in the objective. As an optimal solution is needed that optimizes against all scenarios under which a number Γi of the constraint coefficients can vary in such a way as to maximally influence the objective. Let $J_i = \left\{j; a_{ij}^{\Lambda} > 0\right\}$. If $\Gamma i = 0$, the influence of the uncertain parameter deviations can be completely ignored, while if $\Gamma i = |J_i|$, all possible uncertain parameter deviations are considered, which is indeed most conservative. In general a higher values of Γi increases the level of robustness at the expense of higher nominal cost.

3.5 Summary

In this chapter basics of uncertainty characterization have been discussed. Different types of uncertainties presented in MG operation have been illustrated. Further basics of ARIMA model for modeling of an uncertain parameter have been discussed in non-stationary time series signals. Further on the basis of ARIMA model, ACF and PACF formulation has been discussed. Uncertainty characterization in both stochastic framework and robust optimization framework has been discussed and problem formulation considering uncertain parameter in both stochastic framework and robust optimization framework has been derived.

Chapter 4

Robust Microgrid Operation in Deregulated Environment Considering Wind Uncertainty

Τ

HIS chapter presents proposed robust optimization based approach for optimal generation scheduling of MG in both grid-connected and gridisolated modes. Wind power uncertainty has been modeled through interval forecasting using ARIMA model. A comparative study on daily MG cost of operation using proposed robust optimization based approach with deterministic and stochastic approach has been performed.
CHAPTER - 4 ROBUST MICROGRID OPERATION IN DEREGULATED ENVIRONMENT CONSIDERING WIND UNCERTAINTY

4.1 Introduction

Increasing penetration of renewable energy resources in evolving deregulated power system has great public attention, recently. Among the renewable energy sources, the penetration of wind and solar generators has increased due to their much evolved technology and wide spread availability. To integrate these renewable sources as Distributed Energy Resources (DERs) into the utility grid, MGs play an important role [1]. MG is a small-scale power supply network that is designed to provide power for a small community and can increase the reliability and economics of energy supply [2]. A typical architecture consists of dispatchable, non dispatchable generating unit and consumers. MG can be operated in two modes: grid connected and grid isolated. In grid-connected mode, MG can export its excess power to the grid and can import its deficit power from grid. MG can manage its demand independently in grid-isolated mode [3].

Due to uncontrollable nature of demand and to operate the generators within its technical limitations, the efficient and economic management of the generators is highly recommended. With increasing penetration of renewable sources like wind and solar in the distribution power system, MG management system may become complex. The grid price volatility in deregulated environment of power system has further increased the complexity of MG's energy management system. Therefore, development of efficient approaches for MG management under different uncertainties is necessary.

This chapter presents robust optimization based approach for optimal MG operation in deregulated environment considering renewable power uncertainties. Proposed approach is formulated for generation scheduling problem of MG in both grid-connected and grid-isolated mode. Wind uncertainty can be modelled through interval forecasting using time series based Autoregressive Integrated Moving Average (ARIMA) model. Other uncertainties (like PV generation uncertainty, demand uncertainty and grid prices uncertainty) are modelled using deterministic

point forecasting method. The proposed approach is illustrated through a case study having both dispatchable and non-dispatchable generators through different modes of operation. Further the impact of degree of robustness is analyzed in both cases on the total cost of operation of the MG. A comparative analysis between obtained results using proposed approach and other existing approach shows the strength of proposed approach in cost minimization in MG management.

4.2 Problem Description

MG operator can solve the MG energy scheduling problem with an objective to minimize the total cost of operation considering various constraints. Parameters in both grid-connected and isolated mode are slightly different. The detailed basic formulation of MG energy management formulation is described below:

4.2.1 Grid Connected Mode

In this mode, the MG is connected to the utility grid therefore, MG can Import/Export its deficit/excess power from/to the utility grid. In the deregulated environment, MG can earn revenue by selling its excess power to the spot market. Additionally, MG can also earn revenue by selling its reserve capacity to the spot market if prices are higher than its marginal cost. On the other hand the cost of operation can be minimized by meeting its deficit power from the utility grid when grid power is cheaper than MG's reserve cost. In grid-connected mode, the basic problem is formulated as follows:

(i) **Objective Function**

$$\min \sum_{t=1}^{N_{t}} \left[\sum_{j=1}^{N_{j}} C_{j} \left(P_{j,t} + r_{j,t}^{u} - r_{j,t}^{d} \right) + P_{grid,t} * \lambda_{grid,t} \right]$$
(4.1)

MO solves the generation scheduling problem with an objective to minimize the total cost of operation. The total cost of operation of MG is "The total sum of operating cost of dispatchable generating unit's power, reserves and cost of power imported/exported from/to grid", is termed as total operation cost of MG.

(ii) Power Balance Constraint

The Power balance constraint states that "the sum of scheduled power of

dispatchable generating units, wind power units and PV generation must be equal to day-ahead forecasted demand at any time t". Wind power, PV generation and demand can be forecasted using any existing approach.

$$\sum_{j=1}^{N_j} P_{j,t} + \sum_{w=1}^{N_w} P_{w,t}^f + \sum_{v=1}^{N_v} PV_{v,t}^f = Pd_t^f, \quad \forall t$$
(4.2)

(iii) Real time Power Balance Constraint

$$\sum_{j=1}^{N_j} \left(r_{j,t}^u - r_{j,t}^d \right) + \sum_{w=1}^{N_w} \left(\Delta P_{w,t} \right) + \sum_{v=1}^{N_v} \Delta P V_{v,t} + P_{grid,t} - \Delta P d_t = 0, \quad \forall t$$
(4.3)

Due to error in existing forecasting approaches, the actual values of wind power, PV generation and demand may be different from their forecasted values. The system imbalance created due to the error in forecasting in real-time can be compensated by the reserve capacity of dispatchable units or by the power from the main grid depending on the cost of compensation.

(iv) Imbalance Constraints

$$\Delta P_{w,t} = P_{w,t}^a - P_{w,t}^f, \qquad \forall w, \forall t$$
(4.4)

$$\Delta PV_{v,t} = PV_{v,t}^a - PV_{v,t}^f, \qquad \forall v, \forall t$$
(4.5)

$$\Delta P_{d,t} = P_{d,t}^a - P_{d,t}^f, \qquad \forall d, \forall t$$
(4.6)

In real-time, the deviation of wind power, PV generation and system demand from their forecasted values can be calculated by the corresponding imbalance constraints expressed in equation (4.4)-(4.6). In this work, forecasting error of wind power generation, PV Generation & demand is assumed as a random variable with zero mean and constant standard deviation. The Real value of wind power, PV power and demand is obtained by multiplying the error coefficient to their forecasted values.

(v) Dispatchable Units Power Constraint

$$P_j^{\min} \le P_{j,t} \le P_j^{\max}, \qquad \forall j, \forall t$$
(4.7)

$$r_{j,t}^{u} \le R_{j,t}^{u,\max}, \qquad \forall j, \forall t$$
(4.8)

$$r_{j,t}^{d} \le R_{j,t}^{d,\max}, \qquad \forall j, \forall t$$
(4.9)

$$P_{j,t} + r_{j,t}^u - r_{j,t}^d \ge 0, \qquad \forall j, \forall t$$

$$(4.10)$$

$$P_{j,t} + r_{j,t}^u - r_{j,t}^d \le P_{j,t}^{\max}, \qquad \forall j, \forall t$$

$$(4.11)$$

Equation (4.7) maintains scheduled dispatchable generation unit power within predefined limits, while Constraints (4.8) and (4.9) define the scheduled upward and downward reserve of dispatchable generating units. It must be lesser than their maximum upward and downward reserve capacity respectively. Constraints (4.10) and (4.11) state that sum of scheduled power of dispatchable units with their respective reserves must be within their specified limits.

(vi) Grid Power constraint

$$P_{grid,t} \le P_{grid,t}^{\max}, \qquad \forall t \tag{4.12}$$

Due to transmission network constraints, MG cannot meet the whole demand through grid power when grid prices are lower than operation cost. On the other hand, the complete reserve capacity cannot be sold in the spot market during high market prices. Constraint (4.12) ensures that the power exchange between the main grid & MG is less than predefined capacity of the connecting.

4.2.2 Grid Isolated Mode

In this mode also, similar to grid connected mode, it is also required to minimize the expenses of the MG. However, more attention is given to meet the demand within technical limits. The objective function in the isolated mode is stated as:

(i) Objective Function

$$\min \sum_{t=1}^{N_t} \left[\sum_{j=1}^{N_j} C_j \left(P_{j,t} + r_{j,t}^u - r_{j,t}^d \right) + P_{d,t}^{shed} * \lambda_d^{shed} \right]$$
(4.13)

Similar to the grid connected mode, the objective of MO in this mode is also to minimize the total cost of operation. However, the cost of exchange of power between main grid & MG is replaced by the cost of demand shedding. Since, the cost of demand shedding is very high, it is generally avoided till the complete utilization of the reserve capacity.

(ii) Real time Power Balance Constraint

To maintain system security and reliability, the generation is always equal to

system demand at each instant.

$$\sum_{j=1}^{N_j} \left(r_{j,t}^u - r_{j,t}^d \right) + \sum_{w=1}^{N_w} \left(\Delta P_{w,t} \right) + \sum_{v=1}^{N_v} \Delta P V_{v,t} + P_{d,t}^{Shed} - \Delta P d_t = 0, \quad \forall t$$
(4.14)

The system imbalance created due to volatile wind power, PV generation and demand at real-time is balanced by scheduling the reserve capacities of dispatchable generating units and shedding of demand.

(iii) Demand Shedding Constraint

$$P_{d,t}^{shed} \le P_{d,t} \qquad \qquad \forall t \tag{4.15}$$

The shedding power is always less than equal to the scheduled demand in dayahead scheduling.

(iv) Power Balance Constraint

$$\sum_{j=1}^{j} P_{j,t} + \sum_{w=1}^{N_w} P_{w,t}^f + \sum_{v=1}^{N_v} PV_{v,t}^f = Pd_t^f, \quad \forall t$$
(4.16)

Similar to Grid connected mode, the power balance constraint in this mode, states that "the sum of scheduled power of dispatchable generating units, wind power units and PV generation must be equal to day-ahead forecasted demand at any time t".

(v) Imbalance Constraints

$$\Delta P_{w,t} = P_{w,t}^a - P_{w,t}^f, \qquad \forall w, \forall t$$
(4.17)

$$\Delta PV_{v,t} = PV_{v,t}^a - PV_{v,t}^f, \qquad \forall v, \forall t$$
(4.18)

$$\Delta P_{d,t} = P_{d,t}^a - P_{d,t}^f, \qquad \forall d, \forall t$$
(4.19)

Similar to above discussed grid connected mode, the imbalance constraints defined as real-time deviation of wind power, PV generation and system demand from their forecasted values that can be calculated by the corresponding imbalance constraints expressed in equation (4.17)-(4.19).

(vi) Dispatchable Units Power Constraint

$$P_j^{\min} \le P_{j,t} \le P_j^{\max}, \qquad \forall j, \forall t$$
(4.20)

$$r_{j,t}^{u} \le R_{j,t}^{u,\max}, \qquad \forall j, \forall t$$
(4.21)

$$r_{j,t}^{d} \le R_{j,t}^{d,\max}, \qquad \forall j, \forall t \qquad (4.22)$$

$$P_{j,t} + r_{j,t}^u - r_{j,t}^d \ge 0, \qquad \forall j, \forall t$$

$$(4.23)$$

$$P_{j,t} + r_{j,t}^{u} - r_{j,t}^{d} \le P_{j,t}^{\max}, \qquad \forall j, \forall t$$

$$(4.24)$$

The dispatchable unit power constraints in grid isolated mode are similar to grid connected mode that defined as follows: Equation (4.20) maintains scheduled dispatchable generation unit power within predefined limits, while Constraints (4.21) and (4.22) define the scheduled upward and downward reserve of dispatchable generating units. It must be lesser than their maximum upward and downward reserve capacity respectively. Constraints (4.23) and (4.24) state that sum of scheduled power of dispatchable units with their respective reserves must be within their specified limits.

4.3 Uncertainty Modeling

The main source of uncertainty in the operation of MG in both the modes, in deregulated power systems, is random wind power. Random nature of wind power can be modelled by stochastic process. ARIMA model is the most popular approach for modelling any stochastic process and is widely used for forecasting electricity prices, demand and the prices of commodities other than electricity. As wind power varies with atmospheric conditions, the time series of wind power is non-stationary. Since, ARIMA model is applicable for only stationary data series, so wind power is converted in to stationary series by differentiation. In this work, ARIMA model is used for forecasting upper and lower limit of wind power. The typical ARIMA (p, d, q) model is expressed as

$$Y_{t} = \mu + \sum_{i=1}^{p} \gamma_{i} Y_{t-1} + \sum_{i=1}^{q} \theta_{i} \varepsilon_{t-1} + \varepsilon_{t}$$
(4.25)

Where, Y_t is the forecasted limits of wind power at time t, p is the order of autoregressive coefficients, γ_i and q are the orders of moving average coefficients θ_i . The term ε_t is a normally distributed random number with mean zero and constant variance, also known as white noise or error signal. The term μ is a

constant in the model. The main objective of this work is to develop a robust optimization approach for optimal generation scheduling of MG. For the sake of simplicity, modelling of seasonal impacts in the wind power uncertainty is not considered in the present work, as this work focuses on short term generation scheduling of MG.

4.4 Robust Optimization Approach

A robust optimization approach has been extensively used in engineering and economics field for solving different decision making problems with different uncertainties. For modeling of these problems Robust Optimization can provide a solution that is guaranteed to be "good" for all or most possible realization of the uncertain parameter. In any decision making problem there are three types of robustness (i) constraint robustness (ii) objective robustness (iii) combinational robustness. If uncertainty is present in only constraint parameter, then the problem is defined as constraint robustness problem. If uncertainty is present in only objective function parameter, then the problem is classified as objective robustness problem. If uncertainty is present in both constraint parameter and objective function parameter then it is defined as combinational robustness. Due to effective modeling of uncertain parameters the problem formulated by the robust optimization is tractable and has low computational burden [152].

When degree of robustness is zero, the problem reduces to a deterministic case which solves the problem using the nominal values *i.e.*, expected values of the uncertain coefficients. Degree of robustness can be adjusted according to the trade-off between decision maker's risk preference and the conservatism of the solution [152].

4.5 Proposed Problem Formulation

This section provides the proposed formulation of MG scheduling problem in both Grid connected and Grid isolated mode using robust optimization approach. In this problem, wind power is considered as uncertain parameter that is present in the constraints, therefore the formulated problem has constraint robustness.

4.5.1 Grid Connected Mode

In this mode, problem described in section 2.2 is reformulated by using Robust

Optimization as follows:

(i) Objective Function

Since uncertainty is present in the constraint only, the Objective function, similar to equation (4.1) in which minimization of total cost of operation is represented as the sum of operating cost of dispatchable generating unit's power, reserves and cost of power exchange between main grid & the MG.

$$\min \sum_{t=1}^{N_t} \left[\sum_{j=1}^{N_j} C_j \left(P_{j,t} + r_{j,t}^u - r_{j,t}^d \right) + P_{grid,t} * \lambda_{grid,t} \right]$$
(4.26)

(ii) Power Balance Constraint

$$\sum_{j=1}^{j} P_{j,t} + \sum_{w=1}^{N_w} \left(\sum_{k=1}^{N_k} W_{t,k} + z_k \Gamma_k + \sum_{k=1}^{N_k} q_k \right) + \sum_{v=1}^{N_v} P V_{v,t}^f = P d_t^f, \quad \forall t$$
(4.27)

Power balance constraint is represented in equation (4.27). This constraint states that the sum of scheduled power of dispatchable generating units, wind power units and PV generation must be equal to day-ahead forecasted demand at any time t. But irrespective of the previous formulation, wind power is considered as uncertain parameter and modeled in RO framework, where variables like z_k , q_k and Γ_k are introduced to solve the problem, Where, Γ_k is the degree of robustness which controls the robustness of the problem and z_k , q_k are the dual variables to linearize the non-linearity in the problem. The PV generation and demand can be forecasted using any existing approach.

(iii) Robust Constraints

$$z_k + q_k \ge W_{t,k} y_k, \qquad \forall k \neq 0, \, k \in N_k \tag{4.28}$$

$$q_k \ge 0, \qquad \qquad \forall k, k \in N_k \tag{4.29}$$

$$y_k \ge 0, \qquad \forall k \tag{4.30}$$

$$z_k \ge 0, \qquad \qquad \forall k \tag{4.31}$$

Robust constraints are represented by equations (4.28)-(4.31), Where, equation (4.28) states that dual variable z_k and q_k are taken into account as the known bounds of wind power in such a way that the sum of both variables is always greater than the

wind characterization parameter $\hat{W}_{t,k} y_k \cdot y_k$ is an auxiliary variable, used to obtain the corresponding linear expressions. However, $\hat{W}_{t,k}$ is obtained as the proportionate of the difference of lower and upper bounds of wind power. Constraint (4.29), (4.30) and (4.31) show that the dual variables used in robust approach must be positive.

(iv) Real time Power Balance Constraint

$$\sum_{j=1}^{N_{j}} \left(r_{j,t}^{u} - r_{j,t}^{d} \right) + \sum_{w=1}^{N_{w}} \left(\Delta P_{w,t} \right) + \sum_{v=1}^{N_{v}} \Delta P V_{v,t} + P_{grid,t} - \Delta P d_{t} = 0, \quad \forall t$$
(4.32)

Due to error in existing forecasting approaches, the actual values of wind power, PV generation and demand may be different from their forecasted values. The system imbalance created due to the error in forecasting in real-time can be compensated by the reserve capacity of dispatchable units or by the power from the main grid depending on the cost of compensation.

(v) Imbalance Constraints

$$\Delta P_{w,t} = P_{w,t}^a - P_{w,t}^f, \qquad \forall w, \forall t$$
(4.33)

$$\Delta PV_{v,t} = PV_{v,t}^a - PV_{v,t}^f, \qquad \forall v, \forall t$$
(4.34)

$$\Delta P_{d,t} = P_{d,t}^a - P_{d,t}^f, \qquad \forall d, \forall t$$
(4.35)

In real-time, the deviation of wind power, PV generation and system demand from their forecasted values can be calculated by the corresponding imbalance constraints expressed in (4.33)-(4.35). In this work, forecasting error of wind power generation, PV Generation & demand is assumed as a random variable with zero mean and constant standard deviation. The Real value of wind power, PV power and demand is obtained by multiplying the error coefficient to their forecasted values.

(vi) Dispatchable Units Power Constraint

$$P_j^{\min} \le P_{j,t} \le P_j^{\max}, \qquad \forall j, \forall t$$
(4.36)

$$r_{j,t}^{u} \le R_{j,t}^{u,\max}, \qquad \forall j, \forall t$$

$$(4.37)$$

$$r_{j,t}^d \le R_{j,t}^{d,\max}, \qquad \forall j, \forall t$$

$$(4.38)$$

$$P_{j,t} + r_{j,t}^u - r_{j,t}^d \ge 0, \qquad \forall j, \forall t$$
(4.39)

$$P_{j,t} + r_{j,t}^u - r_{j,t}^d \le P_{j,t}^{\max}, \qquad \forall j, \forall t$$

$$(4.40)$$

Equation (4.36) maintains scheduled dispatchable generation unit power within predefined limits, while Constraints (4.37) and (4.38) define the scheduled upward and downward reserve of dispatchable generating units. It must be lesser than their maximum upward and downward reserve capacity respectively. Constraints (4.39) and (4.40) state that sum of scheduled power of dispatchable units with their respective reserves must be within their specified limits.

(vii) Grid Power Cconstraint

$$P_{grid,t} \le P_{grid,t}^{\max}, \qquad \forall t \tag{4.41}$$

Due to transmission network constraints, MG cannot meet the whole demand through grid power when grid prices are lower than operation cost. On the other hand, the complete reserve capacity cannot be sold in the spot market during high market prices. Constraint (4.41) ensures that the power exchange between the main grid & MG is less than predefined capacity of the connecting.

4.5.2 Grid Isolated Mode

Since wind power uncertainty is involved in the constraints, the MG generation scheduling problem in this mode is formulated using Robust Optimization approach similar to Grid connected mode differing in the objective function and supply-demand equilibrium constraint. Similar to previous one, in this mode also, it is required to minimize the expenses of the MG. However, more attention is given to meet the demand with stable operation. The problem in the isolated mode is formulated as:

(i) Objective Function

Similar to previously defined grid connected mode under RO Approach, the objective of MO in this mode is also the minimization of total operation cost. However, the Objective function is same as (4.13) in grid isolated mode.

$$\min \sum_{t=1}^{N_t} \left[\sum_{j=1}^{N_j} C_j \left(P_{j,t} + r_{j,t}^u - r_{j,t}^d \right) + P_{d,t}^{shed} * \lambda_d^{shed} \right]$$
(4.42)

It states that the sum of operating cost of dispatchable generating units and cost of demand shedding should be minimized.

(ii) Power Balance Constraint

Power balance constraint is same as grid connected mode formulation in RO framework by equation (4.43). For the same, robust constraint is also similar to the previous case *i.e* equation (4.28)-(4.31).

$$\sum_{j=1}^{j} P_{j,t} + \sum_{w=1}^{N_w} \left(\sum_{k=1}^{N_k} W_{t,k} + z_k \Gamma_k + \sum_{k=1}^{N_k} q_k \right) + \sum_{v=1}^{N_v} P V_{v,t}^f = P d_t^f, \quad \forall t$$
(4.43)

Power balance constraint is represented in equation (4.43). This constraint states that the sum of scheduled power of dispatchable generating units, wind power units and PV generation must be equal to day-ahead forecasted demand at any time t. But irrespective of the previous formulation, wind power is considered as uncertain parameter and modeled in RO framework, where variables like z_k , q_k and Γ_k are introduced to solve the problem, Where, Γ_k is the degree of robustness which controls the robustness of the problem and z_k , q_k are the dual variables to linearize the non-linearity in the problem. The PV generation and demand can be forecasted using any existing approach.

(iii) Robust Constraints

$$z_k + q_k \ge W_{t,k} y_k, \qquad \forall k \neq 0, k \in N_k$$

$$(4.44)$$

$$q_k \ge 0, \qquad \qquad \forall k, k \in N_k \tag{4.45}$$

$$y_k \ge 0, \qquad \forall k \tag{4.46}$$

$$z_k \ge 0, \qquad \qquad \forall k \tag{4.47}$$

Robust constraints are represented by equations (4.44) - (4.47), Where, equation (4.44) states that dual variable z_k and q_k are taken into account as the known bounds of wind power in such a way that the sum of both variables is always greater than the wind characterization parameter $\hat{W}_{t,k} y_k$. y_k is an auxiliary variable, used to obtain the corresponding linear expressions. However, $\hat{W}_{t,k}$ is obtained as the proportionate of the difference of lower and upper bounds of wind power. Constraint (4.45), (4.46) and (4.47) show that the dual variables used in robust approach must be positive.

(iv) Real time Power Balance Constraint

$$\sum_{j=1}^{N_j} \left(r_{j,t}^u - r_{j,t}^d \right) + \sum_{w=1}^{N_w} \left(\Delta P_{w,t} \right) + \sum_{v=1}^{N_v} \Delta P V_{v,t} + P_{d,t}^{Shed} - \Delta P d_t = 0, \quad \forall t$$
(4.48)

To maintain system security and reliability, the generation is always equal to system demand at each instant. The system imbalance created due to volatile wind power, PV generation and demand at real-time is balanced by scheduling the reserve capacities of dispatchable generating units and shedding of demand.

(v) Demand Shedding Cconstraint

$$P_{d,t}^{shed} \le P_{d,t} \qquad \qquad \forall t \tag{4.49}$$

The shedding power is always less than equal to the scheduled demand in dayahead scheduling.

(vi) Imbalance Constraints

$$\Delta P_{w,t} = P_{w,t}^a - P_{w,t}^f, \qquad \forall w, \forall t$$
(4.50)

$$\Delta PV_{v,t} = PV_{v,t}^a - PV_{v,t}^f, \qquad \forall v, \forall t$$
(4.51)

$$\Delta P_{d,t} = P_{d,t}^a - P_{d,t}^f, \qquad \forall d, \forall t$$
(4.52)

In real-time, the deviation of wind power, PV generation and system demand from their forecasted values can be calculated by the corresponding imbalance constraints expressed in (4.50)-(4.52). In this work, forecasting error of wind power generation, PV Generation & demand is assumed as a random variable with zero mean and constant standard deviation. The Real value of wind power, PV power and demand is obtained by multiplying the error coefficient to their forecasted values.

(vii) Dispatchable Units Power Constraint

$$P_j^{\min} \le P_{j,t} \le P_j^{\max}, \qquad \forall j, \forall t$$
(4.53)

$$r_{j,t}^{u} \le R_{j,t}^{u,\max}, \qquad \forall j, \forall t \qquad (4.54)$$

$$r_{j,t}^{d} \le R_{j,t}^{d,\max}, \qquad \forall j, \forall t$$
(4.55)

$$P_{j,t} + r_{j,t}^u - r_{j,t}^d \ge 0, \qquad \forall j, \forall t$$
(4.56)

$$P_{j,t} + r_{j,t}^u - r_{j,t}^d \le P_{j,t}^{\max}, \qquad \forall j, \forall t$$

$$(4.57)$$

Equation (4.53) maintains scheduled dispatchable generation unit power within predefined limits, while Constraints (4.54) and (4.55) define the scheduled upward and downward reserve of dispatchable generating units. It must be lesser than their maximum upward and downward reserve capacity respectively. Constraints (4.56) and (4.57) state that sum of scheduled power of dispatchable units with their respective reserves must be within their specified limits.

4.6 Proposed Algorithm

The following algorithm (as shown in Fig. 4.1) is adopted to solve the formulated Robust Optimization based MG generation scheduling problem for both grid connected and isolated mode:

- Step 1: Collect historical data: Collect the historical data of wind speed for a specific location where wind units are located. The wind speed is converted in power using power curve of standard turbine models.
- Step 2: ARIMA model parameters estimation: From collected data, parameters of ARIMA model are estimated. Order of AR and MA terms are determined by the observation of Auto Correlation Function (ACF) and Partial Auto Correlation Function (PACF) plot, respectively. The coefficients of AR and MA terms and the variance of error signal are determined by Least Square method [159].
- Step 3: Interval Forecasting: To forecast upper and lower limits of wind power, ARIMA model expressed in (16) is simulated for each time instant.
- Step 4: Robust optimization Parameter Initialization: The initial output of wind unit is set to be $W_t = W_t^{\text{max}}$, degree of robustness is $\Gamma_k = N_t$ with incremental factor G^k , which takes increasing values in the interval [0, 1] in steps of 0.01, denoted as δ [158].
- Step 5: Iteration Counter Initialization: The total number of iterations N_k is defined and the counter is initialized with k = 1.
- Step 6: Wind Uncertainty Characterization: The upper and lower limits of wind power are obtained from Step 3. The wind power is set in the iteration as $\hat{W}_{t,k} = G^k \left(W_t^{\max} - W_t^{\min} \right)$. Value of wind power is limited in the interval $[W_t^{\max}, W_t^{\max} - \hat{W}_{t,k}]$.
- Step 7: Problem Solution: Formulated MG scheduling problem for both modes are

solved and optimal results are obtained in the form of optimal scheduling of dispatchable units, optimal scheduling of upward and downward reserve, optimal Main grid to MG power exchange, optimal demand shedding.

Step 8: Check Iteration Counter: If iteration $k \le N_k$, update the range of incremental

factor G^k by step δ and repeat Steps 6 and 7. Otherwise, go to next step.

Step 9: Published results: Show obtained optimal cost of MG along with optimal value of scheduled dispatch-able unit generation and reserves.

Step 10: End.



Fig. 4.1 Flow chart of proposed algorithm

4.7 Case Study

This section provides the study of test system, simulation results and discussions.

4.7.1 Data

In this study, installed capacity of PV unit and wind units are considered as 200 kW and 1.34 MW respectively. Wind units consists ENERCON turbine model whose parameters are detailed in manufacturer database [160]. The historical wind speed data of the duration of 27.04.2006 to 08.06.2007, used in this study from publically available database at Illinois Institute of Rural Affair, USA. [161]. Along with renewable units, 10 dispatchable units are considered, whose parameters are given in Table 4.1. Hourly grid price, demand profile and PV generation profile is shown in Table 4.2.

Dispatchable unit	Installed capacity (kW)	Upward reserve	Downward reserve (kW)	Marginal cost (\$/kWh)
1	600	240	240	0.0141
2	600	240	240	0.0222
3	400	160	160	0.02775
4	400	160	160	0.03375
5	300	120	120	0.0321
6	300	120	120	0.0384
7	200	80	80	0.04335
8	200	80	80	0.049125
9	100	40	40	0.04554
10	100	40	40	0.05154

Table 4.1: Dispatchable Unit Parameters

The capacity of line linking the main grid and MG is considered as 5000 kW. Hourly grid price obtained for PJM electricity market is shown in [165]. Short-term generation scheduling on the basis of marginal generation cost is common in evolving electricity markets. The fixed cost has no role in short-term operation of electricity markets. This study focuses on short-term generation scheduling of MG in deregulated environment, considering the marginal cost of generators. Marginal cost of generators is calculated according to their cost functions.

4.7.2 Simulation Results

Using the above data, proposed approach is tested on a typical MG by considering two cases. In Case I, grid connected operation of MG is simulated while in Case II, grid isolated operation of MG is simulated. Both test cases are coded in GAMS platform and final LP problem is solved using CPLEX solver [163].

Time (h)	Price (\$/kw)	Demand (pu)	PV (pu)
1	0.04836	0.784	0
2	0.04461	0.754	0
3	0.043695	0.739	0
4	0.044535	0.736	0
5	0.051765	0.756	0
6	0.069	0.814	0
7	0.08238	0.912	0.002
8	0.080115	0.977	0.008
9	0.08802	0.991	0.035
10	0.090045	0.997	0.1
11	0.09258	1	0.23
12	0.088725	0.997	0.233
13	0.0906	0.992	0.318
14	0.09054	0.989	0.433
15	0.08595	0.981	0.37
16	0.07908	0.971	0.403
17	0.07461	0.964	0.33
18	0.06552	0.956	0.238
19	0.062415	0.949	0.133
20	0.065775	0.945	0.043
21	0.07266	0.962	0.003
22	0.0609	0.950	0
23	0.05253	0.884	0
24	0.044925	0.822	0

Table 4.2: Daily Market Prices, Load and PV Profiles

For uncertainty modeling using ARIMA model first the order of suitable ARIMA model on the basis of collected historical data is determined. Order of AR and MA terms are obtained by the observation of ACF and PACF plot. These plots are shown in Fig. 4.2 and 4.3, respectively. Fig. 4.4 shows the forecasted wind power for 24 hours with upper and lower bounds.



Fig. 4.2 ACF plot of sample wind power data



Fig. 4.3 PACF plot of sample wind power data

Chapter-4



Fig. 4.4 Hourly wind power

From the Fig. 4.3 and 4.4, it is observed that ACF plot is sinusoidal and PACF for first two legs are out of bounds. Thus, ARIMA (3,0,0) is suitable for modeling wind power uncertainty. The estimated values of AR coefficients are 0.779, -0.03, and 0.039. The value of constant term and variance of white noise are 244.098 and 0.264 respectively. On a 95% confidence interval forecasted limits are used in proposed robust optimization based MG scheduling approach.

Case I : Microgrid Operation in Grid Connected Mode

In this case, MG operation in grid-connected mode is considered for simulation using proposed robust optimization method.

Fig. 4.5 shows the obtained scheduled power of dispatchable units in MG during grid connected mode of operation. From the figure, it can be observed that it will be beneficial to schedule maximum capacity from the unit with lower cost than the unit with higher cost. Further Fig. 4.6 and Fig. 4.7 show the obtained scheduled upward and downward reserves of dispatchable units respectively. From the figures, it can be observed that any dispatchable unit is scheduled for one type of reserve requirement at any particular instant of time t. It can be observed that at the time of exporting the power the most costly dispatchable units are scheduled to their downward reserve while in case of importing the power these units are scheduled to their upward reserve. In these figures at time axis, 1 indicates the first hour *i.e.* 00:00 AM to 01:00

AM of the day.



Fig. 4.5 Scheduled power of dispatchable units in Case I



Fig. 4.6 Scheduled upward reserve in Case I



Fig. 4.7 Scheduled downward reserve in Case I

Fig. 4.8 shows power exchange between the main grid and the MG in Case I.



Fig. 4.8 Power imported/exported to/from grid in Case I

From the Fig. 4.8, it can be observed that from Hour 6 to 17 and 20 to 22, power is being exported to the grid and in remaining hours, power is being imported from

the grid. It can be concluded that, when grid price is higher, MG can export their reserve capacity to grid and vice-versa to reduce its operating cost. By comparing Fig. 4.4 and 4.8, it can be seen that, in the above mention time interval the available wind power is lower, leading to import of power from the grid.

Case II : Microgrid Operation in Grid Isolated Mode

Similarly, in this case, MG operation in grid-isolated mode is considered using proposed robust optimization method. The cost of demand shedding is considered as \$3.5/kWh according to Federal Energy Regulation Commission (FERC), Electric Tariff [164]. Similar to the previous case, Fig. 4.9, Fig. 4.10, and Fig. 4.11 show the scheduled dispatchable unit power, upward and downward reserve capacity respectively using proposed robust optimization approach.



Fig. 4.9 Dispatchable unit scheduled power in Case II

The capacity of dispatchable units is fully utilized therefore leading to demand shedding. From Figs 4.10 and 4.11, it can be visualized that MG can utilize its upward reserve capacity during peak demand to reduce demand shedding events.



Fig. 4.10 Scheduled upward reserve in Case II



Fig. 4.11 Scheduled downward reserve in Case II

Demand shedding in Case II is shown in Fig. 4.12. From the figure, it is observed that demand shedding event occurs in Hours 8, 14 and 15. For these peak hours, the availability of wind and solar power is minimum as shown in Fig. 4.2 and Table 4.2.



Fig. 4.12 Demand shedding in Case II

4.7.3 Discussion

A comparative analysis on MG operation is carried out. For this purpose, the same problem is simulated using deterministic and stochastic programming approach. In deterministic approach, wind power output is simply forecasted using ARIMA model and the forecasted value of wind power is considered to compute the scheduling of MG.

Approaches	Operation Cost (\$)			
	Case I	Case II		
Deterministic	857.9346	1500.230		
Stochastic	830.6901 (-3.07 %)	1334.909 (-11.02 %)		
Proposed	814.8662 (-5.02%)	1241.441 (-17.25 %)		

Table 4.3: Comaparitive Analysis of MG Daily Operation Cost

In stochastic programming approach, the wind power uncertainty is modeled by generating a large number of scenarios (*i.e.* 10000) and then reducing them into a smaller number of scenarios (*i.e.*25) as in [56]. The MG operation cost obtained using proposed robust optimization approach is compared with both deterministic and stochastic approaches as shown in Table 4.3. From the Table 4.3, it is observed that in both cases (*i.e.* Grid connected operation of MG and Grid isolated operation of MG)

the daily operating cost is significantly lower in the proposed approach than any other existing approach. It is also observed from Table 4.3 that, The reduction in daily cost using proposed approach, in Case II is higher than compared to Case I. This is because in Case II, small reduction in demand shedding event results in large reduction in total cost of operation. From Table 4.3 in the proposed approach the cost of MG operation is higher in grid-isolated mode than in grid-connected mode. This is due the fact that the reserves and dispatchable units are fully utilized to meet the demand.



Fig. 4.13 Impact of degree of robustness on operational cost of MG

The impact of degree of robustness on operation cost, in both cases can be evaluated by varying the degree of robustness from zero to 36 during simulation. The obtained per unit cost of operation are shown in Fig. 4.13. Per unit cost of operation, at any degree of robustness is obtained by dividing its actual value by the reference value, which represents the cost of operation at zero degree of robustness. From Fig. 4.14, it can be seen that the optimal value of degree of robustness is 24, where per unit cost of operation is minimum. At this degree of robustness, optimal condition is obtained because in this problem wind is considered as an uncertain parameter and its uncertainty is considered for 24 hours. This degree of robustness takes into account all possible deviations of the uncertain parameter for whole day. Additionally, at zero degree of robustness the cost of operation obtained using proposed robust optimization based approach is equal to the cost of operation obtained using deterministic approach.

4.8 Summary

A robust optimization based approach has been proposed in this chapter for optimal generation scheduling of MG in both grid-connected and grid-isolated modes. Wind power uncertainty has been modeled through interval forecasting using ARIMA model. A comparative study on daily MG cost of operation using proposed robust optimization based approach with deterministic and stochastic approach has been performed. A significant reduction in cost of operation clearly shows strength of proposed robust optimization based approach in MG generation scheduling in the deregulated environment. The impact of degree of robustness on the cost of operation in both cases has also evaluated and compared with existing methods. The proposed approach will be more advantageous in incorporating different type of uncertainties such as Solar Power Uncertainty, Demand uncertainty and grid price uncertainty. This work may also be enhanced by incorporating multi-micro grids, different type of consumers and considering different technical constraints. Chapter-4

Chapter 5

Microgrid Operation Considering Solar Power Uncertainty

Τ

HIS chapter details out the stochastic and robust optimization based approaches for optimal generation scheduling of MG in both gridconnected and grid isolated modes. For stochastic and robust optimization base approach, solar power uncertainty has been modeled through scenarios and interval forecasting using ARIMA model, respectively. A comparative study on daily MG cost of operation using proposed robust optimization based approach with deterministic and stochastic approach has been performed.

MICROGRID OPERATION CONSIDERING SOLAR POWER UNCERTAINTY

5.1 Introduction

Due to high reduction of market price of Photovoltaic (PV) panels the economics of PV based solar power generation is becoming attractive for power generating companies in present deregulated power system. Thus, the penetration of solar power is continuously increasing in the both transmission and distribution power system in the form of medium/large solar parks and small PV installations at the buildings. To integrate the solar power sources in the power system the MG is more robust and cost effective as compared to conventional centralized grid. However, several technical and regulatory challenges are created in MG operation due to random and stochastic nature of solar power. Optimal generation scheduling is a major problem for MG operator to effectively manage the balance in between the power supply and demand. The MG generation scheduling considering solar power uncertainty is the main objective of this chapter.

Solar power depends on weather conditions such as rain, dust storms and cloud covers. Due to inherent randomness of these weather conditions solar power is highly variable or random in nature. The size of solar power uncertainty can vary accoriding to size and location of solar PV array. Hence, modeling of solar power uncertainty is a challenging task for MG Operators. Modeling of solar power uncertainty affects the balancing needs and required reserve power, storage devices, with higher costs. Thus, the effective modeling of solar power uncertainty is required for cost effective energy management of MGs. Stochastic and robust optimizations are two effective approaches to model such problems considering different uncertainties. Robust optimization based approach has been proposed for optimal MG operation in deregulated environment by considering wind power uncertainty in Chapter 3.

This chapter presents stochastic and robust optimization based approach for optimal MG generation scheduling by considering solar power uncertainty. Solar power uncertainty is modelled through scenarios and interval forecasting for both stochastic and robust optimization based approaches using ARIMA model. Other uncertainties (like wind power uncertainty, demand uncertainty and grid price uncertainty) are modelled using deterministic point forecasting method. The Proposed approach is illustrated by a case study on MG having wind, PV, dispatchable generating units and loads. Obtained results show the strength of proposed approach in cost minimization over other existing approaches.

5.2 Problem Description

MG operator can solve the MG energy scheduling problem with an objective to minimize the total cost of operation considering various constraints. Parameters in both grid-connected and isolated mode are slightly different. In grid connected mode, the MG is connected to the utility grid, hence, MG can Import/Export its deficit/excess power from/to the utility grid. In the deregulated environment, MG can earn revenue by selling its excess power to the spot market. Additionally, MG can also earn revenue by selling its reserve capacity to the spot market if prices are higher than its marginal cost. On the other hand the cost of operation can be minimized by meeting its deficit power from the utility grid when grid power is cheaper than MG's reserve cost. In grid isolated mode, similar to grid connected mode, it is also required to minimize the expenses of the MG. However, more attention is given to meet the demand within technical limits of MG's parameters. The detailed basic formulation of MG energy management formulation has been described in Chapter 4, Section 4.2.

5.3 Uncertainty Modeling

Solar power is a main source of uncertainty in the operation of MG in deregulated power systems. Random nature of solar power uncertainty can be modelled by stochastic process. ARIMA model is the most popular approach for modelling any stochastic process and is widely used for forecasting of electricity price, demand and the price of commodities other than electricity. As solar power varies with atmospheric conditions, the time series of solar power is non-stationary. Since, ARIMA model is applicable for only stationary data series, solar power is converted in to stationary series by differentiation [159]. In this work, ARIMA model is considered for forecasting upper and lower limit of solar power. The typical ARIMA (p, d, q) model is expressed as

$$Y_t = \mu + \sum_{i=1}^p \gamma_i Y_{t-1} + \sum_{i=1}^q \theta_i \varepsilon_{t-1} + \varepsilon_t$$
(5.1)

Where, Y_t is the forecasted limits of solar power at time t, p is the order of autoregressive coefficients, γ_i and q are the orders of moving average coefficients θ_i . The term ε_i is a normally distributed random number with mean zero and constant variance, also known as white noise or error signal. The term μ is a constant in the model. For stochastic optimization based framework, ARIMA model is used for solar power scenario generation while for robust optimization framework, it is used for solar power interval forecasting. The main objective of this chapter is to develop a stochastic and robust optimization approach for optimal generation scheduling of MG. For the sake of simplicity, modelling of seasonal impacts in the solar power uncertainty is not considered in the present work, as this thesis focuses on short term generation scheduling of MG.

5.4 Stochastic and Robust Optimization Approach

Stochastic and robust optimization approaches have been extensively used in engineering and economics field for solving different decision making problems with involvement of different uncertainties. In stochastic optimization based approach, uncertainties are modeled through a number of scenarios with known probability distributions and the expected value of the objective is optimized. In robust optimization approach uncertainties are modeled as uncertain parameters belonging to a known uncertainty set and problem is optimized for the worst case over that set.

Both Stochastic and Robust optimization are the alternative approaches to deal with uncertain data in a single period and in a multi-period decision making problems. However, the main difficulty associated with the stochastic optimization is to provide the information of probability distribution functions of the underlying stochastic parameters. This requirement creates a heavy burden on MG operator because in many real world situations, such information is unavailable or very difficult to obtain. On the other hand Robust Optimization addresses the uncertain nature of the function without making specific assumptions on probability distributions. Robust Optimization adopts a min-max approach that addresses uncertainty by guaranteeing the feasibility and optimality of the solution against all instances of the parameters within the uncertainty set. Both Stochastic and Robust Optimization approaches are widely used for modeling of decision making problems in power systems such as power trading, unit commitment and economic load dispatch etc. The details of

Stochastic and Robust optimization approaches have been provided in Chapter 3.

5.5 Proposed Problem Formulation in Stochastic Optimization Framework

This Section describes the proposed formulation of MG generation scheduling problem considering solar power uncertainty in stochastic optimization framework with an objective of total generation cost minimization for both grid connected and isolated mode. Solar power uncertainty is modeled through a number of scenarios. Scenarios are possible outcomes with occurrence probabilities. Solar power scenarios are generated using ARIMA model. A large number of scenarios are required for accurate modeling of any uncertainty. But, solving the decision making problems using such large number of scenarios is a challenging task computationally and sometime problems are also intractable. Therefore, generated scenarios are reduced in such a manner that their stochastic properties cannot be lost. Backward reduction algorithm is used for scenarios reduction in this work.

5.5.1 Grid Connected Mode

In this mode, MG is connected to the utility grid, therefore, MG can Import/Export its deficit/excess power from/to the utility grid. In the deregulated environment, MG can earn revenue by selling its excess power and reserve capacity to the spot market. On the other hand the cost of operation can be minimized by meeting its deficit power from the utility grid when grid power is cheaper than MG's reserve cost. In grid-connected mode, the problem is formulated in stochastic optimization framework as:

$$\min \sum_{t=1}^{N_t} \left[\sum_{\omega=1}^{N_{\omega}} \pi_{\omega} \left\{ \sum_{j=1}^{N_j} C_j \left(P_{j,\omega,t} + r_{j,\omega,t}^u - r_{j,t}^d \right) + P_{grid,\omega,t} * \lambda_{grid,t} \right\} \right]$$
(5.2)

$$\sum_{j=1}^{N_j} P_{j,\omega,t} + \sum_{w=1}^{N_w} P_{w,t}^f + \sum_{v=1}^{N_v} PV_{v,\omega,t} = Pd_t^f, \quad \forall \omega, \forall t$$
(5.3)

$$\sum_{j=1}^{N_j} \left(r_{j,\omega,t}^u - r_{j,\omega,t}^d \right) + \sum_{w=1}^{N_w} \left(\Delta P_{w,t} \right) + \sum_{v=1}^{N_v} \Delta P V_{v,\omega,t} + P_{grid,t} - \Delta P d_t = 0, \quad \forall \, \omega, \forall t$$
(5.4)

$$\Delta P_{w,t} = P_{w,t}^a - P_{w,t}^f, \qquad \forall w, \forall t$$
(5.5)

$$\Delta PV_{\nu,\omega,t} = PV_{\nu,t}^a - PV_{\nu,\omega,t}, \qquad \forall \nu, \forall \,\omega, \forall t$$
(5.6)

$$\Delta P_{d,t} = P_{d,t}^a - P_{d,t}^f, \qquad \forall d, \forall t$$
(5.7)

$$P_j^{\min} \le P_{j,\omega,t} \le P_j^{\max}, \qquad \forall j, \forall \omega, \forall t$$
(5.8)

$$r_{j,\omega,t}^{u} \le R_{j,t}^{u,\max}, \qquad \forall j, \forall \omega, \forall t$$
(5.9)

$$r_{j,\omega,t}^d \le R_{j,t}^{d,\max}, \qquad \forall j, \forall \omega, \forall t$$
(5.10)

$$P_{j,\omega,t} + r^{u}_{j,\omega,t} - r^{d}_{j,\omega,t} \ge 0, \qquad \forall j, \forall \omega, \forall t$$
(5.11)

$$P_{j,\omega,t} + r_{j,\omega,t}^{u} - r_{j,\omega,t}^{d} \le P_{j,t}^{\max}, \qquad \forall j, \forall \omega, \forall t$$
(5.12)

$$P_{grid,\omega,t} \le P_{grid,t}^{\max}, \qquad \forall \,\omega, \forall t$$
(5.13)

In the above formulation, the objective (5.2) of scheduling problem is to minimize the total cost of operation, which includes the operating cost of dispatchable generating unit's power, reserves and cost of power imported/exported from/to grid. The objective function is subjected to different constraints expressed in (5.3) to (5.13). Constraint (5.3) defines that the total sum of scheduled power of dispatchable generating units, wind power units and PV generation must be equal to day-ahead forecasted demand for each scenario of solar power at any time *t*. Further, wind power, grid market price and demand can be forecasted using any existing approach.

Due to error in existing forecasting approaches, the actual values of wind power and demand may be different from their forecasted values. The system imbalance created due to the error in forecasting in real-time can be compensated with scheduling of the reserve capacity of dispatchable units or by the power from the main grid depending on the cost of compensation. In real-time, the deviation of wind power generation, PV solar power generation and MG demand from their forecasted values can be calculated by their corresponding imbalance constraints expressed in (5.5.)-(5.7). In this work, forecasting error of wind power generation, PV solar power Generation & demand is assumed as a random variable with zero mean and constant standard deviation. The Real value of wind power, PV solar power power and demand is obtained by multiplying the error coefficient to their forecasted values. Forecasted PV solar power generation is equal to reduced solar power scenarios.

Constraint (5.8) maintains scheduled power of the dispatchable generation units

within predefined limits, while Constraints (5.9) and (5.10) define the scheduled upward and downward reserve of dispatchable generating units. Constraints (5.11) and (5.12) state that sum of scheduled power of dispatchable units with their respective reserves must be within their specified limits.

Due to transmission network constraints, MG cannot meet the whole demand through grid power when grid prices are lower than MG operation cost. On the other hand, the complete reserve capacity cannot be sold in the spot market during the high market price. Constraint (5.13) ensures that the power exchange between the main grid & MG is less than predefined capacity of grid connecting line. It should be noted that the network configuration of the MG is neglected in the proposed work due to following reasons: (i) Size of MG is limited to few MVA, due to this limited capacity and the proximity of load and generation network constraint are not typical constraint in MG operation, and (ii) Network model greatly complicates MG generation scheduling problem without taking any benefit.

5.5.2 Grid Isolated Mode

Similar to grid connected mode in this mode also objective is to minimize the expenses of the MG operation. However, more attention is given to meet the demand within technical limits of MG parameters. The objective function in the isolated mode can be given as follows:

$$\min \sum_{t=1}^{N_{t}} \left[\sum_{\omega=1}^{N_{\omega}} \pi_{\omega} \left\{ \sum_{j=1}^{N_{j}} C_{j} \left(P_{j,\omega,t} + r_{j,\omega,t}^{u} - r_{j,\omega,t}^{d} \right) + P_{d,\omega,t}^{shed} * \lambda_{d}^{shed} \right\} \right]$$
(5.14)

$$\sum_{j=1}^{N_j} \left(r_{j,\omega,t}^u - r_{j,\omega,t}^d \right) + \sum_{w=1}^{N_w} \left(\Delta P_{w,t} \right) + \sum_{v=1}^{N_v} \Delta P V_{v,\omega,t} + P_{d,\omega,t}^{Shed} - \Delta P d_t = 0, \quad \forall \, \omega, \forall t$$
(5.15)

$$P_{d,\omega,t}^{shed} \le P_{d,t} \qquad \qquad \forall \, \omega, \forall t \tag{5.16}$$

$$\sum_{j=1}^{j} P_{j,\omega,t} + \sum_{w=1}^{N_w} P_{w,t}^f + \sum_{v=1}^{N_v} PV_{v,\omega,t} = Pd_t^f, \quad \forall \omega, \forall t$$
(5.17)

$$\Delta P_{w,t} = P_{w,t}^a - P_{w,t}^f, \qquad \forall w, \forall t$$
(5.18)

$$\Delta PV_{\nu,\omega,t} = PV_{\nu,t}^{a} - PV_{\nu,\omega,t}, \qquad \forall \nu, \forall \omega, \forall t$$
(5.19)

$$\Delta P_{d,t} = P_{d,t}^a - P_{d,t}^f, \qquad \forall d, \forall t$$
(5.20)

$$P_{j}^{\min} \le P_{j,\omega,t} \le P_{j}^{\max}, \qquad \forall j, \forall \omega, \forall t$$
(5.21)

$$r_{j,\omega,t}^{u} \le R_{j,t}^{u,\max}, \qquad \forall j, \forall \omega, \forall t$$
(5.22)

$$r_{j,\omega,t}^{d} \le R_{j,t}^{d,\max}, \qquad \forall j, \forall \omega, \forall t$$
(5.23)

$$P_{j,\omega,t} + r^{u}_{j,\omega,t} - r^{d}_{j,\omega,t} \ge 0, \qquad \forall j, \forall \omega, \forall t$$
(5.24)

$$P_{j,\omega,t} + r_{j,\omega,t}^{u} - r_{j,\omega,t}^{d} \le P_{j,t}^{\max}, \qquad \forall j, \forall \omega, \forall t$$
(5.25)

In the above formulation, the objective (5.14) is to minimize net expected cost of MG including operating cost of scheduled generation and reserve capacity and cost of demand shedding for each scenario of solar power at time t. Constraint (5.15) ensures that the power generation is always equal to system demand for each scenario at real time. The system imbalance created due to volatile wind power generation, PV generation and demand at real-time is balanced by scheduling the reserve capacities of dispatchable generating units and shedding of demand. The demand shedding should always be less than equal to the scheduled demand in day-ahead scheduling, which is presented by (5.16). Constraint (5.17) states that the total sum of scheduled power of dispatchable generating units, wind power units and PV generation units must be equal to day-ahead forecasted demand for each solar power scenario at any time t. Imbalance constraints (5.18)-(5.20) express real-time deviation of wind power, PV generation and system demand from their forecasted values. Conventional generation unit power within predefined limits is maintained by (5.21). Constraints (5.22) and (5.23) define the scheduled upward and downward reserve of dispatchable generating units. It must be lesser than their maximum upward and downward reserve capacity respectively. Constraints (5.24) and (5.25) state that total sum of scheduled power of dispatchable units with their respective reserves must be within their specified limits.

5.6 Proposed Problem Formulation in Robust Optimization Framework

This section provides the proposed formulation of MG scheduling problem in both Grid connected and Grid isolated mode in robust optimization framework. Solar power is considered as an uncertain parameter and it is present in the constraint part of the problem, therefore, the MG generation scheduling problem is formulated as constraint robustness.

5.6.1 Grid Connected Mode

In this mode, problem described in section 4.2 is reformulated by using Robust Optimization as follows:

(i) **Objective Function**

$$\min \sum_{t=1}^{N_t} \left[\sum_{j=1}^{N_j} C_j \left(P_{j,t} + r_{j,t}^u - r_{j,t}^d \right) + P_{grid,t} * \lambda_{grid,t} \right]$$
(5.26)

The Objective function is minimization of total cost of operation is as represented as the sum of operating cost of dispatchable generating unit power, reserves and cost of power exchange between main grid & the MG.

(ii) Power Balance Equality Constraint

$$\sum_{j=1}^{j} P_{j,t} + \sum_{\nu=1}^{N_{\nu}} \left(\sum_{k=1}^{N_{k}} PV_{t,k} + z_{k} \Gamma_{k} + \sum_{k=1}^{N_{k}} q_{k} \right) + \sum_{w=1}^{N_{w}} W_{w,t}^{f} = Pd_{t}^{f}, \quad \forall t$$
(5.27)

Power balance constraint is shown in equation (5.27). This constraint states that the total sum of scheduled power of dispatchable generating units, wind power units and PV generation units must be equal to day-ahead forecasted demand at any time t. But irrespective of the previous formulation, here solar power is considered as an uncertain parameter and modeled in RO framework, where variables like z_k , q_k and Γ_k are introduced to solve the problem, Where, Γ_k is the degree of robustness which controls the robustness of the problem and z_k , q_k are the dual variables used to linearize the non-linearity in the problem. The wind power and demand can be forecasted using any existing approach.

$$\sum_{j=1}^{N_{j}} \left(r_{j,t}^{u} - r_{j,t}^{d} \right) + \sum_{w=1}^{N_{w}} \left(\Delta P_{w,t} \right) + \sum_{v=1}^{N_{v}} \Delta P V_{v,t} + P_{grid,t} - \Delta P d_{t} = 0, \quad \forall t$$
(5.28)

Due to error in existing forecasting approaches, the actual values of wind power, PV generation and demand may be different from their forecasted values. The system imbalance created due to the error in forecasting in real-time can be compensated by the reserve capacity of dispatchable units or by the power from the main grid depending on the cost of compensation.
(iii) Robust Constraints

$$z_k + q_k \ge PV_{t,k} \ y_k, \qquad \forall k \neq 0, \, k \in N_k$$
(5.29)

$$q_k \ge 0, \qquad \qquad \forall k, k \in N_k \tag{5.30}$$

$$y_k \ge 0, \qquad \forall k \tag{5.31}$$

$$z_k \ge 0, \qquad \forall k \tag{5.32}$$

Robust constraints are represented by equations (5.29)-(5.30), Where, equation (5.29) states that dual variable z_k and q_k are taken into account as the known bounds of solar power in such a way that the sum of both variables is always greater than the solar power characterization parameter $P\hat{V}_{t,k}y_k$. y_k is an auxiliary variable, used to obtain the corresponding linear expressions. However, $P\hat{V}_{t,k}$ is obtained as the proportionate of the difference of lower and upper bounds of solar power. Further constraint (5.30), (5.31) and (5.32) show that the dual variables used in robust approach must be positive.

(iv) Imbalance Constraints

$$\Delta P_{w,t} = P_{w,t}^a - P_{w,t}^f, \qquad \forall w, \forall t$$
(5.33)

$$\Delta PV_{v,t} = PV_{v,t}^a - PV_{v,t}^f, \qquad \forall v, \forall t$$
(5.34)

$$\Delta P_{d,t} = P_{d,t}^a - P_{d,t}^f, \qquad \forall d, \forall t$$
(5.35)

In real-time, the deviation of wind power, PV generation and system demand from their forecasted values can be calculated by the corresponding imbalance constraints expressed in (5.33)-(5.35). In this work, forecasting error of wind power generation, PV Generation & demand is assumed as a random variable with zero mean and constant standard deviation. The Real value of wind power, PV power and demand is obtained by multiplying the error coefficient to their forecasted values.

(v) Dispatchable Units Power Constraint

$$P_j^{\min} \le P_{j,t} \le P_j^{\max}, \qquad \forall j, \forall t$$
(5.36)

$$r_{j,t}^{u} \le R_{j,t}^{u,\max}, \qquad \forall j, \forall t$$
(5.37)

$$r_{j,t}^d \le R_{j,t}^{d,\max}, \qquad \forall j, \forall t$$
 (5.38)

$$P_{j,t} + r_{j,t}^u - r_{j,t}^d \ge 0, \qquad \forall j, \forall t$$
 (5.39)

$$P_{j,t} + r_{j,t}^u - r_{j,t}^d \le P_{j,t}^{\max}, \qquad \forall j, \forall t$$
(5.40)

Equation (5.36) maintains scheduled dispatchable generation unit power within predefined limits, while Constraints (5.37) and (5.38) define the scheduled upward and downward reserve of dispatchable generating units. It must be lower than their maximum upward and downward reserve capacity respectively. Constraints (5.39) and (5.40) state that sum of scheduled power of dispatchable units with their respective reserves must be within their specified limits.

(vi) Grid Power Cconstraint

$$P_{grid,t} \le P_{grid,t}^{\max}, \qquad \forall t \tag{5.41}$$

Due to transmission network constraints, MG cannot meet the whole demand through grid power when grid prices are lower than operation cost. On the other hand, the complete reserve capacity cannot be sold in the spot market during high market prices. Constraint (5.41) ensures that the power exchange between the main grid & MG is less than predefined capacity of the grid connecting line.

5.6.2 Grid Isolated Mode

Since solar power uncertainty is involved in the constraints, the MG generation scheduling problem is formulated using Robust Optimization approach similar to Grid connected mode. Here, the objective function and supply-demand equilibrium constraint are slightly differed. Similar to previous one, in this mode also, it is required to minimize the expenses of the MG. However, more attention is given to meet the demand with stable operation. The problem in the isolated mode is formulated as:

(i) **Objective Function**

$$\min \sum_{t=1}^{N_t} \left[\sum_{j=1}^{N_j} C_j \left(P_{j,t} + r_{j,t}^u - r_{j,t}^d \right) + P_{d,t}^{shed} * \lambda_d^{shed} \right]$$
(5.42)

Similar to previously defined grid connected mode under RO Approach, the objective of MG generation scheduling problem in this mode is also the minimization of total operation cost. It states that the sum of operating cost of dispatchable

generating units and cost of demand shedding should be minimized.

(ii) Power Balance Constraint

$$\sum_{j=1}^{j} P_{j,t} + \sum_{w=1}^{N_w} \left(\sum_{k=1}^{N_k} W_{t,k} + z_k \Gamma_k + \sum_{k=1}^{N_k} q_k \right) + \sum_{v=1}^{N_v} P V_{v,t}^f = P d_t^f, \quad \forall t$$
(5.43)

Power balance constraint is represented in equation (5.43). This constraint states that the sum of scheduled power of dispatchable generating units, wind power units and PV generation units must be equal to day-ahead forecasted demand at any time *t*. But irrespective of the previous formulation, here solar power is considered as an uncertain parameter and modeled in RO framework, where variables like z_k , q_k and Γ_k are introduced to solve the problem, Where, Γ_k is the degree of robustness which controls the robustness of the problem and z_k , q_k are the dual variables used to linearize the non-linearity in the problem. The wind power generation and demand can be forecasted using any existing approach.

$$\sum_{j=1}^{N_j} \left(r_{j,t}^u - r_{j,t}^d \right) + \sum_{w=1}^{N_w} \left(\Delta P_{w,t} \right) + \sum_{v=1}^{N_v} \Delta P V_{v,t} + P_{d,t}^{Shed} - \Delta P d_t = 0, \quad \forall t$$
(5.44)

To maintain system security and reliability, power generation is always equal to system demand at each instant. The system imbalance created due to volatile wind power, PV generation and demand at real-time is balanced by scheduling the reserve capacities of dispatchable generating units and shedding of demand.

(iii) Robust Constraints

 $\overline{}$

$$z_k + q_k \ge PV_{t,k} \ y_k, \qquad \forall k \neq 0, \ k \in N_k$$
(5.45)

$$q_k \ge 0, \qquad \qquad \forall k, k \in N_k \tag{5.46}$$

$$y_k \ge 0, \qquad \forall k \tag{5.47}$$

$$z_k \ge 0, \qquad \qquad \forall k \tag{5.48}$$

Robust constraints are represented by equations (5.45)-(5.48), Where, equation (5.45) states that dual variable z_k and q_k are taken into account as the known bounds of solar power in such a way that the sum of both variables is always greater than the

solar characterization parameter $P\hat{V}_{t,k}y_k$. y_k is an auxiliary variable, used to obtain the corresponding linear expressions. However, $P\hat{V}_{t,k}$ is obtained as the proportionate of the difference of lower and upper bounds of solar power. Constraint (5.46), (5.47) and (5.48) show that the dual variables used in robust approach must be positive.

(iv) Demand Shedding Cconstraint

$$P_{d,t}^{shed} \le P_{d,t} \qquad \qquad \forall t \tag{5.49}$$

The shedding power is always less than equal to the scheduled demand in dayahead scheduling.

(v) Imbalance Constraints

$$\Delta P_{w,t} = P_{w,t}^a - P_{w,t}^f, \qquad \forall w, \forall t$$
(5.50)

$$\Delta PV_{v,t} = PV_{v,t}^a - PV_{v,t}^f, \qquad \forall v, \forall t$$
(5.51)

$$\Delta P_{d,t} = P_{d,t}^a - P_{d,t}^f, \qquad \forall d, \forall t$$
(5.52)

In real-time, the deviation of wind power, PV generation and system demand from their forecasted values can be calculated by the corresponding imbalance constraints expressed in (5.50)-(5.52). In this work, forecasting error of wind power generation, PV Generation & demand is assumed as a random variable with zero mean and constant standard deviation. The Real value of wind power, PV power and demand is obtained by multiplying the error coefficient to their forecasted values.

(vi) Dispatchable Units Power Constraint

$$P_j^{\min} \le P_{j,t} \le P_j^{\max}, \qquad \forall j, \forall t$$
(5.53)

$$r_{j,t}^{u} \le R_{j,t}^{u,\max}, \qquad \forall j, \forall t$$
(5.54)

$$r_{j,t}^{d} \le R_{j,t}^{d,\max}, \qquad \forall j, \forall t$$
(5.55)

- $P_{j,t} + r_{j,t}^{u} r_{j,t}^{d} \ge 0, \qquad \forall j, \forall t$ (5.56)
- $P_{j,t} + r_{j,t}^{u} r_{j,t}^{d} \le P_{j,t}^{\max}, \qquad \forall j, \forall t$ (5.57)

Equation (5.53) maintains the scheduled dispatchable generation unit power within predefined limits of grid isolated mode operation, while Constraints (5.54) and (5.55) define the scheduled upward and downward reserve of dispatchable generating units, which should be lesser than their maximum upward and downward reserve capacity respectively. Constraints (5.56) and (5.57) state that sum of scheduled power of dispatchable units with their respective reserves must be within their specified limits.

5.7 Proposed Methodology

The following methodology (as shown in Fig. 5.1) is adopted to solve the formulated Stochastic and Robust Optimization based MG generation scheduling problem for both grid connected and isolated mode:

- Step 1: Collect historical data: Collect the historical data of solar irradiation for a specific location where solar/PV units are located. The solar irradiation is converted in power, using power conversion formula.
- Step 2: ARIMA model parameters estimation: From collected data, parameters of ARIMA model are estimated. Order of AR and MA terms are determined by the observation of Auto Correlation Function (ACF) and Partial Auto Correlation Function (PACF) plot, respectively. The coefficients of AR and MA terms and the variance of error signal are determined by Least Square method [159].
- Step 3: Initialize Time Counter: Here 24 hour time period is considered to solve MG generation scheduling under solar power uncertainty. Start with time t = 1.
- Step 4: Choose Stochastic or Robust Approach: Select an approach to solve MG generation scheduling problem. If Stochastic is selected, go to Step 5 and if Robust is selected, go to Step 8.
- Step 5: Solar Power Scenario Generation: To generate solar power scenarios evaluate ARMA model Eq. (5.1).
- Step 6: Solar Power Scenario Reduction: Collect generated solar power scenarios from previous step. The probability distance-based backward algorithm is used for solar power scenario reduction.
- Step 7: Solve MG generation Scheduling Problem: Formulated MG scheduling

problem in stochastic optimization framework in (5.2)-(5.25) is solved. Optimal results are obtained in the form of optimal scheduling of dispatchable units, optimal scheduling of upward and downward reserve, optimal Main grid to MG power exchange, optimal demand shedding.

- Step 8: Interval Forecasting: To forecast upper and lower limits of wind power, ARIMA model expressed in (5.1) is simulated for each time instant.
- Step 9: Robust optimization Parameter Initialization: The initial output of solar/PV unit is set to be $PV_t = PV_t^{\max}$, degree of robustness is $\Gamma_k = N_t$ with incremental factor G^k , which takes increasing values in the interval [0, 1] in steps of 0.01, denoted as δ [158].
- Step 10: Iteration Counter Initialization: The total number of iterations N_k is defined and the counter is initialized with k = 1.
- Step 11: Solar Uncertainty Characterization: The upper and lower limits of solar power are obtained from Step 3. The solar power is set in the iteration as $P\hat{V}_{t,k} = G^k \left(PV_t^{\max} PV_t^{\min} \right).$ Value of solar power is limited in the interval $[PV_t^{\max}, PV_t^{\max} P\hat{V}_{t,k}].$
- Step 12: Problem Solution: Formulated robust MG scheduling problem for both modes in (5-26)-(5.53) are solved and optimal results are obtained in the form of optimal scheduling of dispatchable units, optimal scheduling of upward and downward reserve, optimal power exchange between main grid to MG and optimal demand shedding etc.
- Step 13: Check Iteration Counter: If iteration $k \le N_k$, update the range of incremental factor G^k by step δ and repeat Steps 11 and 12. Otherwise, go to next step.
- Step 14: Check Time Counter: If the desired time period counter, i.e., 24, has achieved, go to the next step; otherwise, update t=t+1 and go to Step 4.
- Step 15: Publish results: Show obtained optimal cost of MG along with optimal value of scheduled dispatchable unit generation and reserves.

Step 16: End.



Fig. 5.1 Flow chart of proposed algorithm

5.8 Results and Discussions

In this study, installed capacity of PV unit and wind units are considered as 200 kW and 1.34 MW respectively. Wind units consists ENERCON turbine model whose parameters are given in detail in manufacturer database [160]. The historical wind speed data of the duration of 27.04.2006 to 08.06.2007, are used in this study from publically available database at Illinois Institute of Rural Affair, USA. [161]. For same site the historical solar irradiation data is obtained from NREL for same duration

Chapter - 5

[162]. Along with renewable units, 10 dispatchable units are considered, whose parameters are given in Table 5.1. Hourly grid price, demand profile and hourly forecasted wind power are shown in Fig. 5.2 and Fig. 5.3 respectively.

Dispatchable unit	Installed capacity (kW)	Upward reserve (kW)	Downward reserve (kW)	Marginal cost (\$/kWh)
1	600	240	240	0.0141
2	600	240	240	0.0222
3	400	160	160	0.02775
4	400	160	160	0.03375
5	300	120	120	0.0321
6	300	120	120	0.0384
7	200	80	80	0.04335
8	200	80	80	0.049125
9	100	40	40	0.04554
10	100	40	40	0.05154

Table 5.1: Dispatchable Unit Parameters



Fig. 5.2 Daily market price (bar plot) and demand profile (line plot)



Fig. 5.3 Forecasted wind power output profile

The capacity of line linking the main grid and MG is considered as 5000 kW. Hourly grid price obtained for PJM electricity market is shown in [165]. Short-term generation scheduling on the basis of marginal generation cost is common in evolving electricity markets. The fixed cost has no role in short-term operation of electricity markets. This work focuses on short-term generation scheduling of MG in deregulated environment, considering the marginal cost of generators. Marginal cost of generators is calculated according to their cost functions.

Using the above data, proposed stochastic and robust approach is tested on a typical MG by considering both grid connected and grid isolated mode. Both modes of MG operation are coded in GAMS platform and final LP problem is solved using CPLEX solver [163].

5.8.1 Solar Power Uncertainty Modeling

For uncertainty modeling using ARIMA model, first the order of suitable ARIMA model on the basis of collected historical data is determined. Order of AR and MA terms are obtained by the observation of ACF and PACF plot. These plots are shown in Fig. 5.4 and 5.5, respectively. From these figures, it is observed that ACF plot is sinusoidal and PACF for first two legs are out of bounds. Thus, ARIMA (3,0,0) is suitable for modeling solar power uncertainty. The estimated values of AR

coefficients are 1.18, -0.309, and -0.032. The value of constant term and variance of white noise are 97.86 and 0.43, respectively.



Fig. 5.4 ACF plot of sample solar power data



Fig. 5.5 PACF plot of sample solar power data

To solve MG generation scheduling problem using proposed stochastic optimization based methodology, 1000 solar power scenarios are generated. Generated scenarios are reduced to 25 scenarios using backward reduction algorithm. Generated 1000 solar power scenarios along with mean scenario (red) and forecasted solar power (black) is shown in Fig. 5.6. Fig. 5.7 shows the reduced 25 solar power scenarios along with mean scenario (red). When

spread of uncertain parameter is not very high, a mean solar power scenario can be used to solve is MG generation scheduling problem. Practically, spread is there in solar power uncertainty thus reduced solar power scenarios are used for solving MG generation scheduling problem in stochastic optimization framework. For deterministic optimization based approach, forecasted solar power is used.



Fig. 5.6 Generated 1000 solar power scenarios (green) along with mean scenario (red) and forecasted solar power (black)



Fig. 5.7 Reduced 25 solar power scenarios (green) along with mean scenario (red) and forecasted solar power (black)

Fig. 5.8 shows the forecasted solar power for 24 hours with upper and lower bounds. On a 95% confidence interval forecasted limits are used in proposed robust

Chapter - 5

optimization based MG scheduling approach.



Fig. 5.8 Hourly forecasted solar power with upper and lower limits

5.8.2 Stochastic Optimization Based MG Operation in Grid Connected Mode

In this section, obtained results by solving formulated stochastic grid-connected MG operation using proposed methodology are discussed. Fig. 5.9 shows the obtained scheduled power of dispatchable units in MG during grid connected mode of operation.



Fig. 5.9 Scheduled power of dispatchable units in grid connected mode

It can be observed from Fig. 5.9, that it will be beneficial to schedule maximum capacity from the unit with lower cost than the unit with higher cost. Further Fig. 5.10 and Fig. 5.11 show the obtained scheduled upward and downward reserves of dispatchable units respectively.



Fig. 5.10 Scheduled downward reserve in grid connected mode



Fig. 5.11 Scheduled upward reserve in grid connected mode

Chapter - 5

It can be observed from Fig. 5.10 and Fig. 5.11, that any dispatchable unit is scheduled for one type of reserve requirement at any particular instant of time t. It can be observed that at the time of exporting the power the most costly dispatchable units are scheduled to their downward reserve while in case of importing the power these units are scheduled to their upward reserve. Fig. 5.12 shows power exchange between the main grid and the MG.



Fig. 5.12 Power imported/exported to/from grid in grid connected mode

From the Fig. 5.12, it can be observed that from hour 7, 9, 15 and 23, power is being exported to the grid and in remaining hours like as hour 1, 3, 4, 6, 11, 16, 17, 19 and 24, power is being imported from the grid. It can be concluded that, when grid price is higher, MG can export their reserve capacity to grid and vice-versa to reduce its operating cost. By comparing Fig. 5.7 and 5.8, it can be seen that, in the above mentioned time periods, the high solar and wind power is available, leading to export power to the grid.

5.8.3 Stochastic Optimization Based MG Operation in Grid Isolated Mode

Similarly, in this case, formulated stochastic MG operation in grid-isolated mode is considered using proposed methodology. The cost of demand shedding is considered as \$3.5/kWh according to Federal Energy Regulation Commission (FERC), Electric Tariff [164]. Similar to the previous case, Fig. 5.13, 5.14, and 5.15 show the scheduled dispatchable unit power, upward and downward reserve capacity respectively using proposed stochastic optimization approach. From Figs 5.14 and 5.15, it can be visualized that MG can utilize its upward reserve capacity during peak demand to reduce demand shedding events.



Fig. 5.13 Dispatchable unit scheduled power in grid isolated mode



Fig. 5.14 Scheduled upward reserve in grid isolated mode.

Chapter - 5



Fig. 5.15 Scheduled downward reserve in grid isolated mode

Demand shedding in stochastic optimization based grid isolated MG operation is shown in Fig. 5.16. From the figure, it is observed that demand shedding event occurs in hours 7, 14, 18 and 20. For these peak hours, availability of wind and solar power is lower as shown in Fig. 5.3 and Fig. 5.6. The capacity of dispatchable units is fully utilized therefore leading to demand shedding.



Fig. 5.16 Demand shedding in grid isolated mode

5.8.4 Robust Optimization Based MG Operation in Grid Connected Mode

In this case, MG operation in grid-connected mode is considered for simulation

using proposed robust optimization method. Fig. 5.17 and 5.18 show the obtained scheduled power of dispatchable units, scheduled upward reserve and scheduled downward reserve respectively during grid connected mode of operation.



Fig. 5.17 Scheduled power of dispatchable units in grid connected mode



Fig. 5.18 Scheduled upward reserve in grid connected mode

From the Fig. 5.17, it can be observed that it will be beneficial to schedule maximum capacity from the unit with lower cost than the unit with higher cost. From

the Fig. 5.18 and Fig. 5.19, it can be observed that any dispatchable unit is scheduled for one type of reserve requirement at any particular instant of time t. It can be observed that at the time of exporting the power the most costly dispatchable units are scheduled to their downward reserve while in case of importing the power these units are scheduled to their upward reserve. Fig. 5.20 shows power exchange between the main grid and the MG in Case I.



Fig. 5.19 Scheduled downward reserve in grid connected mode



Fig. 5.20 Power imported/exported to/from grid in grid connected mode

From the Fig. 5.20, it can be observed that from hour 6 to 17 and 21, power is

exported to the grid and in remaining hours, power is being imported from the grid. It can be concluded that, when grid price is higher, MG can export their reserve capacity to grid and vice-versa to reduce its operating cost.

5.8.5 Robust Optimization Based MG operation in grid isolated mode

Similarly, in this case, MG operation in grid-isolated mode is considered using proposed robust optimization method. The cost of demand shedding is considered as \$3.5/kWh according to Federal Energy Regulation Commission (FERC), Electric Tariff [164]. Similar to the previous case, Fig. 5.21, Fig. 5.22, and Fig. 5.23 show the scheduled dispatchable unit power, upward and downward reserve capacity respectively using proposed robust optimization approach. The capacity of dispatchable units is fully utilized, therefore, leading to demand shedding. From Figs 5.22 and 5.23, it can be visualized that MG can utilize its upward reserve capacity during peak demand to reduce demand shedding events.

Demand shedding in grid isolated mode is shown in Fig. 5.24. From the figure, it is observed that demand shedding event occurs in hours 16, 19 and 24.



Fig. 5.21 Dispatchable unit scheduled power in grid isolated mode



Fig. 5.22 Scheduled upward reserve in grid isolated mode



Fig. 5.23 Scheduled downward reserve in grid isolated mode



Fig. 5.24 Demand shedding in grid isolated mode

5.8.6 Comparative Analysis

A comparative analysis on MG operation is carried out. For this purpose, the same problem is simulated using deterministic, stochastic and robust optimization programming approach. In deterministic approach, solar/PV power output is simply forecasted using ARIMA model and the forecasted value of solar/PV power is considered to compute the scheduling of MG.

Approaches	Operation Cost (\$)		
	Grid Connected Mode	Grid Isolated Mode	
Deterministic	857.9346	1500.230	
Stochastic	820.59 (-4.35 %)	1334.909 (-11.02 %)	
Proposed	794.3616 (-7.49%)	1275.1955 (-15.00 %)	

Table 5.2: Comaparitive Analysis of MG Daily Operation Cost

In stochastic programming approach, the solar power uncertainty is modeled by generating a large number of scenarios (*i.e.* 10000) and then reducing them into a smaller number of scenarios (*i.e.* 25). The MG operation cost obtained using proposed robust optimization approach is compared with both deterministic and stochastic approaches as shown in Table 5.2. From the Table 5.2, it is observed that in both modes (*i.e.* Grid connected operation of MG and Grid isolated operation of MG) the daily operating cost is significantly lower in the proposed approach than any other existing approach. It is also observed from Table 5.2, that the reduction in daily cost

using proposed approach, in grid isolated mode is higher than compared to grid connected mode. This is because in grid isolated mode, small reduction in demand shedding event results in large reduction in total cost of operation. From Table 5.2, in the proposed approach the cost of MG operation is higher in grid-isolated mode than in grid-connected mode. This is due the fact that the reserves and dispatchable units are fully utilized to meet the demand.

5.8.7 Impact of Degree of Robustness

The impact of degree of robustness on operation cost, in both cases can be evaluated by varying the degree of robustness from zero to 24 during simulation. The obtained per unit cost of operation are shown in Fig. 5.25. Per unit cost of operation, at any degree of robustness is obtained by dividing its actual value by the reference value, which represents the cost of operation at zero degree of robustness. From Fig. 5.25, it can be seen that the optimal value of degree of robustness is 12; where per unit cost of operation is minimum. At this degree of robustness, optimal condition is obtained because in this problem solar power is considered as an uncertain parameter and solar power is available for ten hours only in a day, but as seasonal impact on uncertainty is not considered, so its uncertainty can be considered for 12 hours. This degree of robustness takes into account all possible deviations of the uncertain parameter for whole day. Additionally, at zero degree of robustness the cost of operation obtained using proposed robust optimization based approach is equal to the cost of operation obtained using deterministic approach.



Fig. 5.25 Impact of degree of robustness on operational cost of MG

5.9 Summary

Stochastic and robust optimization based approaches have been proposed in this chapter for optimal generation scheduling of MG in both grid-connected and gridisolated modes. For stochastic and robust optimization base approach, solar power uncertainty has been modeled through scenarios and interval forecasting using ARIMA model, respectively. A comparative study on daily MG cost of operation using proposed robust optimization based approach with deterministic and stochastic approach has been performed. A significant reduction in cost of operation clearly shows strength of proposed robust optimization based approach over stochastic and deterministic approach in MG generation scheduling in the deregulated environment. The impact of degree of robustness on the cost of operation in both cases has also been evaluated and compared with existing methods. The proposed approach will be more advantageous in incorporating different type of uncertainties such as Demand uncertainty and grid price uncertainty. This work may also be enhanced by incorporating multi-micro grids, different type of consumers and considering different technical constraints. Chapter - 5

Chapter 6

Robust Microgrid Operation Considering Wind and Solar Power Uncertainty

Τ

HIS chapter enhances the proposed robust optimization based approach for optimal generation scheduling of MG in both grid-connected and grid-isolated modes by considering both wind and solar power uncertainty. Both wind power and solar power uncertainty have been modeled through interval forecasting using ARIMA model. A comparative study on daily MG cost of operation using proposed robust optimization based approach with deterministic and stochastic approach has been performed.

6.1 Introduction

Micro Grid (MG) is commonly referred as an efficient way for integration of DG in the distribution system. In general, DG is a small scale generating unit, installed near the consumers in the distribution system to mitigate transmission losses and network congestion. DG can have a conventional or renewable energy sources. However, DG with renewable energy sources is widely utilized due to their peculiar characteristics such as pollution free, sustainable and freely available energy resources. Among the available renewable energy sources, wind power and solar power would be the most promising renewable sources throughout the world. Although, wind power and solar power generation are stochastic in nature, high penetration of wind and solar DGs in the distribution system leads considerable technical and economical challenges for MG operators in satisfying load demand and maintaining the stable and reliable operation of MGs. Due to wind power and solar power uncertainty, the reserve power requirement in MG management is increased that leads towards increased total MG operating cost.

Uncertainty presents in wind and solar power generation due to forecasting deviation results a great impact on the generation scheduling both in grid connected and grid isolated MG. In Chapter 3, uncertainty modeling approaches are discussed in detail. The impact of wind power uncertainty on the operation of grid connected and grid isolated MG has been investigated using robust optimization approach in Chapter 4. In chapter 5, solar power uncertainty in generation scheduling problem of both grid connected and grid isolated MG has been considered using stochastic and robust optimization approach. However, the impact of one uncertainty to another uncertainty has not been considered, therefore, MG operation considering both wind power and solar power uncertainty is needed to be modeled because both wind power and solar power are common energy sources in the MGs.

This chapter presents robust optimization based approach for optimal MG

operation in deregulated environment considering both wind power and solar power uncertainties. Proposed approach is formulated for generation scheduling problem of MG in both grid-connected and grid-isolated mode. Wind power and solar power uncertainties can be modelled through interval forecasting using time series based ARIMA model. Other uncertainties present in demand profile and grid price has been modelled using deterministic point forecasting method. The Proposed approach is illustrated by a realistic case study on MG having wind power, solar/PV power, dispatchable generating units and loads. Obtained results show the strength of proposed approach in cost minimization over the other existing approaches.

This chapter is organized as follows: Section 6.2 describes the generation scheduling problem of grid connected and isolated MG. Modeling of both wind power and solar power uncertainty is described in Section 6.3. Section 6.4 highlights the Robust Optimization approach. Section 6.5, proposes the formulation of the robust MG energy-scheduling model considering both wind power and solar power generation uncertainties. Section 6.6 introduces the test platform and provides a detailed case study of the proposed algorithm and compares the simulation results obtained using proposed robust optimization based approach and other existing approaches under various operational conditions. Section 6.7 concludes the Chapter.

6.2 Problem Description

MG generation scheduling problem can be modeled with an objective of total cost of operation minimization. The basic architecture of MG generation scheduling problem considering multiple uncertainties (like wind power and solar power uncertainty) has been shown in Fig. 6.1. It is clear from Fig. 6.1 that constraint parameters are slightly different in both grid-connected and grid isolated mode. The dispatchable unit parameters, forecasted demand, uncertain wind and solar power are common input for both grid-connected and isolated MG. Forecasted market price is an input parameter for only grid connected MG. Day ahead and real time power balance, dispatchable unit's generation, upward and downward reserves, power imbalance are common constraints in both mode of operation of MG. In grid-connected mode MG can Import/Export its deficit/excess power from/to the utility grid. While grid isolated mode demand shedding events are occurred when all existing power of MG is utilized to meet the required demand. The detailed basic formulation of MG energy management formulation is described in Section 4.2.



Fig. 6.1 Generation scheduling problem of grid connected and isolated MG

6.3 Uncertainty Modeling

Wind and solar are main source of uncertainty in the operation of grid connected and isolated MG in deregulated power systems. Similar to earlier chapter, random nature of such sources is modelled by stochastic process using ARIMA model. As both wind and solar power are vary with atmospheric conditions, the time series of wind and solar power are non-stationary. Since, ARIMA model is applicable for only stationary data series, so wind and solar power time series models are converted in to stationary series by differentiation [159]. The typical ARIMA (p, d, q) model is expressed as equation (6.1) Chapter-6

$$Y_t = \mu + \sum_{i=1}^p \gamma_i Y_{t-1} + \sum_{i=1}^q \theta_i \varepsilon_{t-1} + \varepsilon_t$$
(6.1)

Where, Y_t is the forecasted wind and solar power along with upper and lower limits at time t, p is the order of autoregressive coefficients, γ_i and q are the orders of moving average coefficients θ_i . The term ε_t is a normally distributed random number with mean zero and constant variance, also known as white noise or error signal. The term μ is a constant in the model.

6.4 Robust Optimization Approach

Robust optimization is nothing but it addresses those uncertain parameters in optimization problems with uncertain parameters which are not described by using probability distributions but uncertainty sets [84]. An uncertainty set has set structure which is used to characterize the uncertain parameter's possible outcomes. In a robust optimization the aim is to determine a solution to an optimization problem in such a way that the obtained solution is feasible for any realization of the uncertain parameters within the uncertainty set and for the worst-case realization of these uncertain parameters, solution is optimal. In any decision making problem there are three types of robustness (i) constraint robustness (ii) objective robustness (iii) combinational robustness. If uncertainty is present in only constraint parameter, then the problem is defined as constraint robustness problem. If uncertainty is present in only objective function parameter, then the problem is classified as objective robustness problem. If uncertainty is present in both constraint parameter and objective function parameter then it is defined as combinational robustness. Due to effective modeling of uncertain parameters the problem formulated by the robust optimization is tractable and has low computational burden [84].

When degree of robustness is zero, the problem reduces to a deterministic case which solves the problem using the nominal values *i.e.*, expected values of the uncertain coefficients. Degree of robustness can be adjusted according to the trade-off between decision maker's risk preference and the conservatism of the solution [85].

6.5 Proposed Problem Formulation

This section provides the proposed formulation of MG scheduling problem in both Grid connected and Grid isolated mode using robust optimization approach. In this problem, wind and solar power is considered as uncertain parameter that is present in the constraints, therefore the formulated problem has constraint robustness.

6.5.1 Grid Connected Mode

In this mode, problem described in section 4.2 is reformulated by using Robust Optimization as follows:

(i) **Objective Function**

$$\min \sum_{t=1}^{N_t} \left[\sum_{j=1}^{N_j} C_j \left(P_{j,t} + r_{j,t}^u - r_{j,t}^d \right) + P_{grid,t} * \lambda_{grid,t} \right]$$
(6.2)

Since uncertainty is present in the constraint only, the Objective function is given in equation (6.2), in which minimization of total cost of operation is represented as the sum of operating cost of dispatchable generating unit's power, reserves and cost of power exchange between main grid & the MG.

(ii) Power Balance Constraint

$$\sum_{j=1}^{j} P_{j,t} + \sum_{w=1}^{N_w} \left(\sum_{k=1}^{N_k} W_{t,k} + z_k \Gamma_k + \sum_{k=1}^{N_k} q_k \right) + \sum_{w=1}^{N_v} \left(\sum_{k'=1}^{N_{k'}} PV_{t,k'} + z_k \cdot \Gamma_{k'} + \sum_{k'=1}^{N_{k'}} q_{k'} \right) = Pd_t^f, \forall t$$
(6.3)

Power balance constraint is represented in equation (6.3). This constraint states that the sum of scheduled power of dispatchable generating units, wind power units and PV generation must be equal to day-ahead forecasted demand at any time t. But irrespective of the previous formulation, both wind power and solar power are considered as uncertain parameter and modeled in RO framework, where variables like z_k, q_k, Γ_k and $z_{k'}, q_{k'}, \Gamma_{k'}$ are introduced to solve the problem, Where, $\Gamma_k, \Gamma_{k'}$ represent the degree of robustness which controls the robustness of the problem in case of wind power uncertainty and solar power uncertainty respectively and $z_k, z_{k'}$ and $q_k, q_{k'}$ are the dual variables to linearize the non-linearity in the problem for both wind power and solar power uncertainty respectively.

(iii) Robust Constraints

$$z_k + q_k \ge W_{t,k} y_k, \qquad \forall k \neq 0, \, k \in N_k \tag{6.4}$$

$$z_{k'} + q_{k'} \ge P\hat{V}_{t,k'}y_{k'}, \quad \forall k' \neq 0, \, k' \in N_{k'}$$
(6.5)

$$q_k, q_{k'} \ge 0 \qquad \qquad \forall k, \forall k', k \in N_k, k' \in N_{k'}$$
(6.6)

$$y_k, y_{k'} \ge 0 \qquad \forall k, \forall k' \tag{6.7}$$

$$z_k, z_{k'} \ge 0 \qquad \forall k, \forall k' \tag{6.8}$$

Robust constraints are represented by equations (6.4)-(6.8), Where, equation (6.4) and (6.5) states that dual variable z_k, z_k and q_k, q_k are taken into account as the known bounds of wind and solar power in such a way that the sum of both variables is always greater than the wind and solar characterization parameters $\hat{W}_{t,k}y_k$ and $P\hat{V}_{t,k}y_k$. y_k and y_k are represent the auxiliary variable in case of wind power and solar power uncertainty respectively, used to obtain the corresponding linear expressions. However, $\hat{W}_{t,k}$ and $P\hat{V}_{t,k}y_k$ are obtained as the proportionate of the difference of lower and upper bounds of wind and solar power, respectively. Constraint (6.6), (6.7) and (6.8) show that the dual variables used in robust approach must be positive.

(iv) Real time Power Balance Constraint

$$\sum_{j=1}^{N_{j}} \left(r_{j,t}^{u} - r_{j,t}^{d} \right) + \sum_{w=1}^{N_{w}} \left(\Delta P_{w,t} \right) + \sum_{v=1}^{N_{v}} \Delta P V_{v,t} + P_{grid,t} - \Delta P d_{t} = 0, \quad \forall t$$
(6.9)

Due to error in existing forecasting approaches, the actual values of wind power, PV generation and demand may be different from their forecasted values. The system imbalance created due to the error in forecasting in real-time can be compensated by the reserve capacity of dispatchable units or by the power from the main grid depending on the cost of compensation.

(v) Imbalance Constraints

$$\Delta P_{w,t} = P_{w,t}^a - P_{w,t}^f \qquad \forall w, \forall t \tag{6.10}$$

$$\Delta PV_{v,t} = PV_{v,t}^a - PV_{v,t}^f, \quad \forall v, \forall t$$
(6.11)

$$\Delta P_{d,t} = P_{d,t}^a - P_{d,t}^f \qquad \forall d, \forall t \tag{6.12}$$

In real-time, the deviation of wind power, PV solar power generation and system demand from their forecasted values can be calculated by the corresponding imbalance constraints expressed in (6.10) - (6.12). In this work, forecasting error of wind power generation, PV Generation & demand is assumed as a random variable with zero mean and constant standard deviation. The Real value of wind power,

solar/PV power and demand is obtained by multiplying the error coefficient to their forecasted values.

(vi) Dispatchable Units Power Constraint

$$P_j^{\min} \le P_{j,t} \le P_j^{\max} \qquad \forall j, \forall t \tag{6.13}$$

$$r_{j,t}^{u} \le R_{j,t}^{u,\max} \qquad \forall j,\forall t$$
(6.14)

$$r_{j,t}^{d} \le R_{j,t}^{d,\max} \qquad \forall j,\forall t \tag{6.15}$$

$$P_{j,t} + r_{j,t}^u - r_{j,t}^d \ge 0 \qquad \forall j, \forall t$$
(6.16)

$$P_{j,t} + r_{j,t}^u - r_{j,t}^d \le P_{j,t}^{\max} \qquad \forall j, \forall t$$
(6.17)

Equation (6.13) maintains scheduled dispatchable generation unit power within predefined limits, while Constraints equation (6.14) and (6.15) define the scheduled upward and downward reserve of dispatchable generating units. It must be lesser than their maximum upward and downward reserve capacity respectively. Constraints equation (6.16) and (6.17) state that sum of scheduled power of dispatchable units with their respective reserves must be within their specified limits.

(vii) Grid Power Cconstraint

$$P_{grid,t} \le P_{grid,t}^{\max}, \quad \forall t \tag{6.18}$$

Due to transmission network constraints, MG cannot meet the whole demand through grid power when grid prices are lower than operation cost. On the other hand, the complete reserve capacity cannot be sold in the spot market during high market prices. Constraint equation (6.18) ensures that the power exchange between the main grid & MG is less than predefined capacity of the connecting.

6.5.2 Grid Isolated Mode

Since wind and solar power uncertainty is involved in the constraints, the MG generation scheduling problem in this mode is formulated using Robust Optimization approach similar to Grid connected mode differing in the objective function and supply-demand equilibrium constraint. Similar to previous one, in this mode also, it is required to minimize the expenses of the MG. However, more attention is given to meet the demand with stable operation. The problem in the isolated mode is

formulated as:

(i) **Objective Function**

Similar to previously defined grid connected mode under RO Approach, the objective of MO in this mode is also the minimization of total operation cost. It states that the sum of operating cost of dispatchable generating units and cost of demand shedding should be minimized.

$$\min \sum_{t=1}^{N_t} \left[\sum_{j=1}^{N_j} C_j \left(P_{j,t} + r_{j,t}^u - r_{j,t}^d \right) + P_{d,t}^{shed} * \lambda_d^{shed} \right]$$
(6.19)

(ii) Power Balance Constraint

Power balance constraint is same as grid connected mode formulation in RO framework by equation (6.2).

$$\sum_{j=1}^{j} P_{j,t} + \sum_{w=1}^{N_{w}} \left(\sum_{k=1}^{N_{k}} W_{t,k} + z_{k} \Gamma_{k} + \sum_{k=1}^{N_{k}} q_{k} \right) + \sum_{v=1}^{N_{v}} \left(\sum_{k=1}^{N_{k}} PV_{t,k} + z_{k} \Gamma_{k} + \sum_{k=1}^{N_{k}} q_{k} \right) = Pd_{t}^{f}, \forall t$$
(6.20)

Power balance constraint is represented in equation (6.20). This constraint states that the sum of scheduled power of dispatchable generating units, wind power units and PV generation must be equal to day-ahead forecasted demand at any time t. But irrespective of the previous formulation, both wind power and solar power are considered as uncertain parameter and modeled in RO framework, where variables like z_k, q_k, Γ_k and $z_{k'}, q_{k'}, \Gamma_{k'}$ are introduced to solve the problem, Where, $\Gamma_k, \Gamma_{k'}$ represent the degree of robustness which controls the robustness of the problem in case of wind power uncertainty and solar power uncertainty respectively and $z_k, z_{k'}$ and $q_k, q_{k'}$ are the dual variables to linearize the non-linearity in the problem for both wind power and solar power uncertainty respectively.

(iii) Robust Constraints

$$z_k + q_k \ge W_{t,k} y_k, \qquad \forall k \neq 0, \, k \in N_k \tag{6.21}$$

$$z_{k'} + q_{k'} \ge P\hat{V}_{t,k'}y_{k'}, \quad \forall k' \neq 0, \, k' \in N_{k'}$$
(6.22)

$$q_k, q_{k'} \ge 0, \quad \forall k, k', k \in N_k, k' \in N_{k'}$$

$$(6.23)$$

$$y_k, y_k' \ge 0, \quad \forall k, k' \tag{6.24}$$

$$z_k, z_{k'} \ge 0, \quad \forall k, k' \tag{6.25}$$

Robust constraints are represented by equations (6.21)-(6.25), Where, equation (6.21) and (6.22) states that dual variable $z_k, z_{k'}$ and $q_k, q_{k'}$ are taken into account as the known bounds of wind and solar power in such a way that the sum of both variables is always greater than the wind and solar characterization parameters $\hat{W}_{t,k} y_k$ and $P\hat{V}_{t,k'} y_{k''}$. y_k and $y_{k'}$ are represent the auxiliary variable in case of wind power and solar power uncertainty respectively, used to obtain the corresponding linear expressions. However, $\hat{W}_{t,k}$ and $P\hat{V}_{t,k'}$ are obtained as the proportionate of the difference of lower and upper bounds of wind and solar power, respectively. Constraint (6.23), (6.24) and (6.25) show that the dual variables used in robust approach must be positive.

(iv) Real time Power Balance Constraint

$$\sum_{j=1}^{N_{j}} \left(r_{j,t}^{u} - r_{j,t}^{d} \right) + \sum_{w=1}^{N_{w}} \left(\Delta P_{w,t} \right) + \sum_{v=1}^{N_{v}} \Delta P V_{v,t} + P_{d,t}^{Shed} - \Delta P d_{t} = 0, \quad \forall t$$
(6.26)

To maintain system security and reliability, the generation is always equal to system demand at each instant. The system imbalance created due to volatile wind power, PV generation and demand at real-time is balanced by scheduling the reserve capacities of dispatchable generating units and shedding of demand.

(v) Demand Shedding Cconstraint

$$P_{d,t}^{shed} \le P_{d,t} \quad \forall t \tag{6.27}$$

The shedding power is always less than equal to the scheduled demand in dayahead scheduling.

(vi) Imbalance Constraints

$$\Delta P_{w,t} = P_{w,t}^a - P_{w,t}^f, \qquad \forall w, \forall t$$
(6.28)

$$\Delta PV_{v,t} = PV_{v,t}^a - PV_{v,t}^f, \quad \forall v, \forall t$$
(6.29)

$$\Delta P_{d,t} = P_{d,t}^a - P_{d,t}^f, \qquad \forall d, \forall t$$
(6.30)

Chapter-6

In real-time, the deviation of wind power, PV generation and system demand from their forecasted values can be calculated by the corresponding imbalance constraints expressed in (6.28)-(6.30). In this work, forecasting error of wind power generation, PV Generation & demand is assumed as a random variable with zero mean and constant standard deviation. The Real value of wind power, PV power and demand is obtained by multiplying the error coefficient to their forecasted values.

(vii) Dispatchable Units Power Constraint

$$P_j^{\min} \le P_{j,t} \le P_j^{\max}, \qquad \forall j, \forall t$$
(6.31)

$$r_{j,t}^{u} \le R_{j,t}^{u,\max}, \qquad \forall j, \forall t$$
(6.32)

$$r_{j,t}^{d} \le R_{j,t}^{d,\max}, \qquad \forall j, \forall t$$
(6.33)

$$P_{j,t} + r_{j,t}^u - r_{j,t}^d \ge 0, \qquad \forall j, \forall t$$
 (6.34)

$$P_{j,t} + r_{j,t}^u - r_{j,t}^d \le P_{j,t}^{\max}, \qquad \forall j, \forall t$$
(6.35)

Equation (6.31) maintains scheduled dispatchable generation unit power within predefined limits, while Constraints (6.32) and (6.33) define the scheduled upward and downward reserve of dispatchable generating units. It must be lesser than their maximum upward and downward reserve capacity respectively. Constraints (6.34) and (6.35) state that sum of scheduled power of dispatchable units with their respective reserves must be within their specified limits.

6.6 Proposed Simulation Procedure

The following simulation procedure (as shown in Fig. 6.2) is adopted to solve the formulated Robust Optimization based MG generation scheduling problem for both grid connected and isolated mode:

- Step 1: Collect historical data: Collect the historical data of wind speed and PV solar power generation for a specific location where wind and solar units are located. The wind speed is converted in power using power curve of standard turbine models.
- Step 2: ARIMA model parameters estimation: From collected data, parameters of ARIMA model are estimated. Order of AR and MA terms are determined by the observation of Auto Correlation Function (ACF) and Partial Auto
Correlation Function (PACF) plot, respectively. The coefficients of AR and MA terms and the variance of error signal are determined by Least Square method [159].

- Step 3: Load Dispatchable Units and Demand Data: Collect specification data of dispatchable generation units including their cost parameters, minimum and maximum generation capacities, and upward and downward reserve capacities. Hourly forecasted demand profile is uploaded for one day generation scheduling of MG.
- Step 4: Initialize Time Counter: Here 24 time periods are considered to solve MG generation scheduling under both solar and wind power uncertainty. Start with time t = 1.
- Step 5: Forecast Wind and Solar Power Interval: To forecast upper and lower limits of wind and solar power, ARIMA model expressed in (6.1) is simulated for each time instant.
- Step 6: Select Mode of Operation of MG: Select either grid connected or isolated mode for generation scheduling of MG. If grid connected mode is selected, go to next step. Otherwise, grid isolated mode is selected, go to Step 13.
- Step 7: Load Market Prices: Load forecasted electricity market price for optimal generation scheduling of MG according to grid prices.
- Step 8: Robust optimization Parameter Initialization: For grid connected MG' generation scheduling, the initial output of wind and PV solar power unit is set to be $W_t = W_t^{\text{max}}$ and $PV_t = PV_t^{\text{max}}$, degree of robustness is $\Gamma_k = N_t$ with incremental factor G^k , which takes increasing values in the interval [0, 1] in steps of 0.01, denoted as δ [158].
- Step 9: Iteration Counter Initialization: The total number of iterations N_k is defined and the counter is initialized with k = 1.
- Step 10: Uncertainty Characterization: The upper and lower limits of wind and solar power are obtained from Step 5 for grid connected MG generation scheduling. The wind and solar power are set in the iteration as $\hat{W}_{t,k} = G^k \left(W_t^{\max} - W_t^{\min} \right)$ and $P\hat{V}_{t,k'} = G^{k'} \left(PV_t^{\max} - PV_t^{\min} \right)$, respectively. Value of wind and solar power is limited in the interval $[W_t^{\max}, W_t^{\max} - \hat{W}_{t,k}]$ and $[PV_t^{\max}, PV_t^{\max} - P\hat{V}_{t,k'}]$.

- Step 11: Solve Grid Connected MG Generation Scheduling Problem: Formulated MG scheduling problem for grid connected mode in Section 6.5.1 is solved and optimal results are obtained in the form of optimal scheduling of dispatchable units, optimal scheduling of upward and downward reserve, optimal Main grid to MG power exchange..
- Step 12: Check Iteration Counter: If iteration $k \le N_k$, update the range of incremental factor G^k by step δ and repeat Steps 10 and 11. Otherwise, go to step 18.
- Step 13: Robust optimization Parameter Initialization: For solving grid isolated MG generation scheduling problem using robust optimization based approach the initial outputs of wind and solar power unit are set to be $W_t = W_t^{\max}$ and $PV_t = PV_t^{\max}$, degree of robustness is $\Gamma_k = N_t$ with incremental factor G^k , which takes increasing values in the interval [0, 1] in steps of 0.01, denoted as δ [158].
- Step 14: Iteration Counter Initialization: The total number of iterations N_k is defined and the counter is initialized with k = 1.
- Step 15: Uncertainty Characterization: The upper and lower limits of wind and solar power are obtained from Step 5. The wind power and solar power are set in the iteration as $\hat{W}_{t,k} = G^k \left(W_t^{\max} W_t^{\min} \right)$ and $P\hat{V}_{t,k} = G^k \left(PV_t^{\max} PV_t^{\min} \right)$. Value of wind and solar power are limited in the interval $[W_t^{\max}, W_t^{\max} \hat{W}_{t,k}]$ and $[PV_t^{\max}, PV_t^{\max} P\hat{V}_{t,k}]$, respectively.
- Step 16: Solve Grid Isolated MG Generation Scheduling Problem: Formulated MG scheduling problem for grid isolated mode is solved and optimal results are obtained in the form of optimal scheduling of dispatchable units, optimal scheduling of upward and downward reserve, optimal demand shedding.
- Step 17: Check Iteration Counter: If iteration $k \le N_k$, update the range of incremental factor G^k by step δ and repeat Steps 15 and 16. Otherwise, go to next step.
- Step 18: Check Time Counter: If the desired time period counter, i.e., 24, is achieved, go to the next step; otherwise, update t=t+1 and go to Step 4.
- Step 19: Published results: Show obtained optimal cost of MG along with optimal value of scheduled dispatch-able unit generation and reserves.
- Step 20: End.



Fig. 6.2 Flow chart of proposed algorithm

6.7 Result and Discussions

6.7.1 Data

In this study, installed capacity of PV unit and wind units are considered as 200 kW and 1.34 MW respectively similar to previous chapters. Wind units consists of ENERCON turbine model whose parameters are given in detail in manufacturer

Chapter-6

database [160]. The historical wind speed data of the duration of 27.04.2006 to 08.06.2007, are used in this study from publically available database at Illinois Institute of Rural Affair, USA. [161]. For same site the historical solar irradiation data is obtained from NREL for same duration [162]. Along with renewable units, 10 dispatchable units are considered whose parameters are discussed in Table 4.1 (chapter 4). Hourly grid prices and demand profile are considered as Table 6.1.

Time (h)	Price (\$/kw)	Demand (pu)
1	0.04836	0.784
2	0.04461	0.754
3	0.043695	0.739
4	0.044535	0.736
5	0.051765	0.756
6	0.069	0.814
7	0.08238	0.912
8	0.080115	0.977
9	0.08802	0.991
10	0.090045	0.997
11	0.09258	1
12	0.088725	0.997
13	0.0906	0.992
14	0.09054	0.989
15	0.08595	0.981
16	0.07908	0.971
17	0.07461	0.964
18	0.06552	0.956
19	0.062415	0.949
20	0.065775	0.945
21	0.07266	0.962
22	0.0609	0.950
23	0.05253	0.884
24	0.044925	0.822

Table 6.1: Daily Market Price and Demand Profile

Similar to earlier chapters, ARIMA model is used for modeling wind and solar power uncertainty. For uncertainty modeling using ARIMA model first the order of suitable ARIMA model on the basis of collected historical data is determined. Order of AR and MA terms are obtained by the observation of ACF and PACF plot. Forecasted limits of both wind power and solar power for 24 hours with upper and lower bounds are similar to previous figures discussed in Fig. 4.4 (Chapter 4) and Fig. 5.8 (Chapter 5). On a 95% confidence interval forecasted limits are used in proposed robust optimization based MG scheduling approach.

6.7.2 Case I: Microgrid Operation in Grid Connected Mode

In this case, MG operation in grid-connected mode is considered for simulation using proposed robust optimization method. Fig. 6.3 shows the scheduled power of dispatchable units in MG during grid connected mode of operation. From the figure, it can be observed that it will be beneficial to schedule maximum capacity from the unit with lower cost than the unit with higher cost.



Fig. 6.3 Scheduled power of dispatchable units

Fig. 6.4 and Fig. 6.5 show the obtained scheduled upward and downward reserves of dispatchable units respectively. From the figures, it can be observed that any dispatchable unit is scheduled for one type of reserve requirement at any particular instant of time t. It can be observed that at the time of exporting the power the most

costly dispatchable units are scheduled to their downward reserve while in case of importing the power these units are scheduled to their upward reserve.



Fig. 6.4 Scheduled upward reserve



Fig. 6.5 Scheduled downward reserve

Fig. 6.6 shows power exchange between the main grid and the MG in grid connected mode. From the figure, it can be observed that from Hour 6 to 17 and 21,

power is being exported to the grid and in remaining hours, power is being imported from the grid.



Fig. 6.6 Power imported/exported to/from grid

It can be concluded that, when grid price is higher, MG can export their reserve capacity to grid and vice-versa to reduce its operating cost.

6.7.3 Case II: Microgrid Operation in Grid Isolated Mode

Similarly, in this case, MG operation in grid-isolated mode is considered using proposed robust optimization method. The cost of demand shedding is considered as \$3.5/kWh according to Federal Energy Regulation Commission (FERC), Electric Tariff [164]. Similar to the previous case, Figs 6.7, 6.8, and 6.9 show the scheduled dispatchable unit power, upward and downward reserve capacity respectively using proposed robust optimization approach. Demand shedding in Case II is shown in Fig. 6.10. From the figure, it is observed that demand shedding event occurs in Hours 10, 16 and 24. The capacity of dispatchable units is fully utilized therefore leading to demand shedding. From Fig. 6.8 and 6.9, it can be visualized that MG can utilize its upward reserve capacity during peak demand to reduce demand shedding events.



Fig. 6.7 Dispatchable unit scheduled power



Fig. 6.8 Scheduled upward reserve



Fig. 6.9 Scheduled downward reserve



Fig. 6.10 Demand shedding

6.7.4 Discussion

A comparative analysis on MG operation is carried out. For this purpose, the same problem is simulated using deterministic and stochastic programming approach. In deterministic approach, wind power output is simply forecasted using ARIMA model and the forecasted value of wind power is considered to compute the scheduling of MG.

Approaches	Daily MG Operation Cost (\$)	
	Case I	Case II
Deterministic	857.9346	1500.230
Stochastic	814.2935 (-5.08 %)	1284.09 (-14.40 %)
Proposed	747.7512 (-12.84%)	1195.191 (-20.33 %)

Table 6.2: Comparative Analysis of MG Daily Operation Cost

In stochastic programming approach, the wind power and solar power uncertainty is modeled by generating a large number of scenarios (i.e. 1000) and then reducing them into a smaller number of scenarios (i.e.25) as in [56]. The MG operation cost obtained using proposed robust optimization approach is compared with both deterministic and stochastic approaches as shown in Table 6.2. From the Table 6.2, it is observed that in both cases (i.e. Grid connected operation of MG and Grid isolated operation of MG) the daily operating cost is significantly lower in the proposed approach than any other existing approach. It is also observed from Table 6.2 that, The reduction in daily cost using proposed approach, in Case II (grid isolated mode) is higher than compared to Case I (grid connected mode). This is because in Case II, small reduction in demand shedding event results in large reduction in total cost of operation. From Table 6.2 in the proposed approach the cost of MG operation is higher in grid-isolated mode than in grid-connected mode. This is due the fact that the reserves and dispatchable units are fully utilized to meet the demand.



Fig. 6.11 Impact of degree of robustness on operational cost of MG.

The impact of degree of robustness on operation cost, in both cases can be evaluated by varying the degree of robustness from zero to 25 during simulation. The obtained per unit cost of operation are shown in Fig. 6.11. Per unit cost of operation, at any degree of robustness is obtained by dividing its actual value by the reference value, which represents the cost of operation at zero degree of robustness. From Fig. 6.11, it can be seen that the optimal value of degree of robustness is 12, where per unit cost of operation is minimum. At this degree of robustness, optimal condition is obtained because in this problem both wind power and solar power are considered as an uncertain parameter and its uncertainty is considered for 24 hours in case of wind power and 12 hours for the case of solar/PV power. This degree of robustness takes into account all possible deviations of the uncertain parameter for whole day. Additionally, at zero degree of robustness the cost of operation obtained using proposed robust optimization based approach is equal to the cost of operation obtained using deterministic approach.

6.8 Summary

A robust optimization based approach has been proposed in this chapter for optimal generation scheduling of MG in both grid-connected and grid-isolated modes. Both wind power and solar power uncertainty have been modeled through interval forecasting using ARIMA model. A comparative study on daily MG cost of operation using proposed robust optimization based approach with deterministic and stochastic approach has been performed. A significant reduction in cost of operation clearly shows strength of proposed robust optimization based approach in MG generation scheduling in the deregulated environment. The impact of degree of robustness on the cost of operation in both cases has also evaluated and compared with existing methods. The proposed approach will be more advantageous in incorporating different type of uncertainties such as demand uncertainty and grid price uncertainty. This work may also be enhanced by incorporating multi-micro grids, different type of consumers and considering different technical constraints. Chapter-6

Chapter 7

Conclusions and Suggestions for Further Work

HIS chapter summarizes the thesis by outlining the major contributions and findings from the research. It further proposes some future works that can be done to improve MG generation scheduling problem.

CONCLUSIONS AND SUGGESTIONS FOR FURTHER WORK

7.1 General

Penetration of renewable energy resources to the transmission and distribution power systems is continuously increasing throughout the world for mitigation of the present energy and environmental challenges in cost effective and sustainable manner. Among the renewable energy sources, penetration of the wind and solar power is increasing rapidly due to their widespread availability and mature technology. MGs are efficient way to integrate such renewable sources as Distributed Energy Resources (DERs) into the distribution grids. However, integration of such energy sources causes various technical and economic challenges in the operation of MGs, due to random nature of wind and solar power. Thus, energy management of MGs is complex decision making problem that is main focus of presented thesis. Complexity of MGs Energy Management System (EMS) is further increased due to restructuring of power system and market price volatility due to MG based EMS throughout the world. Therefore, development of efficient approaches for MG management under different uncertainties is required.

This research work attempts to develop efficient approaches for MG management with an objective to minimize the total cost of operation subject to different constraint parameters considering renewable power uncertainties. This addresses multiple concerns associated with generation scheduling problem of MG in both grid connected and grid isolated mode in the robust optimization framework. Effectiveness of proposed robust optimization based approach over stochastic and deterministic approach has been evaluated through comparative analysis.

A summary of main findings of the research work carried out in this thesis and future scope in this area are presented in subsequent sections.

7.2 Summary of Significant Findings

The research work undertaken in this thesis is initiated by developing an understanding of MG generation scheduling problem in both grid connected and grid isolated mode considering wind power uncertainty. A robust optimization based approach has been proposed for optimal generation scheduling of MG in both grid-connected and grid-isolated modes. Wind power uncertainty has been modeled through interval forecasting using ARIMA model. Further a comparative study on daily MG cost of operation using proposed robust optimization based approach with deterministic and stochastic approach has been performed. For this purpose, the same problem is simulated using deterministic and stochastic programming approach. In deterministic approach, wind power output is forecasted using ARIMA model and the forecasted value of wind power is considered to compute the scheduling of MG. The impact of degree of robustness on total operation cost of MG in both cases has also been evaluated by varying the degree of robustness. Per unit cost of operation, at any degree of robustness is obtained by dividing its actual value by the reference value, which represents the cost of operation at zero degree of robustness.

Obtained results, thus show a significant reduction in total operating cost, which clearly highlights the strength of proposed robust optimization based approach in MG generation scheduling problem in the deregulated environment. From the analysis based on impact of degree of robustness on the MG operating cost, it was observed that the optimal value of degree of robustness is 24, where per unit cost of MG operation is minimum. At this degree of robustness, optimal condition is obtained because the uncertainty of wind power is considered for 24 hours. This degree of robustness takes into account all possible deviations of the uncertain parameter for whole day. Additionally, at zero degree of robustness the cost of operation obtained using proposed robust optimization based approach is equal to the cost of operation obtained using deterministic approach.

This thesis further contributes by the enhancement of the earlier proposed robust optimization based approach for optimal generation scheduling of MG in both gridconnected and grid-isolated modes considering solar power uncertainty. A stochastic optimization based approach has also been proposed for optimal generation scheduling of both grid connected and isolated MG considering solar power uncertainty. For both stochastic and robust optimization based approaches, solar power uncertainty has been modeled through a number of scenarios and interval forecasting using ARIMA model, respectively. Further this research work contributes by presenting a comparative study on daily MG cost of operation using proposed robust optimization based approach with deterministic and stochastic approach has been performed. Obtained results show a significant reduction in cost of operation, which clearly highlights the strength of proposed robust optimization based approach over stochastic and deterministic approaches in MG generation scheduling in the deregulated environment. The impact of degree of robustness on the cost of operation in both cases has also been evaluated and compared with existing methods.

Finally, this thesis contributes by proposing a robust optimization based approach for optimal generation scheduling of MG in both grid-connected and grid-isolated modes considering wind and solar power uncertainties. Wind power and solar power uncertainties have been modeled through interval forecasting using ARIMA model. A comparative analysis on daily cost of operation of MG using proposed robust optimization based approach with deterministic and stochastic approach has been performed. A significant reduction in cost of operation in the proposed approach shows the strength of proposed robust optimization based approach in MG generation scheduling in the deregulated environment. The impact of degree of robustness on the cost of operation in both cases has also evaluated and compared with existing methods.

MG operation under renewable power uncertainties is a recent relevant and attractive area of research. The area lies at the confluence of multiple streams of knowledge, where electrical engineering, economics and regulation are merged. Such studies help to develop an understanding of inter-relation between issues arising from different fields of knowledge. The present work provides an understanding of MGs generation scheduling for both grid connected and isolated modes under renewable power uncertainties. MG generation scheduling problem considering wind and solar power uncertainties has been modeled in robust and stochastic optimization framework. Proposed work handles the multiple issues of MG generation scheduling problem that would continue to attract research interest.

7.3 Suggestions for Further Work

Research and development is a continuous process. Each step of research work opens many avenues for future research. As a consequence of the investigations carried out, a variety of issues pertaining to MG's generation scheduling under renewable power uncertainties has been handled. Though, there are several inter-related issues to be resolved, some of them require urgent attention due to their wide implications. Following are some of the aspects, identified as promising area for future research work in the realm of study:

- i. ARIMA model is used to model wind and solar power uncertainty through interval forecasting. Modelling of such uncertainties could be further improved by using advanced forecasting models and incorporating information about numerical weather prediction and seasonal variations.
- Demand and grid price uncertainty are modelled deterministically in the present work. Such uncertainties could be modelled similar to wind and solar power uncertainties.
- iii. The presented robust optimization based formulation of MGs' generation scheduling problem can be enhanced by considering network constraints, transmission losses and security constraints in both DC and AC power flow framework.
- Energy storage is an important component of MGs operation particularly in grid isolated MG. Further MG generation scheduling problem could be modeled considering energy storage.
- v. Proposed models and solution algorithms could be further illustrated by practical case studies based on large MGs in Indian power system.
- vi. Consideration of multi-micro grids and different type of consumers could be considered as an extension to the work. This would be another important extension of proposed work.

Apart from these issues for future research, MG's generation scheduling problem under renewable power uncertainty poses several unique operation and planning challenges for MG operators as well as for system operators. These challenges create ample opportunities for researchers in this area.

References

REFERENCES

- R. Lasseter, A. Akhil, C. Marnay, J. Stephens, J. Dagle, R. Guttromson, A. S. Meliopoulous, R. Yinger, and J. Eto, "Integration of distributed energy resources," The CERTS Microgrid Concept, Ed, 2002.
- [2] S. Bossart, "DOE Perspective on Microgrids," in Proc. Advanced Microgrid Concepts & Technologies Workshop, 2012.
- [3] S. Van Broekhoven, N. Judson, S. Nguyen, and W. Ross, "Microgrid Study: Energy Security for DoD Installations," DTIC Document, 2012.
- [4] W.-J. Lee and D. Wetz, "Development of a Smart MicroGrid Testbed," 2013.
- [5] Roozbehani, Mardavij, Munther A. Dahleh, and Sanjoy K. Mitter, "Volatility of power grids under real-time pricing," *IEEE Trans Power Syst.*, vol. 27, no. 4, pp. 1926-1940, 2012.
- [6] Basu, Ashoke Kumar, et al. "Planned scheduling for economic power sharing in a CHP-based micro-grid," *IEEE Trans Power Syst.*, vol. 27, no. 1, pp. 30-38, 2012.
- [7] Derakhshandeh, S. Yaser, et al. "Coordination of generation scheduling with PEVs charging in industrial microgrids," *IEEE Trans Power Syst.*, vol. 28, no. 3, pp. 3451-3461, 2013.
- [8] Crisostomi, Emanuele, et al. "Plug-and-play distributed algorithms for optimized power generation in a microgrid," *IEEE Trans Smart Grid*, vol. 5, no. 4, pp. 2145-2154, 2014.
- [9] Tsikalakis, Antonis G., and N. D. Hatziargyriou, "Centralized control for optimizing microgrids operation," *in Proc. 2011 IEEE/PES General Meeting*, 2011.
- [10] Sanseverino, Eleonora Riva, et al. "An execution, monitoring and replanning approach for optimal energy management in microgrids," *Energy*, vol. 36, no. 5, pp. 3429-3436, 2011.
- [11] Mohammadi, Sirus, Soodabeh Soleymani, and Babak Mozafari, "Scenario-based stochastic operation management of microgrid including wind, photovoltaic, micro-turbine, fuel cell and energy storage devices," *International Journal of Electrical Power & Energy Systems*, vol. 54, pp. 525-535, 2014.
- [12] Malekpour, Ahmad Reza, et al, "Multi-objective stochastic distribution feeder reconfiguration in systems with wind power generators and fuel cells using the

point estimate method," IEEE Trans Power Syst., vol. 28, no. 2, pp. 1483-1492, 2013.

- [13] Chen, Ci, et al. "Smart energy management system for optimal microgrid economic operation," *IET Renewable Power Generation*, vol. 5, no. 3, pp. 258-267, 2011.
- [14] Xu, Yixing, Le Xie, and Chaman Singh, "Optimal scheduling and operation of load aggregators with electric energy storage facing price and demand uncertainties," *in Proc.* IEEE *North American Power Symposium*, 2011.
- [15] ERCOT, "ERCOT Monthly Operational Overview," Ed. 2010.
- [16] Farhangi, Hassan, "The path of the smart grid," IEEE Power and energy magazine, vol. 8, no. 1, pp. 18-28, 2010.
- [17] Schnitzer, Daniel, et al, "Microgrids for rural electrification," *Newsletter* (2014).
- [18] Logenthiran, Thillainathan, et al. "Multi-Agent System (MAS) for short-term generation scheduling of a microgrid," in Proc. 2010 IEEE International Conference Sustainable Energy Technologies (ICSET), 2010.
- [19] Kim, Hak-Man, Tetsuo Kinoshita, and Myong-Chul Shin, "A multiagent system for autonomous operation of islanded microgrids based on a power market environment," *Energies*, vol. 3, no. 12, pp.1972-1990, 2010.
- [20] Kavousi-Fard, Abdollah, "A new fuzzy-based feature selection and hybrid TLA– ANN modelling for short-term load forecasting," *Journal of Experimental & Theoretical Artificial Intelligence*, vol. 25, no. 4, pp. 543-557, 2013.
- [21] Kuzle, Igor, Ninoslav Holjevac, and Tomislav Capuder, "Model Predictive Control for Scheduling of Flexible Microgrid Systems," *in Proc. 9th Conference on Sustainable Development of Energy, Water and Environment Systems– SDEWES, Venice-Istanbul.* 2014.
- [22] Dall'Anese, Emiliano, Hao Zhu, and Georgios Giannakis, "Distributed optimal power flow for smart microgrids," *IEEE Trans. Smart Grid*, vol. 4, no. 3, pp. 1464-1475, 2013.
- [23] Nguyen, Phuong H., Wil L. Kling, and Paulo F. Ribeiro, "A game theory strategy to integrate distributed agent-based functions in smart grids," *IEEE Trans. Smart Grid*, vol. 4, no. 1, pp. 568-576, 2013.
- [24] Mohamed, Faisal A., and Heikki N. Koivo, "Multiobjective optimization using Mesh Adaptive Direct Search for power dispatch problem of microgrid," Int. J Elect. Power & Energ Syst., vol. 42, no. 1, pp.728-735, 2012.

- [25] Roy, Kallol, and Kamal Krishna Mandal, "Hybrid optimization algorithm for modeling and management of micro grid connected system," *Frontiers in Energy*, vol. 8, no. 3, pp. 305-314, 2014.
- [26] Al-Saedi, Waleed, et al, "Power quality enhancement in autonomous microgrid operation using particle swarm optimization," Int. J Elect. Power & Energ Syst.vol. 42, no. 1, pp. 139-149, 2012.
- [27] Hatziargyriou, Nikos. *Operation of Multi-Microgrids*. Wiley-IEEE Press, 2014.
- [28] Jacobson, Mark Z., and Mark A. Delucchi, "Providing all global energy with wind, water, and solar power, Part I: Technologies, energy resources, quantities and areas of infrastructure, and materials," *Energy Policy*, vol. 39, no. 3, pp. 1154-1169, 2011.
- [29] Wang, M. Q., and H. B. Gooi, "Spinning reserve estimation in microgrids," *IEEE Trans. Power Syst.*, vol. 26, no. 3, pp.1164-1174, 2011.
- [30] Ahn, Seon-Ju, et al. "Power scheduling of distributed generators for economic and stable operation of a microgrid," *IEEE Trans. Power Syst.* vol. 4, no. 1, pp. 398-405, 2013.
- [31] Liao, Gwo-Ching. "Solve environmental economic dispatch of Smart MicroGrid containing distributed generation system–Using chaotic quantum genetic algorithm," *Int J Elect Power & Energy Syst.*, vol.43, no. 1, pp.779-787, 2012.
- [32] Niknam, Taher, Rasoul Azizipanah-Abarghooee, and Mohammad Rasoul Narimani. "An efficient scenario-based stochastic programming framework for multi-objective optimal micro-grid operation," *Applied Energy*, vol. 99, pp, 455-470, 2012.
- [33] Cacuci, Dan G., Mihaela Ionescu-Bujor, and Ionel Michael Navon. Sensitivity and uncertainty analysis, volume II: applications to large-scale systems. vol. 2. CRC press, 2005.
- [34] Patel, Mukund R. Wind and solar power systems: design, analysis, and operation. CRC press, 2005.
- [35] Brown, Barbara G., Richard W. Katz, and Allan H. Murphy. "Time series models to simulate and forecast wind speed and wind power," *Journal of climate and applied meteorology*, vol. 23, no. 8, pp.1184-1195, 1984.
- [36] Bacher, Peder, Henrik Madsen, and Henrik Aalborg Nielsen, "Online short-term solar power forecasting," *Solar Energy*, vol. 83, no. 10, pp.1772-1783, 2009.

- [37] Alexiadis, M. C., et al. "Short-term forecasting of wind speed and related electrical power," *Solar Energy*, vol. 63, no. 1, pp. 61-68, 1998.
- [38] Hatziargyriou, Nikos, et al. "Microgrids," IEEE Power and Energy Magazine, vol.5, no. 4, pp. 78-94, 2007.
- [39] Pudjianto, Danny, et al. "Investigation of regulatory, commercial, economic and environmental issues in microgrids," in Porc. IEEE International Conference on Future Power Systems, 2005.
- [40] Venkataraman, Giri, and Chris Marnay, "A larger role for microgrids," *IEEE Power and Energy Magazine*, vol.6, no. 3, pp. 78-82, 2008.
- [41] Salam, A. A., A. Mohamed, and M. A. Hannan, "Technical challenges on microgrids," ARPN Journal of engineering and applied sciences, vol. 3, no. pp. 64-69, 2008.
- [42] Wang, Chengshan, and Peng Li, "Development and Challenges of Distributed Generation, the Micro-grid and Smart Distribution System," *Automation of Electric Power Systems*, vol. 2, pp. 4-10, 2010.
- [43] Basu, Ashoke Kumar, et al. "Microgrids: Energy management by strategic deployment of DERs—A comprehensive survey," *Renewable and Sustainable Energy Reviews*, vol.15, no. 9, pp. 4348-4356, 2011.
- [44] Su, Wencong, Zhiyong Yuan, and Mo-Yuen Chow, "Microgrid planning and operation: Solar energy and wind energy," in Proc. 2010 IEEE PES General Meeting, 2010.
- [45] Colson, C. M., and M. H. Nehrir, "Algorithms for distributed decision-making for multi-agent microgrid power management," *in Proc.* 2011 IEEE PES *General Meeting*, 2011.
- [46] Kim, Hak-Man, and Tetsuo Kinoshita. A new challenge of microgrid operation Security-Enriched Urban Computing and Smart Grid. Springer Berlin Heidelberg, pp. 250-260, 2010.
- [47] Firestone, Ryan, and Chris Marnay, "Energy manager design for microgrids," Lawrence Berkeley National Laboratory, 2005.
- [48] Siddiqui, Afzal S., and Chris Marnay, "Distributed generation investment by a microgrid under uncertainty," *Energy*, vol. 33, no. 12, pp.1729-1737, 2008.
- [49] Khodaei, Amin, "Microgrid optimal scheduling with multi-period islanding constraints," *IEEE Trans. Power Syst.*, vol. 29, no. 3, pp.1383-1392, 2014.

- [50] Izadbakhsh, Maziar, et al. "Short-term resource scheduling of a renewable energy based micro grid," *Renewable Energy*, vol. 75, pp. 598-606, 2015.
- [51] Wu, C. X., F. S. Wen, and Y. L. Lou, "The existed problems and possible solutions of micro-grid based on distributed generation," in Proc. IEEE Third International Conference on Electric Utility Deregulation and Restructuring and Power Technologies, 2008.
- [52] Alvial-Palavicino, Carla, et al. "A methodology for community engagement in the introduction of renewable based smart microgrid," *Energy for sustainable development*, vol.15, no. 3, pp.314-323, 2011.
- [53] Niknam, Taher, Faranak Golestaneh, and Ahmadreza Malekpour, "Probabilistic energy and operation management of a microgrid containing wind/photovoltaic/fuel cell generation and energy storage devices based on point estimate method and self-adaptive gravitational search algorithm," *Energy*, vol. 43, no. 1, pp. 427-437, 2012.
- [54] Liang, Hao, and Weihua Zhuang, "Stochastic modeling and optimization in a microgrid: A survey," *Energies*, vol. 7, no. 4, pp. 2027-2050, 2014.
- [55] Dupačová, Jitka, Nicole Gröwe-Kuska, and Werner Römisch, "Scenario reduction in stochastic programming," *Mathematical programming*, vol.95, no. 3, pp.493-511, 2003.
- [56] Sharma, Kailash Chand, Prerna Jain, and Rohit Bhakar, "Wind power scenario generation and reduction in stochastic programming framework," *Electric Power Components and Systems*, vol.41, no. 3, pp. 271-285, 2013.
- [57] Kaut, Michal, and Stein W. Wallace, "Evaluation of scenario-generation methods for stochastic programming," *Pacific Journal of Optimization*, vol. 3, no. 2, pp. 257-271, 2007.
- [58] Morales, Juan M., et al. "Scenario reduction for futures market trading in electricity markets," *IEEE Trans. Power Syst.*, vol. 24, no. 2, pp.878-888, 2009.
- [59] Mulvey, John M., and A. Eric Thorlacius, "The Towers Perrin global capital market scenario generation system," Worldwide asset and liability modeling, vol.10, pp. 286, 1998.
- [60] Zhou, Qun, Leigh Tesfatsion, and Chen-Ching Liu, "Scenario generation for price forecasting in restructured wholesale power markets," in Proc. 2009 IEEE PES Power Systems Conference and Exposition, 2009.

- [61] Di Domenica, Nico, et al. "Stochastic programming and scenario generation within a simulation framework: an information systems perspective," *Decision Support Systems*, vol. 42, no. 4, pp. 2197-2218, 2007.
- [62] Sutcliffe, Alistair, "Scenario-based requirements engineering," in Proc. 11th IEEE international conference on Requirements engineering, 2003.
- [63] Foley, Aoife M., et al. "Current methods and advances in forecasting of wind power generation," *Renewable Energy*, vol. 37, no. 1, pp. 1-8, 2012.
- [64] Ma, Xi-Yuan, Yuan-Zhang Sun, and Hua-Liang Fang, "Scenario generation of wind power based on statistical uncertainty and variability," *IEEE Trans. Sustainable Energy*, vol. 4, no. 4, pp. 894-904, 2013.
- [65] Di Domenica, Nico, et al. "Scenario generation for stochastic programming and simulation: a modelling perspective," *IMA Journal of Management Mathematics*, vol. 20, pp.1, 1-38, 2009.
- [66] Kaut, Michal, "Scenario tree generation for stochastic programming: Cases from finance," Ph.D. Dissertation, Department of Mathematical Sciences, Norwegian University of Science and Technology, 2003.
- [67] Taylor, James W., Patrick E. McSharry, and Roberto Buizza, "Wind power density forecasting using ensemble predictions and time series models," *IEEE Trans. Energy Conv.*, vol. 24, no. 3. pp. 775-782, 2009.
- [68] Cho, M. Y., J. C. Hwang, and C. S. Chen, "Customer short term load forecasting by using ARIMA transfer function model," *in Proc. IEEE Int. Conf. on Energy Management and Power Delivery*, vol. 1, 1995.
- [69] Matevosyan, Julija, and Lennart Söder, "Minimization of imbalance cost trading wind power on the short-term power market," *IEEE Trans. Power Syst.* vol. 21, no. 3, pp.1396-1404, 2006.
- [70] Bludszuweit, Hans, José Antonio Domínguez-Navarro, and Andrés Llombart,
 "Statistical analysis of wind power forecast error," *IEEE Trans. Power Syst.*,
 vol.23, no. 3, pp. 983-991, 2008.
- [71] Pinson, Pierre, and Robin Girard, "Evaluating the quality of scenarios of short-term wind power generation," *Applied Energy*, vol. 96, no. 12-20, 2012.
- [72] Heitsch, Holger, and Werner Römisch, "Scenario reduction algorithms in stochastic programming," *Computational optimization and applications*, vol. 24, no. 2, pp. 187-206, 2003.

- [73] Feng, Yonghan, and Sarah M. Ryan, "Scenario construction and reduction applied to stochastic power generation expansion planning," *Computers & Operations Research*, vol. 40, no. 1, pp. 9-23, 2013.
- [74] Römisch, W, "Approximations of stochastic programs. Scenario tree reduction and construction," *GAMS Workshop, Heidelberg*, 2003.
- [75] Küchler, Christian, and Stefan Vigerske, "Decomposition of Multistage Stochastic Programs with Recombining Scenraio Trees," *Stochastic Programming E-Print Series 9*, 2007.
- [76] Razali, NM Muhamad, and A. H. Hashim, "Backward reduction application for minimizing wind power scenarios in stochastic programming," in Proc. IEEE 4th International Power Engineering and Optimization Conference (PEOCO), 2010.
- [77] García-González, Javier, et al. "Stochastic joint optimization of wind generation and pumped-storage units in an electricity market," *IEEE Trans. Power Syst.*, vol. 23, no. 2, pp. 460-468, 2008.
- [78] Moghaddam, Amjad Anvari, et al. "Multi-objective operation management of a renewable MG (micro-grid) with back-up micro-turbine/fuel cell/battery hybrid power source," *Energy*, vol. 36, no. 11, pp. 6490-6507. 2011.
- [79] Palma-Behnke, Rodrigo, et al. "A microgrid energy management system based on the rolling horizon strategy," *IEEE Trans. Smart Grid*, vol.4, no. 2, pp. 996-1006, 2013.
- [80] Aghaei, Jamshid, and Mohammad-Iman Alizadeh, "Multi-objective selfscheduling of CHP (combined heat and power)-based microgrids considering demand response programs and ESSs (energy storage systems)," *Energy*, vol. 55, pp. 1044-1054, 2013.
- [81] Zakariazadeh, Alireza, Shahram Jadid, and Pierluigi Siano, "Smart microgrid energy and reserve scheduling with demand response using stochastic optimization," *Int. J Elect. Power & Energy Syst.*, vol. 63, pp. 523-533, 2014.
- [82] Siano, Pierluigi, et al. "Real time operation of smart grids via FCN networks and optimal power flow," *IEEE Trans. Indust Infor.*, vol.8, no.4, pp. 944-952, 2012.
- [83] Khodayar, Mohammad E., Masoud Barati, and Mohammad Shahidehpour, "Integration of high reliability distribution system in microgrid operation," *IEEE Trans. Smart Grid*, vol. 3, no. 4, pp. 1997-2006, 2012.

- [84] Mulvey, John M., Robert J. Vanderbei, and Stavros A. Zenios, "Robust optimization of large-scale systems." *Operations research*, vol. 43, no.2, pp. 264-281, 1995.
- [85] Ahuja, Ravindra K., Rolf H. Möhring, and Christos Zaroliagis, eds. *Robust and* online large-scale optimization: models and techniques for transportation systems. vol. 5868. Springer, 2009.
- [86] Ben-Tal, Aharon, and Dick Den Hertog, "Immunizing conic quadratic optimization problems against implementation errors," Technical Report. Tiburg University, 2011.
- [87] Sniedovich, Moshe, "Fooled by local robustness," *Risk Analysis*, vol. 32, no. 10, pp. 1630-1637, 2012.
- [88] Nemirovski, Arkadi, "On safe tractable approximations of chance constraints," *European Journal of Operational Research*, vol. 219, no. 3, pp. 707-718, 2012.
- [89] Thiele, Aurélie, "A note on issues of over-conservatism in robust optimization with cost uncertainty," *Optimization*, vol. 59, no. 7, pp.1033-1040, 2010.
- [90] Gancarova, Martina, and Michael J. Todd, "A robust robust optimization result," *Operations Research Letters*, vol. 40, no. 1, pp. 2-5, 2012.
- [91] Gaspar-Cunha, António, and José A. Covas, "Robustness in multi-objective optimization using evolutionary algorithms," *Computational Optimization and Applications*, vol. 39, no. 1, pp. 75-96, 2008.
- [92] Roy, Bernard, "Robustness in operational research and decision aiding: A multifaceted issue," *European Journal of Operational Research*, vol. 200, no. 3, pp. 629-638, 2010.
- [93] Gabrel, Virginie, Cécile Murat, and Lei Wu, "New models for the robust shortest path problem: complexity, resolution and generalization," *Annals of Operations Research*, vol. 207, no. 1, pp. 97-120, 2013.
- [94] Kalaı, Rim, Claude Lamboray, and Daniel Vanderpooten, "Lexicographic αrobustness: An alternative to min–max criteria" *European Journal of Operational Research*, vol. 220, no. 3, pp. 722-728, 2012.
- [95] Morrison, Tod. *A new paradigm for robust combinatorial optimization: Using persistence as a theory of evidence*. Diss. University of Colorado Denver, 2010.

- [96] Assavapokee, Tiravat, Matthew J. Realff, and Jane C. Ammons, "Min-max regret robust optimization approach on interval data uncertainty," *Journal of Optimization Theory and Applications*, vol.137, no.2, pp. 297-316, 2008.
- [97] Thiele, Aurélie, Tara Terry, and Marina Epelman, "Robust linear optimization with recourse," Technical report, Lehigh University, Bethlehem, PA, USA, 2009.
- [98] Terry, Tara L. Robust linear optimization with recourse: Solution methods and other properties. Ph.D. thesis. University of Michigan, Ann Arbor, MI, 2009.
- [99] Minoux, Michel, "On 2-stage robust LP with RHS uncertainty: complexity results and applications," *Journal of Global Optimization*, vol. 49, no.3, pp. 521-537, 2011.
- [100] Chen, Xin, and Yuhan Zhang, "Uncertain linear programs: Extended affinely adjustable robust counterparts, *Operations Research*, vol.57, no.6, pp.1469-1482, 2009.
- [101] Chen, Xin, Melvyn Sim, and Peng Sun, "A robust optimization perspective on stochastic programming," *Operations Research*, vol. 55, no.6, pp.1058-1071, 2007.
- [102] Bertsimas, Dimitris, Dan A. Iancu, and Pablo A. Parrilo, "Optimality of affine policies in multistage robust optimization," *Mathematics of Operations Research*, vol. 35, no. 2, pp. 363-394, 2010.
- [103] Iancu, Dan Andrei. Adaptive robust optimization with applications in inventory and revenue management. Diss. Massachusetts Institute of Technology, 2010.
- [104] Kuhn, Daniel, Wolfram Wiesemann, and Angelos Georghiou, "Primal and dual linear decision rules in stochastic and robust optimization," *Mathematical Programming*, vol. 130, no. 1, pp. 177-209, 2011.
- [105] O'Donoghue, Brendan, Yang Wang, and Stephen Boyd, "Min-max approximate dynamic programming," in Proc. IEEE 2011 International Symposium on Computer-Aided Control System Design (CACSD), 2011.
- [106] Shapiro, Alexander, "A dynamic programming approach to adjustable robust optimization," *Operations Research Letters*, vol. 39, no. 2, pp. 83-87, 2011.
- [107] Vayanos, Phebe, Daniel Kuhn, and Berç Rustem, "A constraint samplings approach for multi-stage robust optimization," *Automatica*, vol. 48, no. 3, pp. 459-471, 2012.
- [108] Van Parys, Bart, et al. "Distributionally robust control of constrained stochastic systems," *IEEE Trans Autom Cont.*, vol. 61, no. 2, pp. 430-442, 2016.

- [109] Chen, Xin, Melvyn Sim, and Peng Sun, "A robust optimization perspective on stochastic programming," *Operations Research*, vol. 55, no. 6, pp.1058-1071, 2007.
- [110] Bertsimas, Dimitris, and David B. Brown, "Constructing uncertainty sets for robust linear optimization," *Operations research*, vol. 57, no. 6, pp. 1483-1495, 2009.
- [111] Natarajan, Karthik, Dessislava Pachamanova, and Melvyn Sim, "Constructing risk measures from uncertainty sets," *Operations Research*, vol. 57, no. 5, pp. 1129-1141, 2009.
- [112] Dentcheva, Darinka, and Andrzej Ruszczyński, "Robust stochastic dominance and its application to risk-averse optimization," *Mathematical Programming*, vol. 123, no. 1, pp. 85-100, 2010.
- [113] Iancu, Dan A., and Nikolaos Trichakis, "Pareto efficiency in robust optimization," *Management Science*, vol. 60, no. 1, pp. 130-147, 2013.
- [114] Zhang, Yin, "General robust-optimization formulation for nonlinear programming," *Journal of optimization theory and applications*, vol. 132, no. 1, pp. 111-124, 2007.
- [115] Teo, Kwong Meng. *Nonconvex robust optimization*. Diss. Massachusetts Institute of Technology, 2007.
- [116] Boni, Odellia, A. Ben-Tal, and A. Nemirovski, "Robust solutions to conic quadratic problems and their applications," *Optimization and Engineering*, vol. 9, no. 1, pp. 1-18, 2008.
- [117] Hsiung, Kan-Lin, Seung-Jean Kim, and Stephen Boyd, "Tractable approximate robust geometric programming," *Optimization and Engineering*, vol. 9, no. 2, pp. 95-118, 2008.
- [118] Shen, Peiping, Yuan Ma, and Yongqiang Chen, "A robust algorithm for generalized geometric programming," *Journal of Global Optimization*, vol. 41, no. 4, pp. 593-612, 2008.
- [119] Mutapcic, Almir, and Stephen Boyd, "Cutting-set methods for robust convex optimization with pessimizing oracles," *Optimization Methods & Software*, vol. 24, no. 3, pp. 381-406, 2009.
- [120] Takeda, Akiko, Shunsuke Taguchi, and Tsutomu Tanaka, "A relaxation algorithm with a probabilistic guarantee for robust deviation optimization," *Computational Optimization and Applications*, vol. 47, no. 1, pp. 1-31, 2010.

- [121] Bertsimas, D., Nohadani, O, Teo, K.M., "Robust optimization for unconstrained simulation-based problems," *Operations Research*, vol. 58, pp. 161-178, 2010.
- [122] Bertsimas, Dimitris, and Omid Nohadani, "Robust optimization with simulated annealing," *Journal of Global Optimization*, vol. 48, no. 2, pp. 323-334, 2010.
- [123] Kawas, Ban, and Aurélie Thiele, "A log-robust optimization approach to portfolio management," OR Spectrum, vol. 33, no. 1, pp. 207-233, 2011.
- [124] Kawas, Ban, and Aurélie Thiele, "Short sales in log-robust portfolio management," *European Journal of Operational Research*, vol. 215, no. 3, pp. 651-661, 2011.
- [125] Jiang, Ruiwei, et al, "Benders decomposition for the two-stage security constrained robust unit commitment problem," *Online Optimization*, vol.1, pp. 10-22, 2011.
- [126] Lasserre, Jean B, "Min-max and robust polynomial optimization," Journal of Global Optimization, vol. 51, no. 1, pp. 1-10, 2011.
- [127] Sungur, Ilgaz. *The robust vehicle routing problem*. ProQuest, 2007.
- [128] Montemanni, Roberto, et al. "The robust traveling salesman problem with interval data," *Transportation Science*, vol. 41, no. 3, pp. 366-381, 2007.
- [129] Schneider, Stefan, Pasha Shabalin, and Martin Bichler, "On the robustness of nonlinear personalized price combinatorial auctions," *European Journal of Operational Research*, vol. 206, no. 1, pp. 248-259, 2010.
- [130] Zhang, Muhong, "Two-stage minimax regret robust uncapacitated lot-sizing problems with demand uncertainty," *Operations Research Letters*, vol. 39, no. 5, pp. 342-345, 2011.
- [131] Remli, Nabila, and Monia Rekik, "A robust winner determination problem for combinatorial transportation auctions under uncertain shipment volumes," *Transportation Research Part C: Emerging Technologies*, vol. 35, pp. 204-217, 2013.
- [132] Goren, Selcuk, and Ihsan Sabuncuoglu, "Robustness and stability measures for scheduling: single-machine environment," *IIE Transactions*, vol. 40, no. 1, pp. 66-83, 2008.
- [133] Hazır, Öncü, Mohamed Haouari, and Erdal Erel, "Robust scheduling and robustness measures for the discrete time/cost trade-off problem," *European Journal of Operational Research*, vol. 207, no. 2, pp. 633-643, 2010.

References

- [134] Mittal, Shashi. Algorithms for discrete, non-linear and robust optimization problems with applications in scheduling and service operations. Diss. Massachusetts Institute of Technology, 1959.
- [135] Mudchanatongsuk, Supakom, Fernando Ordóñez, and Jie Liu, "Robust solutions for network design under transportation cost and demand uncertainty," *Journal of the Operational Research Society*, vol. 1, pp. 652-662, 2008.
- [136] Zhang, Yin, "General robust-optimization formulation for nonlinear programming." *Journal of optimization theory and applications*, vol. 132, no. 11, pp. 111-124, 2007.
- [137] Mudchanatongsuk, Supakom, Fernando Ordóñez, and Jie Liu, "Robust solutions for network design under transportation cost and demand uncertainty," *Journal of the Operational Research Society*, pp. 652-662, 2008.
- [138] Torres Soto, Joaquin. Dynamic and robust capacitated facility location in time varying demand environments. Diss. 2010.
- [139] Minoux, Michel, "Robust network optimization under polyhedral demand uncertainty is NP-hard," *Discrete Applied Mathematics*, vol. 158, no. 5, pp. 597-603, 2010.
- [140] Blanquero, Rafael, Emilio Carrizosa, and Eligius MT Hendrix, "Locating a competitive facility in the plane with a robustness criterion," *European Journal of Operational Research*, vol. 215, no. 1, pp. 21-24, 2011.
- [141] Baron, Opher, Joseph Milner, and Hussein Naseraldin, "Facility location: A robust optimization approach," *Production and Operations Management*, vol. 20, no. 5, pp. 772-785, 2011.
- [142] Bardossy, Maria G, "Dual-Based Local Search for Deterministic, Stochastic and Robust Variants of the Connected Facility Location Problem," Ph.D. thesis. University of Maryland College Park. College Park, MD, USA, 2011.
- [143] Gabrel, Virginie, et al. "Robust location transportation problems under uncertain demands," *Discrete Applied Mathematics* 164 (2014): 100-111.
- [144] Zhang, M. U. H. O. N. G., and Y. O. N. G. P. E. I. Guan. "Two-stage robust unit commitment problem." *Online Optimization*, vol. 1, pp. 1-10, 2009.
- [145] Ribas, Gabriela P., Silvio Hamacher, and Alexandre Street, "Optimization under uncertainty of the integrated oil supply chain using stochastic and robust programming," *International Trans. Operational Research*, vol. 17, no. 6, pp. 777-796, 2010.

- [146] Jiang, R., Zhang, M., Li, G., Guan, Y., "Two-stage robust power grid optimization," Technical report, University of Florida, Gainesville, FL, USA, 2010.
- [147] Bertsimas, Dimitris, David Gamarnik, and Alexander Anatoliy Rikun,
 "Performance analysis of queueing networks via robust optimization," *Operations research*, vol. 59, no. 2, pp. 455-466, 2011.
- [148] Goryashko, A. P., and Arkadi S. Nemirovski, "Robust energy cost optimization of water distribution system with uncertain demand," *Automation and Remote Control*, vol. 75, no. 10, pp. 1754-1769, 2014.
- [149] Jiang, Ruiwei, Jianhui Wang, and Yongpei Guan, "Robust unit commitment with wind power and pumped storage hydro," *IEEE Trans. Power Syst.*, vol. 27, no. 2, pp. 800-810, 2012.
- [150] Ben-Tal, Aharon, and Arkadi Nemirovski, "Robust optimization-methodology and applications," *Mathematical Programming*, vol. 92, no. 3, pp. 453-480, 2002.
- [151] Ben-Tal, Aharon, Laurent El Ghaoui, and Arkadi Nemirovski. *Robust optimization*. Princeton University Press, 2009.
- [152] Bertsimas, Dimitris, David B. Brown, and Constantine Caramanis, "Theory and applications of robust optimization," *SIAM review*, vol. 53, no. 3, pp. 464-501, 2011.
- [153] Bertsimas, Dimitris, Dessislava Pachamanova, and Melvyn Sim, "Robust linear optimization under general norms," *Operations Research Letters*, vol. 32, no. 6, pp. 510-516, 2004.
- [154] Yu, Chian-Son, and Han-Lin Li, "A robust optimization model for stochastic logistic problems," *International Journal of Production Economics*, vol. 64, no. 1, pp. 385-397, 2000.
- [155] Bertsimas, Dimitris, and Melvyn Sim, "Robust discrete optimization and network flows," *Mathematical programming*, vol. 98, no. 1-3, pp. 49-71, 2003.
- [156] Bertsimas, Dimitris, et al. "Adaptive robust optimization for the security constrained unit commitment problem," *IEEE Trans Power Syst.*, vol. 28, no. 1, pp. 52-63, 2013.
- [157] Street A, Oliveira F, Arroyo JM, "Contingency-constrained unit commitment with n-k security criterion: a robust optimization approach," *IEEE Trans. Power Syst.*, vol. 26, no. 3, pp. 1581-1590, 2013.

References

- [158] Baringo L, Conejo AJ, "Offering strategy via robust optimization," *IEEE Trans. Power Syst.*, vol. 26, no. 3, pp. 1418-1425, 2011.
- [159] D. C. Montgomery, C. L. Jennings, and M. Kulahci. *Introduction to time series analysis and forecasting*. vol. 526. John Wiley & Sons, 2011.
- [160] ENERCON Wind Turbines, Wellesweiler, Germany. [Online]. Available: http://www.enercon.de/en-en/Windenergieanlagen.htm
- [161] Illinois Institute of Rural Affair, USA. [Online]. Available: http://www.illinoiswind.org/index.asp
- [162] National Renewable Energy Laboratory, [Online]. Available: http://www.nrel.gov/solar/index.asp
- [163] General Algebraic Modeling System, GAMS, 2013. [Online]. Available: http://www.gams.com/
- [164] Federal Energy Regulation Commission (FERC), Electric Tariff, [Online]. http://www.ferc.gov/
- [165] PJM Electricity market, [online]. Available: http://www.pjm.com

Publications from the Work
Published Work

SCI Journal

 R. A. Gupta and Nandkishor Gupta "A Robust Optimization Based Approach for Microgrid Operation in Deregulated Environment" in Energy Conversion and Management, Elsevier vol.93, pp 121-131,2015.

IEEE International Conferences:

- R. A. Gupta and Nandkishor Gupta "Generation Scheduling at PCC in Grid Connected Microgrid" *IEEE* International Conference on Recent Advances and Innovations in Engineering (ICRAIE-2014), pp. 1-5, 2014.
- R. A. Gupta and Nandkishor Gupta "Robust Microgrid Operation Considering Renewable Power Uncertainties" in *IEEE* National Power System Conference (NPSC-2014), IIT, Guwahati, India pp. 1-5, 2014.

Communicated (Under Review) Work

- 4. R. A. Gupta and Nandkishor Gupta "A Robust Approach for Microgrid Operation considering Solar Power Uncertainty" under review for Electric Power System Research, Elsevier.
- 5. R. A. Gupta and Nandkishor Gupta "Microgrid Operation Management considering Solar and Wind power Uncertainties" under review for Energy Conversion and Management, Elsevier.